First print



New South Wales

Fire and Emergency Services Levy Bill 2017

Explanatory note

This explanatory note relates to this Bill as introduced into Parliament.

Overview of Bill

The object of this Bill is to provide for the imposition of a fire and emergency services levy (the *levy*). The levy will be payable on all land (other than government land) that is situated in the area of a council or on Lord Howe Island.

The person liable for the levy will generally be the owner of the land. However, in some cases a lessee of land owned by the Crown or a State owned corporation will be liable for the levy.

The levy is payable in respect of each financial year, starting with the year commencing on 1 July 2017.

There are 2 components to the levy: a fixed component and an ad valorem component.

The fixed component of the levy is a base rate specified in the proposed Act (which can be varied by regulation). The ad valorem component of the levy is calculated by applying an ad valorem rate specified by the Treasurer to the land value of the land. The Treasurer calculates the ad valorem rate in accordance with a formula that takes into account the funding targets of the various emergency services (fire brigades, rural fire brigades and state emergency services).

Both components of the levy vary according to the classification of the land on which the levy is charged.

Land can be classified into the following property sectors under the proposed Act:

- (a) government land,
- (b) public benefit land,
- (c) farmland,
- (d) residential land,

- (e) industrial land,
- (f) commercial land.

Only government land is exempt from the levy.

Councils will be responsible for classifying land in their area into those property sectors.

The levy is to be charged by, and collected by, councils on behalf of the State. Councils will be responsible for recovering the levy along with the rates and charges that they levy under the *Local Government Act 1993*. Councils will be required to account for the levy by paying a proportion of all money collected as rates, charges and levies to the State.

The levy replaces the emergency services contributions that are currently required to be made by insurance companies. Those contributions are abolished by the proposed Act and provision is also made for the winding up of those contribution schemes.

Outline of provisions

Part 1 Preliminary

Part 1 provides for the name and commencement of the proposed Act and defines words and expressions used in the proposed Act.

Part 1 also provides that the proposed Act applies to Lord Howe Island. For that purpose, the Island is taken to be a council area and the Lord Howe Island Board is taken to be a council.

Part 2 Fire and emergency services levy

Part 2 provides for imposition of the levy on all land situated in a council area (other than government land) and other critical aspects of the levy (which are as described in the Overview above).

Part 3 Amount of levy

Part 3 provides for the 2 components of the levy (the fixed component and the ad valorem component).

The fixed component is the base rate for the property sector into which the land is classified. That amount is specified in the proposed Act (but can be varied by regulation).

The ad valorem component is the amount calculated by applying the ad valorem rate specified by the Treasurer for the relevant property sector to the land value of the land on which the levy is charged. In most cases, the land value of land will be the land value specified in a general valuation provided by the Valuer-General. For land on Lord Howe Island and other land that is not currently valued by the Valuer-General, the regulations can specify a land value or method by which land value is determined. The Valuer-General is then required to determine the value of the land on the basis of those regulations.

There will be discounts for vacant land and for land owned by eligible pensioners.

Part 4 Method for calculating ad valorem rate

Part 4 contains detailed provisions about how the Treasurer calculates the ad valorem rate of the levy for each property sector. These calculations are based on the funding target of each emergency services organisation, the collection costs that will be payable under the scheme, the amount of revenue raised from the levy in previous years, and the amount actually spent on emergency services in those years.

Part 5 Classification of land for levy

Part 5 requires councils to classify all land within their area for the purpose of charging the levy. Land can be classified into the following property sectors:

- (a) government land,
- (b) public benefit land,
- (c) farmland,
- (d) residential land,
- (e) industrial land,
- (f) commercial land.

Residential land, industrial land and commercial land can also be sub-classified as vacant land (in which case the levy is discounted).

The Part provides for the criteria for classification and for the classification process. This includes provision for appeals to the Land and Environment Court.

Councils will be required to provide information about their classification and sub-classification decisions to the Valuer-General.

Part 6 Charging and collection of levy by council

Part 6 requires a council to charge the levy on all leviable land in its area. All land is leviable land, other than government land. The procedure for the charging of the levy is similar to the procedure for the charging of rates by the council. The levy is charged annually, but can be paid in quarterly instalments.

The Treasurer has power to approve guidelines relating to the waiver or reduction of the levy because of hardship. The Chief Commissioner of State Revenue (the *Chief Commissioner*) may waive or reduce the levy in accordance with those guidelines. If a person claims a waiver or reduction of the levy dishonestly, the Chief Commissioner can recover the waived amount from that person.

A council can recover an unpaid levy as a debt payable to the council. Levies, when paid, can be held in the council's consolidated fund. A council can enter into arrangements with the Chief Commissioner for the transfer of its debt recovery functions to the Chief Commissioner, in respect of any specified levy or class of levies.

Part 6 also requires the Treasurer to make payments to councils (*collection payments*) to reimburse them for the ongoing cost of collecting and recovering the levy on behalf of the State, responding to or defending appeals against decisions made by councils under the proposed Act and other costs incurred in connection with the proposed Act.

Part 7 Payment of levy to State

Part 7 requires a council to pay a proportion of all relevant payments made to it (that is, payments made for assessed charges or rates, or for assessed levies) to the Chief Commissioner. These amounts are to be paid to the Chief Commissioner in the form of collection instalments. The payments represent the proportion of money collected by councils that is attributable to the levy. The Part sets out how payments are to be calculated, when they are to be made and record keeping requirements. Parts of the *Taxation Administration Act 1996* will apply to those payments as if they were a tax payable by the council. This will enable the Chief Commissioner to assess the amounts payable (if necessary) and to exercise investigation and enforcement functions under that Act in relation to the amounts payable. It will also enable the Chief Commissioner to charge interest on overdue amounts (but not penalty tax).

Part 8 Harmonisation with council rates and charges

Part 8 applies various provisions of the *Local Government Act 1993*, the regulations under that Act and the *Strata Schemes Development Act 2015* to the levy, and to the council's exercise of functions under the proposed Act. The objective is to ensure that the charging of the levy is harmonised, insofar as is practical, with the levying of rates and charges by councils. A council will be able to recover the levy in the same way as it recovers rates and charges (including by selling land).

Part 9 Compliance and enforcement

Part 9 enables the Valuer-General to monitor and review council compliance with their classification obligations under the proposed Act and the Chief Commissioner to monitor and audit council compliance with their financial obligations under the proposed Act. Both officers will be able to give to the Treasurer a certificate (a *compliance certificate*) about their assessment of the council's compliance.

The Treasurer will be able to order a council to remedy or restrain any breach of the proposed Act or to withhold collection payments if a council breaches the Act. The issue of a compliance certificate that specifies compliance issues will be sufficient justification for that action.

Part 10 Miscellaneous

Part 10 contains miscellaneous provisions relating to the levy, including provisions that:

- (a) enable the Treasurer to require specified information about the levy to be included in a levy notice, and
- (b) permit information to be shared by the agencies involved in the administration of the proposed Act, and
- (c) enable the Treasurer to delegate functions under the proposed Act, and
- (d) provide for service of notices under the proposed Act, and
- (e) provide for the application of the *Recovery of Imposts Act 1963* to the levy, and
- (f) enable the making of regulations under the proposed Act.

Schedule 1 Public benefit land

Schedule 1 sets out the criteria for classification of land as public benefit land.

Schedule 2 Industrial land

Schedule 2 sets out the criteria for classification of land as industrial land.

Schedule 3 Savings, transitional and other provisions

Schedule 3 contains savings, transitional and other provisions consequent on the enactment of the proposed Act. The provisions make the levy payable from 1 July 2017. The provisions also:

- (a) validate things done by councils and others in anticipation of the commencement of the scheme, and
- (b) provide for certain matters that may be dealt with by regulation under the proposed Act, pending the making of regulations, and
- (c) set deadlines for initial land classifications under the scheme and adjust the deadlines for other things that must be done in 2017 to facilitate the start of the scheme, and
- (d) permit the Treasurer to add to the revenue target, in the initial period of the scheme, the costs of the Emergency Services Levy Insurance Monitor and the start-up costs of the scheme (which means that those costs can be recouped via the levy).

Schedule 4 Amendment of other legislation

Monitor to investigate overcharging under scheme

Schedule 4.1 amends the *Emergency Services Levy Insurance Monitor Act 2016* to require the Emergency Services Levy Insurance Monitor (the *Monitor*) to investigate and assess overcharging by insurance companies under the existing emergency services contributions scheme. The Monitor will be able to assess whether insurance companies have over-collected levies in the final 2 years of the scheme. If an insurance company is assessed to be liable for an over-collection amount, the insurance company will be required to refund the amount to policy holders or, if that is not practicable, to pay the amount to the State.

The Monitor will also be able to require insurance companies to provide information about emergency services levy reform in invoices or other statements.

Abolition of emergency services contributions

Schedule 4.2, 4.6 and 4.7 amend the following Acts to provide for the abolition of the current emergency services funding scheme and the winding up of that scheme:

- (a) Fire Brigades Act 1989,
- (b) Rural Fires Act 1997,
- (c) State Emergency Service Act 1989.

Under the current scheme, insurance companies are required to make contributions towards emergency services funding. Those contributions will continue to be required for the financial year commencing on 1 July 2016, but are abolished from 1 July 2017. Insurance companies will be required to lodge a final return so that a final assessment can be made of their liability under the scheme. If an insurance company is assessed to have under-contributed in the final year of the scheme, the insurance company will be liable to pay a contribution deficit.

Councils will continue to be required to contribute to emergency services funding. They will continue to be required collectively to contribute 11.7% of the annual funding target for each service.

Other amendments

Schedule 4.3 amends the *Land and Environment Court Act 1979* to provide for appeals against classification and sub-classification decisions by councils under the proposed Act and for judicial review.

Schedule 4.4 and 4.5 make consequential amendments to the *Local Government Act 1993* and the *Local Government (General) Regulation 2005*.

Schedule 4.8 amends the *Taxation Administration Act 1996* to signpost the application of some of the provisions of that Act to collection instalments payable by councils under the proposed Act. **Schedule 4.9** amends the *Valuation of Land Act 1916*:

- (a) to require land valuations that are done for rating and charging purposes, or for the purposes of the levy, to be provided at least once every 3 years (instead of once every 4 years), and
- (b) to make it clear that the Act applies to the levy.

First print



New South Wales

Fire and Emergency Services Levy Bill 2017

Contents

| | | | Page |
|--------|------|---|------|
| Part 1 | Pre | liminary | |
| | 1 | Name of Act | 2 |
| | 2 | Commencement | 2 |
| | 3 | Definitions | 2 |
| | 4 | Application of Act to Lord Howe Island | 3 |
| Part 2 | Fire | and emergency services levy | |
| | 5 | Fire and emergency services levy | 4 |
| | 6 | Land on which levy is charged | 4 |
| | 7 | Who is liable to pay levy | 4 |
| | 8 | Levy is payable for each financial year | 4 |
| | 9 | Amount of levy | 4 |
| | 10 | Levy payable on each separate parcel of land | 4 |
| | 11 | Levy payable to council | 5 |
| | 12 | Council must charge and collect levy on behalf of State | 5 |
| | 13 | Levy to be apportioned over whole year | 5 |
| | 14 | No levy for government land | 5 |

Page

| Part 3 | Amount of levy | | | | |
|--------|--|--|--|--|--|
| | Division 1 Fundamentals | | | | |
| | 15 Components of levy: fixed and ad valorem 16 Fixed component of levy 17 Ad valorem component of levy 18 Land value of land | 6 6 6 | | | |
| | Division 2 Unvalued land | | | | |
| | Unvalued land—meaning Regulations may specify land value of unvalued land Valuer-General to ascertain land value of unvalued land Application of Valuation of Land Act 1916 | 7 7 7 8 | | | |
| | Division 3 Discounts | | | | |
| | 23 Discount for vacant land 24 Discount for eligible pensioners 25 Pensioner discount amount 26 Change of circumstances | 8 8 9 9 | | | |
| Part 4 | Method for calculating ad valorem rate | | | | |
| | Division 1 Treasurer to calculate ad valorem rate | | | | |
| | Published ad valorem rate Ad valorem rate formula Revenue target for property sector Relevant proportion Estimate of non-vacant properties Estimate of vacant properties Estimate of pensioner discount properties Councils to provide levy estimate information to Valuer-General Information to be used by Valuer-General to provide estimates | 10 10 11 11 11 12 12 12 | | | |
| | Division 2 FESL revenue target for ad valorem calculation | | | | |
| | 36 FESL revenue target 37 Funding target 38 Collection cost 39 Recovered over-collection amount | 13 13 13 14 | | | |
| Part 5 | Classification of land for levy | | | | |
| | Division 1 Councils required to classify land | | | | |
| | 40 Land to be classified into property sectors 41 Steps for classification 42 Requirements for classification as government land 43 Requirements for classification as public benefit land 44 Requirements for classification as farmland | 15 15 16 17 17 | | | |

- Requirements for classification as residential land 17 45
- 46 Requirements for classification as industrial land 17 Classification of land not yet being used 18 47

| | | | | Page |
|--------|-------|--------------------|--|------|
| | Divis | sion 2 | Classification process | |
| | 48 | Declar | ation of property sector | 18 |
| | 49 | | ve date for classification | 18 |
| | 50 | | of classification | 18 |
| | 51 | | ation for classification or review of classification | 18 |
| | 52 | | il to decide application | 19 |
| | Divis | sion 3 | Sub-classification of land as vacant land | |
| | 53 | Sub-cl | assification of land as vacant land | 19 |
| | 54 | | a for sub-classification | 20 |
| | 55 | | ation for sub-classification | 20 |
| | 56 | | il to decide application | 20 |
| | 57 | Effectiv | ve date for sub-classification | 21 |
| | 58 | Sub-cl | assifications to be reviewed | 21 |
| | Divis | sion 4 | Revocation of sub-classification of land | |
| | 59 | | ation of sub-classification | 21 |
| | 60 | | ve date for revocation of sub-classification | 21 |
| | 61 | | of revocation | 22 |
| | 62 | | ation for review of revocation | 22 |
| | 63 | Counc | il to decide application | 22 |
| | Divis | sion 5 | Appeals | |
| | 64 | | Is to Land and Environment Court | 23 |
| | 65 | Counc | il to give Valuer-General notice of appeal | 23 |
| | 66 | Power | s of Land and Environment Court on appeal | 23 |
| | | sion 6 | Provision of information to Valuer-General | |
| | 67 | | ils to provide annual return to Valuer-General | 23 |
| | 68 | valuer | -General may require further information | 24 |
| | | sion 7 | Miscellaneous | |
| | 69 | | about classification and sub-classification | 24 |
| | 70 | | and use changes | 24 |
| | 71 | Adjusti sub-cla | ment of levy following change in classification or assification | 24 |
| | 72 | Chang | es deemed to take effect at end of quarterly instalment period | 25 |
| | 73 | Notice | of change of circumstances is required | 25 |
| | 74 | | fication of mixed development land | 25 |
| | 75 | Parcel | of land in 2 or more areas | 26 |
| | 76 | Power | s of entry and inspection | 26 |
| Part 6 | Cha | rging | and collection of levy by council | |
| | Divis | sion 1 | Charging of levy | |
| | 77 | Counc | il must charge levy | 27 |
| | 78 | | b be charged by notice | 27 |
| | 79 | Timing | for service of levy notice | 27 |
| | 80 | | or payment of levy | 27 |
| | 81 | Payme | ent by single instalment | 28 |

| | 82 83 | - | nt by quarterly instalments I of interest on overdue levies | 28 28 |
|--------|---|--|---|--|
| | Divis | sion 2 | Waiver or reduction of levy | |
| | 84 85 86 87 88 | Chief C Liable p Applica | ip guidelines commissioner may waive or reduce levy person required to make full and true disclosure tion of Taxation Administration Act 1996 nees to levy include interest on levy | 29 29 29 30 30 |
| | Divis | sion 3 | Recovery of levy | |
| | 89 90 91 92 93 | Levies Transfe Record | ery of levy as debt paid to be kept in council's consolidated fund er of levy recovery to Chief Commissioner of levies on payments | 30 31 31 32 32 |
| Part 7 | Рау | ment o | f levy to State | |
| | Divis | sion 1 | Preliminary | |
| | 94 95 96 97 | Relevar Assess | l required to apportion relevant payments nt payment—meaning ed rate or charge—meaning ed levy—meaning | 33 33 33 34 |
| | Divis | sion 2 | Payment of collection instalments to State | |
| | 98 99 100 101 102 103 104 | Collecti Reconc Amount Calcula Paymer | nt payments to be applied towards old debt first on instalments to be paid to Chief Commissioner ciliation periods t of collection instalment tion of levy proportion nt dates for reconciliation periods I to keep a record of relevant payments | 34 34 35 35 35 35 |
| | Divis | sion 3 | Functions of Chief Commissioner | |
| | 105 106 107 108 | Applica Council | commissioner may recover collection instalment from council tion of certain provisions of Taxation Administration Act 1996 Is to provide annual return to Chief Commissioner on of other information to Chief Commissioner | 36 36 36 37 |
| Part 8 | Har | monisa | tion with council rates and charges | |
| | 109 110 111 112 113 114 115 116 117 | Applica Aggrega Levy is Writing Collecti Transfe Transfe | ots and company titles treated as separate parcels tion of Act to company title ation of parcels a charge on land off levy and accrued interest on of levy from occupier er of land to pay levy er of land sion of land | 38 38 38 38 38 38 39 39 39 |

Page

| | | | Page |
|------------|------------|---|----------|
| | 118 | Definition of "eligible pensioner" | 39 |
| | 119 | Waiving payment of levy | 39 |
| | 120 | Certificate as to levy | 39 |
| | 121 | Notice of transfer of land | 39 |
| | 122 | Notices | 39 |
| | 123 | Proof of ownership | 40 |
| | 124 | Recovery of levy | 40 |
| | 125 | Sale of land to recover levy | 40 |
| | 126 127 | Regulations relating to rates and charges Application of strata provisions | 40 40 |
| | 127 | Power to modify and apply further legislative provisions | 40 |
| | 129 | References to council extend to Lord Howe Island Board | 40 |
| Part 9 | Cor | mpliance and enforcement | |
| | Divis | ision 1 Preliminary | |
| | 130 | Definitions | 42 |
| | Divis | ision 2 Monitoring role of Valuer-General | |
| | 131 | Monitoring of council compliance with classification obligations | s 42 |
| | 132 | Reviews by Valuer-General | 42 |
| | 133 | Valuer-General may issue compliance certificate | 42 |
| | Divis | ision 3 Monitoring role of Chief Commissioner | |
| | 134 | Monitoring of council compliance with financial obligations | 43 |
| | 135 | Audits by Chief Commissioner | 43 |
| | 136 | Chief Commissioner may issue compliance certificate | 43 |
| | | ision 4 Treasurer's enforcement powers | |
| | 137 | Order by Treasurer | 44 |
| | 138 | Withholding collection payments | 44 |
| Part 10 | Mis | scellaneous | |
| | 139 | Notice relating to fire and emergency services levy | 46 |
| | 140 | Sharing of information between councils and agencies | 46 |
| | 141 | Disclosure of information to Treasurer | 47 |
| | 142 | Notices to be in writing | 47 |
| | 143 144 | Act binds the Crown Delegation of functions | 47 47 |
| | 145 | Service of documents | 47 |
| | 146 | Application of Recovery of Imposts Act 1963 | 48 |
| | 147 | General tax exemption for council does not apply | 48 |
| | 148 | Nature of proceedings for offences | 48 |
| | 149 | Regulations | 48 |
| | 150 | Review of Act | 48 |
| Schedule ' | | Public benefit land | 49 |
| Schedule 2 | | Industrial land | 51 |
| Schedule 3 | | Savings, transitional and other provisions | 54 |
| Schedule 4 | 4 | Amendment of other legislation | 58 |



New South Wales

Fire and Emergency Services Levy Bill 2017

No , 2017

A Bill for

An Act to provide for the charging and collection of a fire and emergency services levy; to abolish the emergency services insurance contribution scheme; and to make related amendments to other legislation.

| The | Legisl | ature of New South Wales enacts: | 1 |
|-----|--------|--|----------------|
| Par | t 1 | Preliminary | 2 |
| 1 | Nam | e of Act | 3 |
| | | This Act is the Fire and Emergency Services Levy Act 2017. | 4 |
| 2 | Com | mencement | 5 |
| 2 | Con | This Act commences on the date of assent to this Act. | |
| | | | 6 |
| 3 | Defi | nitions | 7 |
| | (1) | In this Act: | 8 |
| | | ad valorem component of levy— see section 17. | 9 |
| | | ad valorem rate formula—see section 28. | 10 |
| | | area means an area within the meaning of the Local Government Act 1993. | 11 |
| | | Note. Lord Howe Island is taken to be an area under this Act. | 12 |
| | | assessed levy—see section 97. | 13 |
| | | assessed rate or charge—see section 96. | 14 |
| | | <i>Chief Commissioner</i> means the Chief Commissioner of State Revenue under the <i>Taxation Administration Act 1996</i> . | 15 16 |
| | | collection instalment means a collection instalment payable under section 99. | 17 |
| | | commercial land means land classified as commercial land under this Act. | 18 |
| | | <i>compliance certificate</i> means a compliance certificate issued by the Valuer-General under section 133 or by the Chief Commissioner under section 136. | 19 20 |
| | | compliance issue—see sections 133 and 136. | 21 |
| | | council means a council within the meaning of the Local Government Act 1993. | 22 |
| | | Court means the Land and Environment Court. | 23 |
| | | <i>Crown</i> means the Crown in right of New South Wales and includes any NSW Government agency and any statutory body representing the Crown in right of New South Wales. | 24 25 26 |
| | | <i>eligible pensioner</i> in relation to a levy on leviable land means a person who is an eligible pensioner in relation to land under Chapter 15 of the <i>Local Government Act</i> 1993. | 27 28 29 |
| | | farmland means land classified as farmland under this Act. | 30 |
| | | <i>FESL revenue target</i> means the FESL revenue target determined by the Treasurer under section 36. | 31 32 |
| | | financial year means a period of 12 months commencing on 1 July in each year. | 33 |
| | | <i>fixed component</i> of levy—see section 16. | 34 |
| | | <i>function</i> includes a power, authority or duty, and <i>exercise</i> a function includes perform a duty. | 35 36 |
| | | government land means land classified as government land under this Act. | 37 |
| | | hardship guidelines—see section 84. | 38 |
| | | industrial land means land classified as industrial land under this Act. | 39 |
| | | land includes a stratum within the meaning of the Valuation of Land Act 1916. | 40 |
| | | <i>land value</i> —see section 18 (6). | 41 |
| | | lease has the same meaning as it has in the Local Government Act 1993. | 42 |
| | | lease for value means a lease for more than nominal rent. | 43 |

| | | levia | <i>ble land</i> has the meaning given by section 6. | 1 |
|---|------|----------------------------------|--|----------------------------|
| | | | (or <i>FESL</i>) means the fire and emergency services levy payable under this Act. | 2 |
| | | - | estimate information—see section 34. | 3 |
| | | • | <i>notice</i> means the notice by which a levy is charged (see section 78). | 4 |
| | | • | recovery arrangement—see section 91. | 5 |
| | | • | valuation—see section 21. | 6 |
| | | liabl | <i>e person</i> , in relation to land, means a person who is liable for a levy for that land ho would be liable for a levy for the land if the land were leviable land. | 7 8 |
| | | | <i>e State owned corporation</i> means a State owned corporation that is declared to liable State owned corporation by a regulation under section 42. | 9 10 |
| | | Lord | Howe Island means the Island under the Lord Howe Island Act 1953. | 11 |
| | | | <i>itor</i> means the Emergency Services Levy Insurance Monitor appointed under <i>Emergency Services Levy Insurance Monitor Act 2016</i> . | 12 13 |
| | | owne | er has the same meaning as it has in the Local Government Act 1993. | 14 |
| | | Valu | el of land includes a parcel that comprises a stratum within the meaning of the <i>ation of Land Act 1916</i> . | 15 16 |
| | | unde | See also section 10 for the circumstances in which land is a separate parcel of land this Act. | 17 18 |
| | | - · | nent order—see section 86. | 19 |
| | | • | <i>ioner discount amount</i> —see section 25. | 20 |
| | | | <i>erty sector</i> —see section 40. | 21 |
| | | • | <i>ic benefit land</i> means land classified as public benefit land under this Act. | 22 |
| | | - | ished ad valorem rate—see section 27. | 23 |
| | | - | terly instalment period—see section 82. | 24 |
| | | | ant payment—see section 95. | 25 |
| | | | ant proportion—see section 30. | 26 |
| | | | <i>lential land</i> means land classified as residential land under this Act. | 27 |
| | | the . | <i>cowned corporation</i> means a State owned corporation (within the meaning of <i>State Owned Corporations Act 1989</i>) or a subsidiary of a State owned oration. | 28 29 30 |
| | | unva | <i>lued land</i> —see section 19. | 31 |
| | | vaca | nt land means land sub-classified as vacant land under this Act. | 32 |
| | | vaca | nt land discount—see section 23. | 33 |
| | | Note. interp | . The <i>Interpretation Act 1987</i> contains definitions and other provisions that affect the pretation and application of this Act. | 34 35 |
| | (2) | Note | s included in this Act do not form part of this Act. | 36 |
| | (3) | to pa that o Note . | is Act, a reference to a financial year, when immediately preceded by a reference rticular years (in the format of [<i>year</i>]/[<i>year</i>]), is a reference to the financial year commences and ends in those years (respectively). For example, a reference to the 2017/2018 financial year is a reference to the financial commencing in 2017 and ending in 2018. | 37 38 39 40 41 |
| 4 | Appl | - | n of Act to Lord Howe Island | 42 |
| | ••• | | levy payable under this Act extends to land situated on Lord Howe Island and, | 43 |
| | | for th | nat purpose: | 44 |
| | | (a) | Lord Howe Island is taken to be an area under this Act, and | 45 |
| | | (b) | the Lord Howe Island Board is taken to be the council for that area and has all the functions of a council in relation to that area under this Act. | 46 47 |

| Par | t 2 | Fire and emergency services levy | 1 |
|--------|------|---|-----------------------|
| 5 | Fire | and emergency services levy | 2 |
| | | A fire and emergency services levy is payable on all leviable land. | 3 |
| 6 Land | | l on which levy is charged | 4 |
| | (1) | For the purposes of this Act, <i>leviable land</i> is any land situated in the area of a council that is within one of the following property sectors: (a) public benefit land, (b) farmland, (c) residential land, | 5 6 7 8 9 |
| | | (d) industrial land, | 10 |
| | | (e) commercial land. | 11 |
| | (2) | Land is within a property sector if the land is classified to be within that property sector under this Act. | 12 13 |
| 7 | Who | is liable to pay levy | 14 |
| | (1) | The owner for the time being of leviable land is liable to pay the levy, except as provided by this section. | 15 16 |
| | (2) | If leviable land owned by the Crown or a State owned corporation (other than a liable State owned corporation) is the subject of a lease for value and the dominant use of the land is for the purposes of the lease, the lessee under the lease is liable to pay the levy. | 17 18 19 20 |
| | (3) | If there are 2 or more owners, or 2 or more lessees, who are liable to pay the levy in respect of the same land, they are jointly and severally liable to pay the levy. | 21 22 |
| | | Note. Government land is exempt from the levy. Land owned by the Crown or a State owned corporation is to be classified as government land under this Act unless the land is the subject of a lease for value and the dominant use of the land is for the purposes of the lease. There are some exceptions to that classification requirement. | 23 24 25 26 |
| 8 | Levy | is payable for each financial year | 27 |
| | | The levy is payable for each financial year. | 28 |
| | | Note. The levy can be paid as a single instalment or as quarterly instalments (see Part 6). | 29 |
| 9 | Amo | unt of levy | 30 |
| | | The amount of the levy payable is provided for by Part 3. | 31 |
| 10 | Levy | payable on each separate parcel of land | 32 |
| | (1) | A separate levy is payable for each separate parcel of land. | 33 |
| | (2) | Land is taken to be a separate parcel of land for the purposes of this Act: | 34 |
| | | (a) if the land is separately valued under the <i>Valuation of Land Act 1916</i> , or | 35 |
| | | (b) if the land is the subject of one valuation under that Act, or | 36 |
| | | (c) in any other circumstances provided for by or under this Act. | 37 |
| | (3) | Subsection (2) does not affect the application of this Act to land that is not valued under the <i>Valuation of Land Act 1916</i> . | 38 39 |
| | (4) | This section is subject to Parts 5 and 8 and any regulations under this section. Note. Part 5 (Division 7) and Part 8 provide for further circumstances in which land is treated as a separate parcel of land for levy purposes. | 40 41 42 |

| | (5) | The regulations may make further provision for the circumstances in which something is, or is not, taken to be a separate parcel of land under this Act. | 1 2 |
|----|-------|--|----------|
| 11 | Levy | v payable to council | 3 |
| | | The levy is payable to the council for the area in which the leviable land is situated. | 4 |
| 12 | Cou | ncil must charge and collect levy on behalf of State | 5 |
| | (1) | A council is responsible for charging the levy on land situated in the council's area and for collecting the levy. | 6 7 |
| | (2) | A council charges and collects the levy on behalf of the State. | 8 |
| | (3) | The council must account for amounts collected by paying collection instalments to the Chief Commissioner in accordance with this Act. | 9 10 |
| 13 | Levy | to be apportioned over whole year | 11 |
| | (1) | A levy is proportionate to the portion of the year for which land is leviable land. | 12 |
| | (2) | A levy for land that is within a particular property sector is proportionate to the portion of the year for which the land is within that property sector. | 13 14 |
| 14 | No le | evy for government land | 15 |
| | | There is no levy payable in respect of government land. Note. Land that is not within a council area is also excluded because it is not leviable land. | 16 17 |

| Par | t 3 | Amount of levy | 1 |
|------|-------|---|----------------------------|
| Divi | sion | 1 Fundamentals | 2 |
| 15 | Com | ponents of levy: fixed and ad valorem | 3 |
| | (1) | The levy payable is the total of the following: | 4 |
| | | (a) the fixed component of the levy, | 5 |
| | | (b) the ad valorem component of the levy. | 6 |
| | (2) | The components of the levy are based on the property sector that the land is within. Note. Part 5 sets out how land is classified into different property sectors. | 7 8 |
| 16 | Fixed | d component of levy | 9 |
| | (1) | The fixed component of the levy is the base rate for the property sector that the land is within. | 10 11 |
| | (2) | The <i>base rate</i> is as follows: | 12 |
| | | (a) \$100 for public benefit land, | 13 |
| | | (b) \$100 for residential land, | 14 |
| | | (c) \$200 for farmland, | 15 |
| | | (d) \$200 for industrial land, | 16 |
| | | (e) \$200 for commercial land. | 17 |
| | (3) | The regulations may prescribe a different base rate for any property sector. If they do, the base rate for that property sector is the prescribed rate. | 18 19 |
| | (4) | The regulations may provide for adjustments to the base rate for any property sector to account for movements in the consumer price index. If they do, the fixed component of the levy is the adjusted base rate (as provided for by the regulations) for the property sector that the land is within for the financial year for which the levy is charged. | 20 21 22 23 24 |
| 17 | Ad v | alorem component of levy | 25 |
| | (1) | The ad valorem component of the levy is calculated by applying the relevant ad valorem rate to the land value of the parcel of land on which the levy is charged. | 26 27 |
| | (2) | The relevant ad valorem rate is the published ad valorem rate for the property sector that the land is within for the financial year for which the levy is charged. | 28 29 |
| | (3) | The council that charges the levy is to calculate the ad valorem component of the levy. | 30 31 |
| 18 | Land | value of land | 32 |
| | (1) | The land value of land is the land value (within the meaning of the <i>Local Government Act 1993</i>) used by the council for the purpose of levying rates under Chapter 15 of the <i>Local Government Act 1993</i> , subject to this section. | 33 34 35 |
| | (2) | For any period in which the rates payable in respect of land are postponed under Division 2 of Part 8 of Chapter 15 of the <i>Local Government Act 1993</i> , the levy is to be charged as if the land value of the land were the non-attributable value of the land (despite section 61 of the <i>Valuation of Land Act 1916</i>). | 36 37 38 39 |
| | (3) | The <i>non-attributable value</i> of land is the land value of the land for the purpose of levying rates under Chapter 15 of the <i>Local Government Act 1993</i> minus the | 40 41 |

attributable part of the land value (within the meaning of Division 2 of Part 8 of Chapter 15 of the *Local Government Act 1993*).

- (4) If the council does not levy rates on the land, the land value of the land is the land value of the land determined in accordance with the *Valuation of Land Act 1916*.
- (5) However, if the land is unvalued land, the land value of the land is the land value of the land determined in accordance with Division 2.
- (6) A reference in this Act to the land value of land, for levy purposes, is a reference to the land value of land determined as provided for by this section and Division 2. Note. Section 61 of the Valuation of Land Act 1916 provides that land value must be ascertained by reference to the valuation list, and any supplementary lists, provided by the Valuer-General under Part 5 of the Valuation of Land Act 1916. Section 62 of the Valuation of Land Act 1916 provides that the most recent valuation list is to be used to determine land value.

Division 2 Unvalued land

19 Unvalued land—meaning

- (1) The following land is *unvalued land*:
 (a) land on Lord Howe Island,
 (b) any other land declared by the regulations to be unvalued land.
 - (2) Land is not unvalued land if it is required to be valued for the purposes of another tax.
 - (3) Land is required to be valued for the purposes of another tax if the Valuer-General is required, under the *Valuation of Land Act 1916*, to provide a valuation of the land which is used for the purposes of a rate or charge that is payable under the *Local Government Act 1993*, land tax or any other rate or tax (other than the levy).

20 Regulations may specify land value of unvalued land

- (1) The regulations may specify the land value of unvalued land, or the method by which the land value of unvalued land is to be calculated, for the purposes of this Act.
- (2) The land value of unvalued land, or the method by which the land value is calculated, may be specified by reference to the owner of the land, the liable person in relation to the land, the use of the land, the location of the land or in any other way.
- (3) The regulations may specify a zero land value for unvalued land. In that case, the land value of the land is zero and no levy is payable on the land (including no fixed component).
- (4) The regulations have effect despite anything to the contrary in the *Valuation of Land Act 1916*.

21 Valuer-General to ascertain land value of unvalued land

- (1) The Valuer-General is to ascertain the land value of each parcel of land that is unvalued land in accordance with the regulations.
- (2) The land value of unvalued land is the land value as ascertained by the Valuer-General under this section.
- (3) A valuation made by the Valuer-General under this section is a *levy valuation*.
- (4) This section does not apply if the regulations specify a zero land value for the unvalued land.

22 Application of Valuation of Land Act 1916

Levy valuations are to be entered in the Register of Land Values kept under the *Valuation of Land Act 1916*.

- (2) Information entered in the Register of Land Values about levy valuations is to be included in any valuation list or supplementary lists compiled by the Valuer-General and furnished to a rating or taxing authority under the *Valuation of Land Act 1916*.
- (3) Subject to this Division, the Valuation of Land Act 1916 and the regulations under that Act apply, with any necessary modifications, to levy valuations and to the information about levy valuations entered in the Register of Land Values in the same way as they apply to general valuations and to the information about general valuations entered in the Register of Land Values under that Act. Note. Parts 3 and 4 of the Valuation of Land Act 1916 permit objections to, and appeals against, valuations made by the Valuer-General. Part 5 of the Valuation of Land Act 1916, among other things, requires councils to use the valuation list as the basis for charging rates or taxes.
- (4) For the purpose of applying the *Valuation of Land Act 1916* to levy valuations of unvalued land on Lord Howe Island:
 - (a) Lord Howe Island is taken to be an area, and
 - (b) the Lord Howe Island Board is taken to be a rating or taxing authority, and a council, for that area.
- (5) The regulations under the *Valuation of Land Act 1916* may modify the application of, or disapply, any provision of the *Valuation of Land Act 1916* or the regulations under that Act in relation to unvalued land and levy valuations.

Division 3 Discounts

23 Discount for vacant land

- (1) A liable person is eligible for a vacant land discount in respect of the levy payable on leviable land if the land is vacant land.
- (2) The levy payable for a financial year is to be reduced by the vacant land discount (expressed as a percentage) for the financial year for which the levy is charged.
- (3) The *vacant land discount* is 0.5 or, if another rate is prescribed by the regulations, that rate.
- (4) The discount applies to the fixed and ad valorem components only and not to any interest that accrues on overdue payments of the levy.
- (5) The discount applies only to the proportion of the levy that is payable for the number of quarterly instalment periods during the financial year in which the land is vacant land.

24 Discount for eligible pensioners

- (1) A liable person is eligible for a pensioner discount in respect of the levy payable on leviable land if the liable person is an eligible pensioner.
- (2) The levy payable for a financial year is to be reduced by the pensioner discount amount.
- (3) The pensioner discount amount is proportionate to the number of quarterly instalment periods during the financial year in which the liable person is an eligible pensioner.

| (4) | Only one pensioner discount can be claimed in respect of a parcel of land, even if more than one eligible pensioner is a liable person in respect of the land. | 1 2 |
|-----|--|----------------------|
| (5) | If the pensioner discount amount is more than the levy that (but for the discount) would be payable, the levy payable is zero. | 3 4 |
| Pen | sioner discount amount | 5 |
| (1) | The <i>pensioner discount amount</i> is: | 6 |
| | (a) \$50 in the 2017/2018 financial year, and | 7 |
| | (b) the indexed amount in any subsequent financial year. | 8 |
| (2) | The indexed amount is the amount determined by multiplying \$50 by A/B, where: | 9 |
| | <i>A</i> is the Sydney CPI number for December in the financial year before the financial year for which the indexed amount is being determined, and | 10 11 |
| | B is the Sydney CPI number for December 2016, and | 12 |
| | A/B is calculated to the nearest 3 decimal places (and a fourth decimal place of 5 is to be rounded up). | 13 14 |
| (3) | If the indexed amount is not a multiple of 10 cents, the amount is to be rounded to the nearest 10 cent multiple (and an amount of 5 cents is to be rounded up). | 15 16 |
| (4) | The Treasurer is to publish a notice on the NSW legislation website by 30 April before the commencement of each financial year (starting with the 2018/2019 financial year) specifying the indexed amount for that financial year, calculated in accordance with this section. | 17 18 19 20 |
| (5) | In this section: | 21 |
| | <i>Sydney CPI number</i> means the Consumer Price Index (All Groups Index) for Sydney issued by the Australian Statistician or, if the Australian Statistician fails or ceases to issue that number, another number or index prescribed by the regulations. | 22 23 24 |
| Cha | nge of circumstances | 25 |
| (1) | If circumstances occur during a financial year that cause an entitlement to a pensioner discount on the levy to arise, the entitlement is taken to arise at the beginning of the first day of the quarterly instalment period that starts immediately after the quarterly instalment period during which those circumstances occurred. | 26 27 28 29 |
| (2) | If circumstances occur during a financial year that cause an entitlement to a pensioner discount on the levy to cease, the entitlement is taken to cease at the end of the last day of the quarterly instalment period during which those circumstances occurred. | 30 31 32 |
| (3) | If, at the time the entitlement is taken to arise or cease, the levy for the whole year has been paid in full, the portion of the payment that is proportionate to the number of quarterly instalment periods remaining after that time is to be credited to the levy payable in respect of the remaining part of the year. | 33 34 35 36 |

Part 4 Method for calculating ad valorem rate

Division 1 Treasurer to calculate ad valorem rate

27 Published ad valorem rate

(1) The Treasurer must, before the commencement of each financial year, determine an ad valorem rate of the levy for the next financial year.

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

29

30

31

32

33

34

35

36

37

38

39

40

41

- (2) The determination must specify an ad valorem rate of the levy for each property sector (other than government land).
- (3) The determination is to be made by order published on the NSW legislation website.
- (4) The order is to be published on the NSW legislation website by the date of 30 April that occurs immediately before the commencement of the financial year.
- (5) An ad valorem rate specified in the order for a property sector is the *published ad valorem rate* for that property sector for the financial year for which it is specified.
- (6) If an order specifying the ad valorem rate of the levy for a financial year is not published on the NSW legislation website by the date of 30 April that occurs immediately before the commencement of that financial year, the published ad valorem rate for a property sector is taken to be the published ad valorem rate for the financial year that immediately precedes the financial year.
- (7) Subsection (6) does not apply to the ad valorem rate for the 2017/2018 financial year. Note. Schedule 3 permits things required to be done by 30 April 2017 to be done later, if this Act has not commenced by 23 April.

28 Ad valorem rate formula

- (1) The Treasurer is to determine the ad valorem rate of the levy for a property sector using the ad valorem rate formula.
- (2) The *ad valorem rate formula* is the following formula:

$$A = \frac{R - (F \times (N_{nv} + (D \times N_v))) + (C \times P)}{(V_{nv} + (D \times V_v))}$$

where:

A is the ad valorem rate of the levy for the property sector for the financial year for which the determination is being made.

R is the revenue target for the property sector for the financial year.

F is the fixed component of the levy for the property sector for the financial year.

 N_{nv} is the estimated number of non-vacant properties in the property sector for the financial year.

D is 1 minus the vacant land discount for the financial year.

 N_v is the estimated number of vacant properties in the property sector for the financial year.

C is the pensioner discount amount for the financial year.

P is the estimated number of pensioner discount properties in the property sector for the financial year.

 V_{nv} is the estimated value of non-vacant properties in the property sector for the financial year.

 V_{v} is the estimated value of vacant properties in the property sector for the financial year.

| | (3) | For the public benefit land and farmland property sectors, $N_{\rm v}$ and $V_{\rm v}$ are taken to be zero. | 1 2 |
|----|-------|---|----------------------------|
| | (4) | Expressions used in the ad valorem rate formula have the meanings given by this Division. | 3 4 |
| 29 | Reve | nue target for property sector | 5 |
| | | The <i>revenue target</i> for a property sector for a financial year is the amount determined by applying the relevant proportion for the property sector to the FESL revenue target for that financial year. Note. Division 2 sets out how the Treasurer determines the FESL revenue target. | 6 7 8 9 |
| 30 | Polo | vant proportion | |
| 30 | | | 10 |
| | (1) | The <i>relevant proportion</i> for each property sector is as follows: (a) for public benefit land—0.33%, | 11 |
| | | (a) for farmland— 4.56% , | 12 13 |
| | | (c) for residential land—58.07%, | 13 14 |
| | | (d) for industrial land—10.38%, | 14 |
| | | (e) for commercial land—26.66%. | 16 |
| | (2) | The regulations may specify a different relevant proportion for any property sector and, if they do, the relevant proportion for the property sector is the proportion specified in the regulations. | 17 18 19 |
| | (3) | The regulations may specify different proportions for different financial years. | 20 |
| 31 | Estin | nate of non-vacant properties | 21 |
| | (1) | The <i>estimated number of non-vacant properties</i> in a property sector is an estimate adopted by the Valuer-General, for a financial year, of the number of non-vacant parcels of land in that property sector in that financial year. | 22 23 24 |
| | (2) | The <i>estimated value of non-vacant properties</i> in a property sector is an estimate adopted by the Valuer-General, for a financial year, of the total land value, for levy purposes, of all non-vacant parcels of land in that property sector in that financial year. | 25 26 27 28 |
| | (3) | A parcel of land is non-vacant if it is not sub-classified as vacant land. | 29 |
| | (4) | The Valuer-General is to prepare and adopt an estimated number of non-vacant properties and an estimated value of non-vacant properties, for each property sector, and give each estimate to the Treasurer by 15 March in the year in which the financial year for which the estimate is made commences or by a later date approved by the Treasurer. | 30 31 32 33 34 |
| | (5) | An estimate of value is to be rounded to the nearest multiple of \$100,000 (with an amount of \$50,000 rounded up). | 35 36 |
| | (6) | Estimates are not required for the government land property sector. | 37 |
| | (7) | In this section, a reference to a parcel of land is a reference to a parcel of land that is a separate parcel of land under this Act. | 38 39 |
| 32 | Estin | nate of vacant properties | 40 |
| | (1) | The <i>estimated number of vacant properties</i> in a property sector is an estimate adopted by the Valuer-General, for a financial year, of the number of vacant parcels of land in that property sector in that financial year. | 41 42 43 |

| | (2) | The <i>estimated value of vacant properties</i> in a property sector is an estimate adopted by the Valuer-General, for a financial year, of the total land value, for levy purposes, of all vacant parcels of land in that property sector in that financial year. | 1 2 3 |
|----|-------|--|----------------------|
| | (3) | The Valuer-General is to prepare and adopt an estimated number of vacant properties and an estimated value of vacant properties, for each property sector, and give each estimate to the Treasurer by 15 March in the year in which the financial year for which the estimate is made commences or by a later date approved by the Treasurer. | 4 5 6 7 |
| | (4) | An estimate of value is to be rounded to the nearest multiple of \$100,000 (with an amount of \$50,000 rounded up). | 8 9 |
| | (5) | Estimates are not required for the government land, public benefit land or farmland property sectors. | 10 11 |
| | (6) | In this section, a reference to a parcel of land is a reference to a parcel of land that is a separate parcel of land under this Act. | 12 13 |
| 33 | Estir | nate of pensioner discount properties | 14 |
| | (1) | The <i>estimated number of pensioner discount properties</i> in a property sector is an estimate adopted by the Valuer-General, for a financial year, of the number of parcels of land in that property sector that will be eligible for a pensioner discount in that financial year. | 15 16 17 18 |
| | (2) | The Valuer-General is to prepare and adopt an estimated number of pensioner discount properties, for each property sector, and give each estimate to the Treasurer by 15 March in the year in which the financial year for which the estimate is made commences or by a later date approved by the Treasurer. | 19 20 21 22 |
| | (3) | An estimate is not required for the government land property sector. | 23 |
| | (4) | In this section, a reference to a parcel of land is a reference to a parcel of land that is a separate parcel of land under this Act. | 24 25 |
| 34 | Cour | ncils to provide levy estimate information to Valuer-General | 26 |
| | (1) | Each council is to provide to the Valuer-General, by 15 February in each year or by a later date prescribed by the regulations, a return in a form approved by the Valuer-General that contains the levy estimate information for the council's area. | 27 28 29 |
| | (2) | The <i>levy estimate information</i> means the following: | 30 |
| | | (a) information about the classification and sub-classification of land in the council's area, | 31 32 |
| | | (b) information about the number of parcels of land in the council's area that are eligible for a pensioner discount under this Act, | 33 34 |
| | | (c) any other information that the Valuer-General reasonably requires for the purpose of exercising his or her functions under this Division. | 35 36 |
| | (3) | The levy estimate information must be up to date as of 31 December in the year before the return is required to be provided. | 37 38 |
| | (4) | In this section, a reference to a parcel of land is a reference to a parcel of land that is a separate parcel of land under this Act. | 39 40 |
| 35 | Infor | mation to be used by Valuer-General to provide estimates | 41 |
| | (1) | The Valuer-General is to make the estimates required to be made by the Valuer-General under this Division on the basis of the levy estimate information provided to the Valuer-General by councils in accordance with this Act and any other information that the Valuer-General considers relevant. | 42 43 44 45 |

(2)If a council fails to provide any levy estimate information for the council's area by 1 the date it is required to provide that information under this Act, the Valuer-General 2 may make the estimate on the basis of any other information available to the 3 Valuer-General. 4 (3)The Valuer-General may, in adopting an estimate under this Division, make any 5 adjustments or allowances that the Valuer-General considers it appropriate to make. 6 Division 2 FESL revenue target for ad valorem calculation 7 FESL revenue target 8 The Treasurer is to determine the FESL revenue target for a financial year using the (1)9 following formula: 10 $ER_{t} = (FT_{t} \times 0.81) + CC_{t} + (ER_{t-2} - AR_{t-2}) - OC_{t}$ where: 11 ER_t is the FESL revenue target for the financial year (represented by "t"). 12 FT_t is the funding target for the financial year. 13 CC_t is the collection cost for the levy in the financial year. 14 $ER_{t,2}$ is the FESL revenue target for the financial year that commenced 2 years 15 before the period "t" (represented as "t-2"). 16 $AR_{t,2}$ is the actual FESL revenue for the financial year that commenced 2 years 17 before the period "t" (represented as "t-2"), as certified by the Treasurer. 18 OC_t is the recovered over-collection amount for the financial year. 19 (2)To avoid doubt, if $ER_{t-2} - AR_{t-2}$ is a negative amount, that amount is subtracted from 20 the total. 21 When calculating the FESL revenue target for the 2017/2018 financial year and the (3)22 2018/2019 financial year, $ER_{t-2} - AR_{t-2}$ is taken to be zero. 23 (4)The FESL revenue target is to include any additions made under Part 2 of Schedule 3. 24 Note. Under Schedule 3, the Monitor budget and start-up costs for the levy can be added to 25 the FESL revenue target during the initial period of the scheme. 26 Expressions used in the FESL revenue target formula have the meanings given by (5) 27 this Division. 28 Funding target 29 The *funding target* for a financial year is the total of the following funding targets 30 for the financial year: 31 (a) the SES funding target (within the meaning of Part 5A of the *State Emergency* 32 Service Act 1989), 33 the fire brigades funding target (within the meaning of Part 5 of the Fire (b) 34 Brigades Act 1989), 35 the rural fire brigade funding target (within the meaning of Part 5 of the Rural (c)36 Fires Act 1997). 37 **Collection cost** 38 The *collection cost* for the levy in a financial year is the amount calculated by the (1)39 Treasurer in accordance with this section. 40

37

38

36

(2) The Treasurer is to calculate the collection cost for a financial year using the following formula:

$$CC_{t} = EC_{t} + (AC_{t-2} - EC_{t-2})$$

where:

 CC_t is the collection cost for the financial year (represented by "t").

 EC_t is the estimated collection cost for the financial year.

 AC_{t-2} is the actual collection cost for the financial year that commenced 2 years before the period "t" (represented as "t-2").

 EC_{t-2} is the estimated collection cost for the financial year that commenced 2 years before the period "t" (represented as "t-2").

- (3) The Treasurer must, before 30 April in each year, prepare and adopt an estimate of the collection cost for the next financial year (which is the *estimated collection cost* for that financial year).
- (4) The estimated collection cost for a financial year is the Treasurer's estimate of the total of the following:
 - (a) the collection payments that will be made to councils during the financial year,
 - (b) the costs that will be incurred by or on behalf of the Crown in exercising functions under this Act during the financial year.
- (5) The *actual collection cost* for a financial year is the amount certified by the Treasurer to be the total of the following:
 - (a) all collection payments that were made to councils during the financial year,
 - (b) all costs that were incurred by or on behalf of the Crown in exercising functions under this Act during the financial year.
- (6) For the 2017/2018 financial year and the 2018/2019 financial year, $AC_{t-2} EC_{t-2}$ is taken to be zero.

39 Recovered over-collection amount

- (1) The *recovered over-collection amount* for a financial year is the sum of all over-collection amounts paid to or recovered by the Chief Commissioner under Part 3A of the *Emergency Services Levy Insurance Monitor Act 2016* in the period of 12 months ending on the date of 31 March occurring before the commencement of the financial year, as determined by the Chief Commissioner.
- (2) The Chief Commissioner is to advise the Treasurer of the recovered over-collection amount by the date of 15 April that occurs before the commencement of the financial year or by a later date approved by the Treasurer.

| Part | : 5 | Classification of land for levy | 1 |
|-------|--------------|---|----------------------|
| Divis | sion ' | Councils required to classify land | 2 |
| 40 | Land | to be classified into property sectors | 3 |
| | (1) | its area to be within one of the following categories (each of which is a <i>property sector</i> for the purposes of this Act): | 4 5 6 |
| | | | 7 |
| | | | 8 |
| | | | 9 |
| | | | 10 |
| | | | 11 |
| | (-) | | 12 |
| | (2) | | 3 4 |
| | (3) | The land must be classified in accordance with this Part. | 15 |
| | (4) | | 16 17 |
| 41 | Steps | s for classification | 18 |
| | (1) | | 19 20 |
| | (2) | | 21 22 |
| | (3) | | 23 24 |
| | | (a) consider whether the land is public benefit land, and | 25 |
| | | | 26 27 |
| | (4) | | 28 29 |
| | | | 30 31 |
| | | | 32 33 |
| | (5) | | 34 35 |
| | | | 36 |
| | | | 37 38 |
| | (6) | public benefit land, farmland, residential land or industrial land, the council is to classify the land as commercial land. | 39 10 11 12 |
| | | | 13 |

| 42 | Requirements for classification as government land | | | |
|----|--|---|----------------------------|--|
| | (1) | Land meets the requirements for classification as <i>government land</i> , subject to the regulations, if: | 2 3 | |
| | | (a) the land is owned by the Crown, unless: | 4 | |
| | | (i) the regulations declare that the land is not government land, or | 5 | |
| | | (ii) the land is leased and subsection (2) applies to the land, or | 6 | |
| | | (b) the land is owned by a State owned corporation, unless: | 7 | |
| | | (i) the regulations declare that the corporation is a liable State owned corporation, or | 8 9 | |
| | | (ii) the land is leased and subsection (2) applies to the land, or | 10 | |
| | | (c) the land is owned by the Commonwealth, or | 11 | |
| | | (d) the land is owned by a council, unless: | 12 | |
| | | (i) the regulations declare that the land is not government land, or | 13 | |
| | | (ii) the land is leased and subsection (2) applies to the land, or | 14 | |
| | | (e) the land is situated within any part of the Western Division (within the meaning of the <i>Crown Lands Act 1989</i>) that is not constituted as an area under the <i>Local Government Act 1993</i> , or | 15 16 17 | |
| | | (f) the land is the premises of a mission to which the <i>Diplomatic Privileges and</i> <i>Immunities Act 1967</i> of the Commonwealth applies, or | 18 19 | |
| | | (g) the land is declared by the regulations to be government land. | 20 | |
| | (2) | Land that is owned by the Crown, a State owned corporation or a council does not meet the requirements for classification as government land if the land is the subject of a lease for value and the dominant use of the land is for the purposes of the lease. | 21 22 23 | |
| | (3) | However, subsection (2) does not apply if: | 24 | |
| | | (a) the land is owned by the New South Wales Land and Housing Corporation, or by the Aboriginal Housing Office, and the dominant use of the land under the lease is for residential accommodation, or | 25 26 27 | |
| | | (b) the land is owned by the Crown and leased to an individual who is employed in the service of the Crown, in his or her capacity as an employee, and the dominant use of the land under the lease is for residential accommodation, or | 28 29 30 | |
| | | (c) the land is owned by a State owned corporation and leased to an individual who is employed by the State owned corporation, in his or her capacity as an employee, and the dominant use of the land under the lease is for residential accommodation, or | 31 32 33 34 | |
| | | (d) the land is owned by a council and leased to an individual who is employed by the council, in his or her capacity as an employee, and the dominant use of the land under the lease is for residential accommodation, or | 35 36 37 | |
| | | (e) a regulation under subsection (1) (g) declares the land to be government land (unless otherwise provided by the regulations). | 38 39 | |
| | (4) | Residential accommodation means the type of residential accommodation that permits rateable land to be categorised as residential under section 516 of the <i>Local Government Act 1993</i> . | 40 41 42 | |
| | (5) | A provision of a regulation that is made under this section may specify the date on and from which the provision takes effect for classification purposes. The date may occur before, on or after the date of publication of the regulation on the NSW legislation website but not before the commencement of the financial year in which publication occurs. | 43 44 45 46 47 | |

| | (6) | If no date is specified, any change in the requirements for classification of land made by the regulation is taken to have effect at the beginning of the first date of the next quarterly instalment period after the regulation takes effect. | 1 2 3 | | | | |
|----|--|--|----------------------|--|--|--|--|
| 43 | Requirements for classification as public benefit land | | | | | | |
| | (1) | Land meets the requirements for classification as <i>public benefit land</i> if: | 5 | | | | |
| | | (a) the land is not-for-profit land, and | 6 | | | | |
| | | (b) the liable person is using the land for a purpose referred to in Schedule 1, and | 7 | | | | |
| | | (c) that purpose is the dominant use of the land. | 8 | | | | |
| | (2) | Land is <i>not-for-profit land</i> if: | 9 | | | | |
| | | (a) the land is not used for any profit-making purpose, or | 10 | | | | |
| | | (b) the dominant use of the land is for a purpose that is not a profit-making purpose. | 11 12 | | | | |
| | (3) | Land may be used for a profit-making purpose even if no profit is made. | 13 | | | | |
| | (4) | The regulations may declare any use of land to be, or not to be, for a profit-making purpose. | 14 15 | | | | |
| | (5) | The regulations may amend Schedule 1. | 16 | | | | |
| | (6) | A provision of a regulation that is made under this section may specify the date on and from which the provision takes effect for classification purposes. The date may occur on or after the date of publication of the regulation on the NSW legislation website. | 17 18 19 20 | | | | |
| | (7) | If no date is specified, any change in the requirements for classification of land made by the regulation is taken to have effect at the beginning of the first date of the next quarterly instalment period after the regulation takes effect. | 21 22 23 | | | | |
| 44 | Requ | irements for classification as farmland | 24 | | | | |
| | | Land meets the requirements for classification as <i>farmland</i> if it is categorised as farmland by the council for ratings purposes under the <i>Local Government Act 1993</i> or would be so categorised if it were rateable land under that Act. | 25 26 27 | | | | |
| 45 | Requirements for classification as residential land | | | | | | |
| | | Land meets the requirements for classification as <i>residential land</i> if it is categorised as residential by the council for ratings purposes under the <i>Local Government Act 1993</i> or would be so categorised if it were rateable land under that Act. | 29 30 31 | | | | |
| 46 | Requirements for classification as industrial land | | | | | | |
| | (1) | Land meets the requirements for classification as <i>industrial land</i> if the dominant use of the land is for a purpose referred to in Schedule 2. | 33 34 | | | | |
| | (2) | The regulations may amend Schedule 2. | 35 | | | | |
| | (3) | A provision of a regulation that is made under this section may specify the date on and from which the provision takes effect for classification purposes. The date may occur on or after the date of publication of the regulation on the NSW legislation website. | 36 37 38 39 | | | | |
| | (4) | If no date is specified, any change in the requirements for classification of land made by the regulation is taken to have effect at the beginning of the first date of the next quarterly instalment period after the regulation takes effect. | 40 41 42 | | | | |
| | | | | | | | |

47 Classification of land not yet being used

Section 519 of the *Local Government Act 1993* applies to the classification of land as public benefit land or industrial land under this Act in the same way as it applies to the categorisation of land as farmland or residential under that Act.

Division 2 Classification process

48 Declaration of property sector

- (1) A council classifies land to be within a particular property sector by declaring the land to be within that property sector.
- (2) A council may change the classification of a parcel of land at any time by making another declaration.
- (3) A classification of a parcel of land ceases to have effect when a subsequent classification of the land takes effect.
- (4) A council must change the classification of land that is in a property sector if the requirements for classification in that property sector change and, as a result, the land ceases to meet the requirements for classification in that property sector.
- (5) A council may classify or change the classification of land on its own initiative or on the application of a liable person.

49 Effective date for classification

- (1) A classification of land takes effect from the date specified for the purpose in the declaration of the council (this is the *effective date* for the classification).
- (2) The effective date for the classification must be a date of 1 July, 1 October, 1 January or 1 April.

Note. These dates are the beginning of each quarterly instalment period for the levy.

(3) The effective date for the classification may be a date that is before the date the declaration is made.

Note. See also sections 72 and 73.

50 Notice of classification

- (1) A council must give notice to a liable person of the classification declared for any parcel of land for which the person is a liable person, and the effective date for the classification, within 30 days after it is made.
- (2) The notice must, if the council classified the land on its own initiative:
 - (a) state that the person has the right to apply to the council for a review of the classification of the land, or of the effective date for the classification, or both, and
 - (b) state that the person has the right to appeal to the Land and Environment Court if dissatisfied with the council's review.
- (3) Notice is not required to be given if the land is classified as government land.

51 Application for classification or review of classification

- (1) A liable person may apply to the council at any time:
 - (a) to have the person's land classified as being within a particular property sector for the purposes of this Act, or
 - (b) for a review of a classification by the council of the person's land or the effective date for the classification (or both).

(2) An application:

- (a) must be in a form approved by the council, and
- (b) must include a description of the land, and
- (c) must nominate the property sector that the applicant considers the land should be within (if a different classification is sought), and

- (d) must set out the reasons why the applicant considers the land should be within that property sector, and
- (e) must nominate the effective date for the classification that is sought by the applicant, and
- (f) must be accompanied by the fee (if any) charged by the council for the application.
- (3) If the council has reasonable grounds for believing that the land does not meet the requirements for classification for the nominated property sector, or that the effective date for classification nominated is not appropriate, it may notify the applicant of any further information it requires in order to decide the application.
- (4) The regulations may prescribe a maximum fee that may be charged by a council for making an application under this section.

52 Council to decide application

- (1) After considering an application for classification or review of classification, and any further information provided at the request of the council, the council must declare the property sector for the land and specify the effective date for the classification.
- (2) The council must declare the land to be within the property sector nominated in the application unless it has reasonable grounds for believing that the land does not meet the requirements for classification within that property sector.
- (3) The council must notify the applicant of its decision.
- (4) If the council declares the land to be within a property sector that is not the property sector nominated by the applicant or declares an effective date for the classification that is not the date nominated by the applicant, the council must include in the notice:
 - (a) the reasons for the decision, and
 - (b) information about appeals to the Land and Environment Court (including the time limit for making an appeal).
- (5) If the council has not notified the applicant of its decision within 40 days after the application is made to it, the council is taken, at the end of the 40-day period, to have declared the land to be within its existing property sector with the same effective date as it previously specified.
- (6) The fee (if any) for making an application for classification or review of classification is to be refunded:
 - (a) if the council declares the land to be within the property sector nominated by the applicant, or
 - (b) if the council fails to notify the applicant of a decision within 40 days after the application is made to it.

Division 3 Sub-classification of land as vacant land

53 Sub-classification of land as vacant land

(1) A council may sub-classify a parcel of land as vacant land if the land meets the requirements for sub-classification as vacant land.

| | (2) | | may be sub-classified as vacant land only if a liable person for the land applies council for that sub-classification, in accordance with this Part. | 1 2 | | |
|----|-------|---------------------------------|---|----------------------|--|--|
| | (3) | A cou | uncil sub-classifies land as vacant land by declaring the land to be vacant land. | 3 | | |
| 54 | Crite | Criteria for sub-classification | | | | |
| | | Land | meets the requirements for sub-classification as vacant land if: | 5 | | |
| | | (a) | the land is residential land, industrial land or commercial land, and | 6 | | |
| | | (b) | there are no buildings or structures on the land that are being used or that could be used for a residential, industrial or commercial purpose, and | 7 8 | | |
| | | (c) | the land is not being used for storage or treatment of goods, materials or any other thing, and | 9 10 | | |
| | | (d) | the land and any use of the land meets any other requirements for sub-classification as vacant land specified in the regulations. | 11 12 | | |
| 55 | Appl | icatior | n for sub-classification | 13 | | |
| | (1) | | ble person for land may apply to the relevant council at any time to have the on's land declared to be vacant land. | 14 15 | | |
| | (2) | An aj | pplication: | 16 | | |
| | | (a) | must be in a form approved by the council, and | 17 | | |
| | | (b) | must include a description of the land concerned, and | 18 | | |
| | | (c) | must include the reasons why the applicant considers the land to be vacant land, and | 19 20 | | |
| | | (d) | must nominate the date from which the applicant considers the land should be sub-classified as vacant land, and | 21 22 | | |
| | | (e) | must be accompanied by the fee (if any) charged by the council for the application. | 23 24 | | |
| | (3) | requi furthe | e council has reasonable grounds for believing that the land does not meet the rements for sub-classification as vacant land, it may notify the applicant of any er information it requires in order to be satisfied that the land is within that ategory. | 25 26 27 28 | | |
| | (4) | The r maki | regulations may prescribe a maximum fee that may be charged by a council for ng an application under this section. | 29 30 | | |
| 56 | Cour | ncil to | decide application | 31 | | |
| | (1) | | considering an application for land to be declared to be vacant land, and any er information provided by the applicant, the council must either: | 32 33 | | |
| | | (a) | declare the land to be vacant land, or | 34 | | |
| | | (b) | refuse the application. | 35 | | |
| | (2) | | council must declare the land to be vacant land if the council is satisfied that the satisfies the requirements for sub-classification as vacant land. | 36 37 | | |
| | (3) | | council must notify the applicant of its decision. The council must include the ns for its decision if it refuses the application. | 38 39 | | |
| | (4) | If the | council refuses the application, the council must include in the notice: | 40 | | |
| | | (a) | the reasons for the decision, and | 41 | | |
| | | (b) | information about appeals to the Land and Environment Court (including the time limit for making an appeal). | 42 43 | | |
| | | | | | | |

| | (5) | If the council has not notified the applicant of its decision within 40 days after the application is made to it, the council is taken, at the end of the 40-day period, to have refused the application. | 1 2 3 |
|------|-------|--|----------------------|
| | (6) | The fee (if any) for making an application that land be declared to be vacant land is to be refunded: | 4 5 |
| | | (a) if the council declares the land to be vacant land, or | 6 |
| | | (b) if the council fails to notify the applicant of a decision within 40 days after the application is made to it. | 7 8 |
| 57 | Effec | ctive date for sub-classification | 9 |
| | (1) | A declaration that a parcel of land is vacant land takes effect from a date specified for the purpose in the declaration (this is the <i>effective date</i> for the sub-classification). | 10 11 |
| | (2) | The effective date for the sub-classification must be a date of 1 July, 1 October, 1 January or 1 April. | 12 13 |
| | | Note. These dates are the beginning of each quarterly instalment period for the levy. | 14 |
| | (3) | The effective date for the sub-classification may be a date that is before the date the declaration is made. | 15 16 |
| | (4) | However, the earliest effective date for the sub-classification is the date of 1 July in the financial year in which the application is received by the council. The effective date for the sub-classification cannot be in a previous financial year (despite any other provision of this Act). | 17 18 19 20 |
| | | Note. See also sections 72 and 73. | 21 |
| 58 | Sub- | classifications to be reviewed | 22 |
| | (1) | A council must ensure that it reviews the sub-classification of any land as vacant land at least once every 4 years. | 23 24 |
| | (2) | A council is to give written confirmation to the Valuer-General that it has complied with this section if requested to do so by the Valuer-General. | 25 26 |
| Divi | sion | 4 Revocation of sub-classification of land | 27 |
| 59 | Revo | ocation of sub-classification | 28 |
| | (1) | A sub-classification of land as vacant land ceases to have effect when it is revoked by the council. | 29 30 |
| | (2) | A council may revoke the sub-classification of land as vacant land at any time by making a declaration to that effect. | 31 32 |
| | (3) | A council must revoke the sub-classification of land as vacant land if it ceases to meet the requirements for sub-classification as vacant land. | 33 34 |
| | (4) | A council must consider whether the sub-classification of land as vacant land should be revoked if information comes to its attention that the land is not vacant land. | 35 36 |
| | (5) | Sub-classification of land as vacant land is taken to cease to have effect if the classification of the land changes and the new classification is government land, public benefit land or farmland. | 37 38 39 |
| 60 | Effec | ctive date for revocation of sub-classification | 40 |
| | (1) | A revocation of sub-classification takes effect on a date specified by the council in its declaration that the sub-classification is revoked (this is the <i>effective date</i> for the revocation). | 41 42 43 |

| | (2) | The effective date for the revocation must be a date of 1 July, 1 October, 1 January or 1 April. Note. These dates are the beginning of each quarterly instalment period for the levy. | 1 2 3 |
|----|-------|---|-------------|
| | (3) | The effective date for the revocation may be a date that is before the date the declaration is made. | 4 5 |
| 61 | Notio | ce of revocation | 6 |
| | (1) | A council must give notice to a liable person of the revocation of a sub-classification of land as vacant land, for any parcel of land for which the person is a liable person, and the effective date for the classification, within 30 days after it is made. | 7 8 9 |
| | (2) | The notice must: | 10 |
| | | (a) state that the person has the right to apply to the council for a review of the revocation, or of the effective date for the revocation, or both, and | 11 12 |
| | | (b) state that the person has the right to appeal to the Land and Environment Court if dissatisfied with the council's review. | 13 14 |
| 62 | Appl | ication for review of revocation | 15 |
| | (1) | A liable person may apply to the council at any time for a review of any of the following decisions of a council: | 16 17 |
| | | (a) a decision to revoke the sub-classification of the person's land as vacant land, | 18 |
| | | (b) a decision to specify the effective date for the revocation. | 19 |
| | (2) | An application: | 20 |
| | | (a) must be in a form approved by the council, and | 21 |
| | | (b) must include a description of the land, and | 22 |
| | | (c) must set out the reasons why the applicant considers the land should continue to be sub-classified as vacant land (if sub-classification is sought), and | 23 24 |
| | | (d) must nominate the effective date for the revocation of sub-classification of the land as vacant land (if sub-classification is no longer sought), and | 25 26 |
| | | (e) must be accompanied by the fee (if any) charged by the council for the application. | 27 28 |
| | (3) | The regulations may prescribe a maximum fee that may be charged by a council for an application under this section. | 29 30 |
| 63 | Cour | ncil to decide application | 31 |
| | (1) | After considering an application for review of a council decision under this Division, and any further information provided at the request of the council, the council must: | 32 33 |
| | | (a) affirm its decision, or | 34 |
| | | (b) set aside its decision. | 35 |
| | (2) | If the council sets aside its decision, it may substitute a new decision. | 36 |
| | (3) | The council must notify the applicant of its decision. | 37 |
| | (4) | If the council affirms its original decision, the council must include in the notice: | 38 |
| | | (a) the reasons for the decision, and | 39 |
| | | (b) information about appeals to the Land and Environment Court (including the time limit for making an appeal). | 40 41 |
| | (5) | If the council has not notified the applicant of its decision within 40 days after the application is made to it, the council is taken, at the end of the 40-day period, to have | 42 43 |

affirmed its decision to revoke sub-classification with the same effective date as it previously specified.

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

29

30

31

32

33

34

35

36

37

38

39

- (6) The fee (if any) for making an application that land be declared to be vacant land is to be refunded:
 - (a) if the council sets aside its original decision, or
 - (b) if the council fails to notify the applicant of a decision within 40 days after the application is made to it.

Division 5 Appeals

64 Appeals to Land and Environment Court

- (1) A liable person who is dissatisfied with a relevant decision may appeal to the Land and Environment Court against the decision.
- (2) In this Division, a *relevant decision* means:
 - (a) a decision that the council makes on an application by the person for the classification of the person's land, or
 - (b) a decision that the council makes on an application for a review of the classification of the person's land, or the effective date for the classification, or both, or
 - (c) a decision that the council makes on an application by the person for the sub-classification of the person's land as vacant land, or
 - (d) a decision that the council makes on an application for a review of the revocation of the sub-classification of the person's land as vacant land, or the effective date for the revocation, or both.
- (3) An appeal must be made no later than 30 days after notice of the relevant decision is given to the person.

65 Council to give Valuer-General notice of appeal

- (1) A council is to give the Valuer-General notice of any appeal against a relevant decision of the council that is made to the Land and Environment Court.
- (2) The notice must be given to the Valuer-General within 7 days after the council is given notice of the appeal.

66 Powers of Land and Environment Court on appeal

The Land and Environment Court may do any of the following on an appeal against a relevant decision:

- (a) affirm or vary the decision of the council,
- (b) set aside the decision of the council and make a new decision in substitution for that decision,
- (c) order the council to refund the fee paid for making the application for classification, sub-classification or review (as the case requires).

Division 6 Provision of information to Valuer-General

67 Councils to provide annual return to Valuer-General

Each council is to provide to the Valuer-General, by 15 February in each year, a return for the year ending on the preceding 31 December relating to the decisions 41 made by the council under this Part.

| | (2) | The return is to contain the following information about the year ending on the preceding 31 December: | 1 2 | | |
|------|------------------------|---|----------------|--|--|
| | | (a) information about the decisions made by the council under this Part, including any information specified in the regulations, | 3 4 | | |
| | | (b) any other information that the Valuer-General reasonably requires for the purpose of exercising his or her functions under this Act. | 5 6 | | |
| | (3) | The return is to be in a form approved by the Valuer-General. | 7 | | |
| | (4) | The information provided must be up to date as of 31 December in the year before the return is required to be provided. | 8 9 | | |
| | (5) | The regulations may make further provision for returns under this section. | 10 | | |
| | (6) | Without limiting subsection (5), the regulations may require the general manager or the public officer of a council to certify the accuracy and completeness of information contained in the return. | 11 12 13 | | |
| 68 | Valu | er-General may require further information | 14 | | |
| | (1) | The Valuer-General may, at any time, by notice to a council, require the council to provide the following to the Valuer-General: | 15 16 | | |
| | | (a) information about decisions made by the council under this Part (including reasons for decisions), | 17 18 | | |
| | | (b) any other information the Valuer-General reasonably requires for the purpose of exercising his or her functions under this Act. | 19 20 | | |
| | (2) | A council must not fail to comply with any such requirement. | 21 | | |
| | (3) | The Valuer-General may require the general manager or the public officer of a council to certify the accuracy and completeness of any information that the council provides under this section. | 22 23 24 | | |
| Divi | sion | 7 Miscellaneous | 25 | | |
| 69 | Rule | s about classification and sub-classification | 26 | | |
| | (1) | The Treasurer may, by order published on the NSW legislation website, issue rules about the circumstances in which land is or is not to be treated as having met any requirements for classification or sub-classification under this Part. | 27 28 29 | | |
| | (2) | Without limiting subsection (1), the rules may provide for the circumstances in which a use of land for a specified purpose is, or is not, to be treated as the dominant use of the land. | 30 31 32 | | |
| | (3) | A council must comply with the rules when exercising its functions under this Part. | 33 | | |
| 70 | Zoning and use changes | | | | |
| | | A council must review its classification of a parcel of land and its sub-classification (if any) if: | 35 36 | | |
| | | (a) the zoning or designated use for the land under an environmental planning instrument changes, or | 37 38 | | |
| | | (b) a change in use for the land is approved by the council. | 39 | | |
| 71 | Adju | stment of levy following change in classification or sub-classification | 40 | | |
| | (1) | A council must make an appropriate adjustment of any levy paid or payable, and to any instalments payable, by a liable person following a change in classification or sub-classification of land or a revocation of sub-classification. | 41 42 43 | | |

(2) If, at the effective date for a new classification or sub-classification, or a revocation of sub-classification, the levy for the whole year has been paid in full, the portion of the payment that is proportionate to the number of quarterly instalment periods remaining in the year is to be credited to the levy payable in respect of the remaining part of the year.

72 Changes deemed to take effect at end of quarterly instalment period

- (1) If circumstances occur during a financial year that justify a new classification, the sub-classification or the revocation of sub-classification of a person's land, those circumstances are to be treated by a council, for the purpose of setting an effective date for the new classification, sub-classification or revocation, as having occurred on the date that is the first date of the next quarterly instalment period after the quarterly instalment period during which the change actually occurred.
- (2) This section does not apply to a new classification that is made because of the making of a regulation that changes the requirements for classification of land. Note. A regulation that changes classification requirements may specify when it takes effect for classification purposes.

73 Notice of change of circumstances is required

- (1) A liable person must give notice to the council within 30 days of any change in circumstances that justifies:
 - (a) a new classification of the person's land, or
 - (b) the sub-classification of the person's land, or
 - (c) the revocation of sub-classification of the person's land.
- (2) If a liable person fails to give notice of a change in circumstances that would justify a new classification of the land in respect of which a lower levy would be payable, or the sub-classification of the land, within the period required by this section, the council may set an effective date for the classification or sub-classification that is the first day of the next quarterly instalment period after notification (regardless of when the change in circumstances occurred).
- (3) The next quarterly instalment period after notification is the next quarterly instalment period to occur after the quarterly instalment period during which the change in circumstances is notified to the council in accordance with this section.
- (4) A notice under this section must be given in a form and manner approved by the council.
- (5) A council may require the notice to be given in the form of an application for classification or sub-classification of the land.

74 Classification of mixed development land

- (1) If a valuation is furnished under the *Valuation of Land Act 1916* for mixed development land, a council may:
 - (a) classify the part of the land that is non-residential land as industrial land, commercial land or public benefit land, as the case requires (the *first classification*), and
 - (b) classify the remaining part of the land as residential land (the *second classification*).
- (2) In that case, the levy payable is the sum of the following amounts:
 - (a) the amount obtained by applying the apportionment factor to the levy that would be payable for the land if it were wholly within the first classification, 46

(b) the amount obtained by applying the remainder factor to the levy that would be payable for the land if it were wholly within the second classification.

- (3) The apportionment factor is the apportionment factor for the parcel ascertained under section 14X of the *Valuation of Land Act 1916*.
- (4) The remainder factor is the factor obtained by deducting the apportionment factor from 100%.
- (5) In this section, *mixed development land* and *non-residential land* have the same meanings as in Division 5 of Part 1B of the *Valuation of Land Act 1916*.

75 Parcel of land in 2 or more areas

- (1) If different parts of the same parcel of land are situated in different areas, each part of the parcel that is situated in a separate area is taken, for the purposes of this Act, to be a separate parcel of land.
- (2) Accordingly, a council must:
 - (a) classify, and sub-classify (if necessary), the part of the parcel of land that is situated in the council's area, as if that land were a separate parcel of land, and
 - (b) charge a levy in respect of the part of the parcel of land that is within the council's area, as if that land were a separate parcel of land.

Note. Section 28 of the *Valuation of Land Act 1916* provides for the apportionment of a land valuation between parcels of land that are situated in more than one district.

- (3) However, a council may, on application by a liable person, waive or refund part of the fixed component of the levy charged by the council in respect of a parcel of land if satisfied that the parcel of land is situated in more than one area and that, accordingly, more than one levy is payable in respect of the parcel.
- (4) The maximum amount that can be waived or refunded under this section is the amount obtained when the appropriate proportion is deducted from the fixed component of the levy.
- (5) The *appropriate proportion* is the amount that is obtained by dividing the fixed component of the levy that would be payable if the land were situated wholly in the council's area by the number of areas in which the parcel of land is situated.
- (6) This section is in addition to, and does not limit, the functions of a council under Part 8.

76 Powers of entry and inspection

To avoid doubt, Part 2 of Chapter 8 of the *Local Government Act 1993* applies to a council's functions under this Act.

Note. Part 2 of Chapter 8 of the *Local Government Act 1993* enables council employees to enter premises and conduct inspections for the purpose of enabling a council to exercise its functions. These powers could be exercised in connection with the council's classification functions under this Part or other functions under this Act.

| Par | t 6 | Charging and collection of levy by council | 1 |
|------|------|--|----------------|
| Divi | sion | 1 Charging of levy | 2 |
| 77 | Cour | ncil must charge levy | 3 |
| | | A council must charge a levy for each financial year on all leviable land in its area. | 4 |
| 78 | Levy | to be charged by notice | 5 |
| | (1) | A council is to charge a levy by serving notice of the levy on the liable person. | 6 |
| | (2) | The levy notice is to contain the following information: | 7 |
| | | (a) the land on which the levy is charged, | 8 |
| | | (b) the classification of the land, | 9 |
| | | (c) the sub-classification of the land (if any), | 10 |
| | | (d) the amount of the levy payable, | 11 |
| | | (e) the land value, for levy purposes, of the land to which it relates, | 12 |
| | | (f) the date of 1 July by reference to which the land is valued (unless the land is unvalued land), | 13 14 |
| | | (g) any other information specified in the regulations. | 15 |
| | (3) | The levy is to be described as the "NSW Government Fire and Emergency Services Levy" or as the "NSW Govt FESL". The description may include a reference to the property sector that the land is within. | 16 17 18 |
| | (4) | It is not necessary to specify the name of the liable person or the person liable to pay the charge in the notice if the council does not know the person's name. | 19 20 |
| | (5) | If rates and charges are payable in respect of the land under the <i>Local Government Act 1993</i> , the levy notice is to be combined with the notice of the rates or charges that is served under that Act, subject to any exceptions provided for by the regulations. | 21 22 23 |
| | (6) | Subject to the regulations, the levy notice may be combined with the following: | 24 |
| | | (a) notice of classification of the land under this Act, | 25 |
| | | (b) notice of categorisation of the land under the <i>Local Government Act 1993</i> . | 26 |
| | (7) | The regulations may make provision for the form of the levy notice and the manner in which information is to be set out in the levy notice. | 27 28 |
| 79 | Timi | ng for service of levy notice | 29 |
| | (1) | A levy notice may be served at any time after 1 July in the year for which the levy is charged or in a subsequent year. | 30 31 |
| | (2) | A levy notice that is required to effect an adjustment of levy may be served in the year for which the levy is charged or in a subsequent year. | 32 33 |
| 80 | Time | e for payment of levy | 34 |
| | (1) | The levy may be paid in a single instalment or by quarterly instalments, as provided for by this Part. | 35 36 |
| | (2) | If a rate or charge under Chapter 15 of the <i>Local Government Act 1993</i> payable for the same period as the levy is paid by single instalment, the levy must be paid by single instalment. | 37 38 39 |

| | (3) | If a rate or charge under Chapter 15 of the <i>Local Government Act 1993</i> payable for the same period as the levy is paid by quarterly instalments, the levy must be paid by quarterly instalments. | 1 2 3 |
|----|------|---|----------------|
| 81 | Payr | nent by single instalment | 4 |
| | (1) | If a levy is paid by single instalment, the instalment is payable by 31 August. | 5 |
| | (2) | However, if the levy notice is not served by 1 August, the single instalment is payable by 30 November, or the day that is 30 days after service of the notice, whichever is the later. | 6 7 8 |
| 82 | Payr | nent by quarterly instalments | 9 |
| | (1) | If the levy is paid by quarterly instalments, a quarterly instalment is payable for each of the following periods (each of which is a <i>quarterly instalment period</i>): | 10 11 |
| | | (a) the period starting on 1 July and ending on the next 30 September, | 12 |
| | | (b) the period starting on 1 October and ending on the next 31 December, | 13 |
| | | (c) the period starting on 1 January and ending on the next 31 March, | 14 |
| | | (d) the period starting on 1 April and ending on the next 30 June. | 15 |
| | (2) | The start dates and end dates of a quarterly instalment period are inclusive. | 16 |
| | (3) | A quarterly instalment is payable by the date of 31 August, 30 November, 28 February or 31 May that falls within the quarterly instalment period for which the instalment is payable. | 17 18 19 |
| | (4) | However, if the levy notice is not served by 1 August, the first 2 quarterly instalments are payable by 30 November, or by the day that is 30 days after service of the notice, whichever is the later. | 20 21 22 |
| | (5) | Section 562 (2) of the <i>Local Government Act 1993</i> applies in relation to the calculation of quarterly instalments of the levy in the same way as it applies in relation to the calculation of quarterly instalments of a rate or charge under that Act. | 23 24 25 |
| | (6) | On or before 31 October, 31 January and 30 April, a council must send reminder notices to each person whose levy is being paid by quarterly instalments. | 26 27 |
| | (7) | The reminder notice may be sent in conjunction with a reminder notice relating to rates and charges under the <i>Local Government Act 1993</i> but must be sent separately from a levy notice. | 28 29 30 |
| 83 | Асси | rual of interest on overdue levies | 31 |
| | (1) | Interest accrues on any levy or part of a levy that remains unpaid after it becomes due and payable. | 32 33 |
| | (2) | Interest accrues on a daily basis. | 34 |
| | (3) | The rate of interest is that set by the council but must not exceed the rate specified for overdue rates and charges by the Minister administering the <i>Local Government Act 1993</i> under section 566 of that Act. | 35 36 37 |
| | (4) | If the council charges interest on overdue rates and charges under the <i>Local Government Act 1993</i> , the rate of interest set by the council must be the same as the rate of interest charged on overdue rates and charges. | 38 39 40 |
| | (5) | Accrued interest is, for the purpose of its recovery, taken to be part of the levy that is due and payable. | 41 42 |

| | (6) | Interest continues to accrue on an unpaid levy or part of a levy even though judgment for payment of the levy may have been obtained in a court. Interest is not payable on the amount of the judgment (exclusive of any order for costs), despite any other Act. | 1 2 3 |
|------|------|---|----------------------|
| Divi | sion | 2 Waiver or reduction of levy | 4 |
| 84 | Harc | dship guidelines | 5 |
| | (1) | The Treasurer may, by order published in the Gazette, approve guidelines relating to the waiver or reduction of the levy because of hardship. | 6 7 |
| | (2) | The guidelines published under this section are <i>hardship guidelines</i> . | 8 |
| 85 | Chie | f Commissioner may waive or reduce levy | 9 |
| | (1) | The Chief Commissioner may waive or reduce the levy payable by any person in accordance with the hardship guidelines. | 10 11 |
| | (2) | If the Chief Commissioner decides to waive or reduce the levy payable by a person, the Chief Commissioner must give notice of that decision to the council responsible for recovering the levy. | 12 13 14 |
| | (3) | The council is required to give effect to that decision by: | 15 |
| | | (a) waiving or reducing the levy payable, as the case requires, and | 16 |
| | | (b) writing off the amount that is waived or the amount of the reduction. | 17 |
| | (4) | A council may not take proceedings to recover an amount written off under this section. | 18 19 |
| | (5) | If a levy that is waived or reduced is subject to a levy recovery arrangement, subsections (3) and (4) apply to the Chief Commissioner in the same way as they apply to the council. | 20 21 22 |
| | (6) | This section does not limit the power of a council to waive or reduce the levy payable by a person under the provisions of the <i>Local Government Act 1993</i> that apply to the levy under Part 8. | 23 24 25 |
| 86 | Liab | le person required to make full and true disclosure | 26 |
| | (1) | A liable person who makes a claim for a waiver or reduction of a levy by the Chief Commissioner must fully and truly disclose to the Chief Commissioner all the facts and circumstances relating to that claim. Maximum penalty: 100 penalty units. | 27 28 29 30 |
| | (2) | The Chief Commissioner may, by order in writing served on a person (a <i>payment</i> | 31 |
| | | order), require a person to pay to the Chief Commissioner a specified amount if: | 32 |
| | | (a) the Chief Commissioner waived or reduced the levy payable by the person, and | 33 34 |
| | | (b) the Chief Commissioner is satisfied that the person contravened subsection (1) in relation to the person's claim for a waiver or reduction of the levy. | 35 36 |
| | (3) | A payment order may require the person to pay the whole or part of a waived amount to the Chief Commissioner by a date specified in the order (being not less than 30 days after the date of service of the order). | 37 38 39 |
| | (4) | A <i>waived amount</i> is the amount of the levy that was waived or, if the levy was reduced, the amount of the reduction. | 40 41 |
| | (5) | A payment order may require the person to pay interest in respect of the waived amount, calculated as provided for by this Division. | 42 43 |

| (7) A payment order cannot be made more than 3 years after subsection (1) is alleged to have occurred. (8) In this section: <i>Chief Commissioner</i> includes a delegate of the Chief Commission engaged in the administration of this Act. 87 Application of Taxation Administration Act 1996 The applied provisions of the Taxation Administration Act 1995 | 4 ssioner and any other 6 7 |
|---|-----------------------------------|
| <i>Chief Commissioner</i> includes a delegate of the Chief Commission person engaged in the administration of this Act. 87 Application of Taxation Administration Act 1996 The applied provisions of the <i>Taxation Administration Act 199</i> | ssioner and any other 6 7 |
| <i>Chief Commissioner</i> includes a delegate of the Chief Commission person engaged in the administration of this Act. 87 Application of Taxation Administration Act 1996 The applied provisions of the <i>Taxation Administration Act 199</i> | 7 |
| (1) The applied provisions of the <i>Taxation Administration Act</i> 199 | |
| | 8 |
| a payment order as if this Act were a taxation law and the amore order were a tax payable by the person to the Chief Commission | int payable under the 10 |
| (2) The <i>applied provisions</i> of the <i>Taxation Administration Act 19</i> provisions of that Act: | 96 are the following 12 13 |
| (a) Division 1 (Interest) of Part 5, | 14 |
| (b) Division 1 of Part 7, | 15 |
| (c) sections 50 and 55 (which prohibit the giving of f information), | alse and misleading 16 17 |
| (d) Division 1 (Tax officers), Division 2 (Investigation) and of Part 9, | Division 3 (Secrecy) 18 19 |
| (e) sections 114, 115, 116, 118 and 120, | 20 |
| (f) any provisions that are relevant to the interpretation of the | ne above provisions. 21 |
| (3) For the purposes of the applied provisions, a tax default is taken the date the Chief Commissioner waived or reduced the le Commissioner decides that a later date should be treated as the | vy, unless the Chief 23 |
| (4) If the Chief Commissioner decides that a later date should be tre date, a tax default does not occur until that later date. | ated as the tax default 25 26 |
| (5) To avoid doubt, Part 10 of the <i>Taxation Administration Act 199</i> payment order or to the exercise of any of the Chief Commission this Division (including under the applied provisions of the <i>Tax Act 1996</i>). | ner's functions under 28 |
| 88 References to levy include interest on levy | 31 |
| (1) Any accrued interest on the levy may be waived or reduced und same way as the levy. | er this Division in the 32 33 |
| (2) Accordingly, a reference in this Division to the levy includes a interest on the levy. | reference to accrued 34 35 |
| Division 3 Recovery of levy | 36 |
| 89 Recovery of levy as debt | 37 |
| (1) A levy payable to a council under this Act may be recovered by in a court of competent jurisdiction. | the council as a debt 38 39 |
| (2) The levy has the same priority as rates or charges that are ower the <i>Local Government Act 1993</i> . | l to the council under 40 41 |
| Note. Part 8 gives the council the same remedies for the recovery of has for rates and charges. | the levy as the council 42 43 |

| 90 | Levies paid to be kept in council's consolidated fund | | | | |
|----|---|--|---|----------------------|--|
| | (1) | | ey received by a council in payment of the levy is to be held in, and forms part ne council's consolidated fund. | 2 3 | |
| | (2) | A co | uncil may retain any proceeds of investment of the levy money. | 4 | |
| | (3) | | section does not affect the requirement that the council pay collection lments under this Act. | 5 6 | |
| 91 | Tran | sfer o | f levy recovery to Chief Commissioner | 7 | |
| | (1) | A council may enter into an arrangement (a <i>levy recovery arrangement</i>) with the Chief Commissioner that enables the council to refer to the Chief Commissioner, for recovery by the Chief Commissioner, any levy, part of a levy or class of levies, that is payable to the council in respect of land that is not subject to a rate or charge under the <i>Local Government Act 1993</i> . | | | |
| | (2) | | levy, part of a levy or class of levies (a <i>referred levy</i>) is referred to the Chief missioner under a levy recovery arrangement, on the transfer date: | 13 14 | |
| | | (a) | the referred levy (and any interest that accrued before the transfer date) becomes an amount payable to the Chief Commissioner (not to the council), and | 15 16 17 | |
| | | (b) | the referred levy ceases to be an assessed levy of the council, and | 18 | |
| | | (c) | the applied provisions of the <i>Taxation Administration Act 1996</i> apply to the referred levy as if it were a tax under a taxation law payable by the person liable for the levy. | 19 20 21 | |
| | (3) | provi date, | ilure to pay a referred levy is a tax default for the purposes of the applied isions of the <i>Taxation Administration Act 1996</i> and, on and from the transfer interest on any overdue amount accrues in accordance with that Act (and not at ate set by the council). | 22 23 24 25 | |
| | (4) | | e referred levy became overdue before the transfer date, a tax default is taken to r, for the purposes of the applied provisions, on the transfer date. | 26 27 | |
| | (5) | | <i>applied provisions</i> of the <i>Taxation Administration Act 1996</i> are the following isions of that Act: | 28 29 | |
| | | (a) | Division 1 (Interest) of Part 5, | 30 | |
| | | (b) | Division 1 of Part 7, | 31 | |
| | | (c) | sections 50 and 55 (which prohibit the giving of false and misleading information), | 32 33 | |
| | | (d) | Division 1 (Tax officers), Division 2 (Investigation) and Division 3 (Secrecy) of Part 9, | 34 35 | |
| | | (e) | sections 114, 115, 116, 118 and 120, | 36 | |
| | (| (f) | any provisions that are relevant to the interpretation of the above provisions. | 37 38 | |
| | (6) | To avoid doubt, Part 10 of the <i>Taxation Administration Act 1996</i> does not apply to a referred levy or to the exercise of any of the Chief Commissioner's functions under a levy recovery arrangement (including under the applied provisions of the <i>Taxation Administration Act 1996</i>). | | | |
| | (7) | | amount recovered by the Chief Commissioner under a debt recovery agement: | 42 43 | |
| | | (a) | is payable to the Consolidated Fund (not to the council), and | 44 | |
| | | (b) | is not to be included in the council's relevant payments under Part 7. | 45 | |

| (8) | Section 571 of the <i>Local Government Act 1993</i> , as it applies to the levy under Part 8, continues to apply in respect of a referred levy as if a reference in that section to the council were a reference to the Chief Commissioner. | 1 2 3 | |
|-------|--|------------------|--|
| (9) | The regulations may make further provision for the application of the <i>Local Government Act 1993</i> in respect of a referred levy, including by modifying the application of any provision of that Act that applies to a levy that is payable to the council. | 4 5 6 7 | |
| (10) | In this section: | 8 | |
| | <i>transfer date</i> means the date agreed in writing between a council and the Chief Commissioner as the date on which a referred levy is referred to the Chief Commissioner for recovery. | 9 10 11 | |
| Reco | ord of levies | 12 | |
| (1) | A council is required to keep a record of the following: | 13 | |
| | (a) each levy charged by it, | 14 | |
| | (b) the land value, for levy purposes, of each parcel of land on which the levy is charged, | 15 16 | |
| | (c) the classification and any sub-classification of the land, | 17 | |
| | (d) the liable person for each parcel of land (if known). | 18 | |
| (2) | The council may amend the record at any time. | 19 | |
| Colle | ection payments | 20 | |
| (1) | The Treasurer is to make payments to councils to reimburse the reasonable costs incurred by councils that are associated with any of the following: | 21 22 | |
| | (a) collecting and recovering the levy on behalf of the State, | 23 | |
| | (b) responding to, or defending, appeals in the Land and Environment Court against decisions made by the council under this Act, | 24 25 | |
| | (c) otherwise implementing or administering this Act. | 26 | |
| (2) | Payments are to be made from the Consolidated Fund out of money provided by Parliament. | 27 28 | |
| (3) | A payment made by the Treasurer under this section is a <i>collection payment</i> . | | |

| Par | t 7 | Pay | ment of levy to State | 1 | |
|------|------|--|---|----------------|--|
| Divi | sion | 1 | Preliminary | 2 | |
| 94 | Cou | ouncil required to apportion relevant payments | | | |
| | | A co | uncil is to: | 4 | |
| | | (a) | apply all relevant payments made to the council in accordance with this Part, and | 5 6 | |
| | | (b) | pay a proportion of those payments to the Chief Commissioner in accordance with this Part. | 7 8 | |
| 95 | Rele | vant p | ayment—meaning | 9 | |
| | (1) | | <i>levant payment</i> means a payment made to a council in or towards payment of or all of the following: | 10 11 | |
| | | (a) | an assessed rate or charge, | 12 | |
| | | (b) | an assessed levy. | 13 | |
| | (2) | A pa coun | yment is made to a council when it is received, recovered or accepted by the cil. | 14 15 | |
| | (3) | unde | ayment includes the value of any capital contribution received by the council or section 565 of the <i>Local Government Act 1993</i> in payment of an assessed rate marge. | 16 17 18 | |
| 96 | Asse | essed | rate or charge—meaning | 19 | |
| | (1) | | ussessed rate or charge is a rate or charge that has been levied by the council or Chapter 15 of the <i>Local Government Act 1993</i> . | 20 21 | |
| | (2) | | ssessed rate or charge includes any amount payable as accrued interest on an ssed rate or charge. | 22 23 | |
| | (3) | An a | ssessed rate or charge does not include: | 24 | |
| | | (a) | any amount payable for the out-of-pocket expenses incurred by the council in tracing persons that may be recovered as a rate or charge under the <i>Local Government Act 1993</i> , or | 25 26 27 | |
| | | (b) | a rate or charge, or a part of a rate or charge, that has been written off or the payment of which has been waived by the council. | 28 29 | |
| | (4) | | rate or charge is reduced by a council, the reduction is not to be included in an assed rate or charge. | 30 31 | |
| | (5) | | e payment of any part of a rate is postponed under Division 2 of Part 8 of oter 15 of the <i>Local Government Act 1993</i> : | 32 33 | |
| | | (a) | the postponed rate is not an assessed rate or charge, during the period in which payment is postponed, and | 34 35 | |
| | | (b) | any interest on the postponed rate, that accrues under section 592 of that Act, is taken to be an assessed rate or charge. | 36 37 | |
| | (6) | | regulations may make further provision for the calculation of an assessed rate or ge, including by modifying the application of any provision of this section. | 38 39 | |
| | (7) | unles | Minister is not to recommend the making of a regulation under this section as the Minister certifies that the regulation is made with the concurrence of the ester administering the <i>Local Government Act 1993</i> . | 40 41 42 | |

| 97 | Assessed levy—meaning | | | |
|------|-----------------------|-----------------|--|----------------|
| | (1) | An <i>a</i> | ssessed levy is a levy that has been charged by the council under this Act. | 2 |
| | (2) | An a levy. | ssessed levy includes any amount payable as accrued interest on an assessed | 3 4 |
| | (3) | | ssessed levy does not include a levy, or a part of a levy, that has been written off e payment of which has been waived by the council. | 5 6 |
| | (4) | If a l levy. | evy is reduced by a council, the reduction is not to be included in an assessed | 7 8 |
| | (5) | | regulations may make further provision for the calculation of an assessed levy, ding by modifying the application of any provision of this section. | 9 10 |
| | (6) | unles | Minister is not to recommend the making of a regulation under this section is the Minister certifies that the regulation is made with the concurrence of the ster administering the <i>Local Government Act 1993</i> . | 11 12 13 |
| Divi | ision | 2 | Payment of collection instalments to State | 14 |
| 98 | Rele | vant p | ayments to be applied towards old debt first | 15 |
| | (1) | charg | uncil is to apply relevant payments towards the payment of the assessed rates or ges and assessed levies in respect of which the payment was made in the order nich they became due. | 16 17 18 |
| | (2) | relev | he purposes of this Part, a council applies a relevant payment (or a part of a ant payment) <i>towards a financial year</i> if the council applies the relevant nent (or part): | 19 20 21 |
| | | (a) | towards payment of an assessed rate or charge, or an assessed levy, that became due in that financial year, or | 22 23 |
| | | (b) | towards payment of accrued interest on an assessed rate or charge, or an assessed levy, and the assessed rate or charge or assessed levy became due in that financial year. | 24 25 26 |
| 99 | Colle | ection | instalments to be paid to Chief Commissioner | 27 |
| | (1) | A co | uncil is to pay collection instalments to the Chief Commissioner. | 28 |
| | (2) | | collection instalments are payable for any relevant payments that are applied by incil towards the 2017/2018 financial year or a subsequent financial year. | 29 30 |
| | (3) | A co | llection instalment is payable in respect of each reconciliation period. | 31 |
| 100 | Reco | oncilia | tion periods | 32 |
| | (1) | Each | of the following periods is a <i>reconciliation period</i> : | 33 |
| | | (a) | the first quarter reconciliation period—which is the period starting on 1 June and ending on 31 August in the same year, | 34 35 |
| | | (b) | the second quarter reconciliation period—which is the period starting on 1 September and ending on 30 November in the same year, | 36 37 |
| | | (c) | the third quarter reconciliation period—which is the period starting on 1 December and ending on 28 February in the next year (or 29 February in a leap year), | 38 39 40 |
| | | (d) | the fourth quarter reconciliation period—which is the period starting on 1 March and ending on 31 May in the same year. | 41 42 |
| | (2) | The s | start dates and end dates of a reconciliation period are inclusive. | 43 |

| (3) | For the 2017/2018 financial year, the first quarter reconciliation period is taken to |
|-----|---|
| | start on 1 July 2017. |

101 Amount of collection instalment

- (1) The amount of the collection instalment payable is the total of the following:
 - (a) the levy proportion of any relevant payments made to the council during the reconciliation period that were applied towards the financial year in which the reconciliation period ends,

1 2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

29

30

39

40

41

(b) the levy proportion of any relevant payments made to the council during the reconciliation period that were applied towards a previous financial year.

Note. Councils may recover overdue amounts of rates and charges, or levies, over time. Accordingly, this provision ensures that they are still required to account to the Treasurer for the levy component of amounts recovered even when they are recovered after the financial year in which they are due.

(2) The levy proportion of any relevant payments that are applied towards a financial year that commences before the 2017/2018 financial year is taken to be zero.

102 Calculation of levy proportion

(1) The *levy proportion* of any relevant payments made to a council that are applied towards a financial year is to be calculated as follows:

$$L = C \times \frac{B}{(A+B)}$$

where:

L is the levy proportion of the relevant payments.

C is the total of all relevant payments made to the council during the reconciliation period that were applied by the council towards the financial year for which the calculation is being made.

B is the total of all assessed levies charged by the council for the financial year towards which the relevant payments are applied.

A is the total of all assessed rates or charges levied by the council for the financial year towards which the relevant payments are applied.

(2) The levy proportion is to be rounded to the nearest cent (with an amount of half a cent rounded up).

103 Payment dates for reconciliation periods

- A council must pay a collection instalment to the Chief Commissioner on or before (1)31 the payment date for the reconciliation period for which the instalment is payable. 32 (2)The payment dates are as follows: 33 for the first quarter reconciliation period—30 September, (a) 34 (b) for the second quarter reconciliation period—31 December, 35 (c)for the third quarter reconciliation period—31 March, 36 (d) for the fourth quarter reconciliation period—30 June. 37 Council to keep a record of relevant payments 38
- (1) A council is to keep a record of all relevant payments that are made to the council and the financial year towards which those relevant payments are applied.
- (2) The record is to contain any other information required by the regulations.

| Divi | sion | Functions of Chief Commissioner | 1 |
|------|--|--|----------------------|
| 105 | Chie | Commissioner may recover collection instalment from council | 2 |
| | | If the whole or part of a collection instalment payable by a council is not paid to the Chief Commissioner as required, the unpaid amount is recoverable by the Chief Commissioner from the council as a debt in a court of competent jurisdiction. | 3 4 5 |
| 106 | Appl | cation of certain provisions of Taxation Administration Act 1996 | 6 |
| | (1) | The applied provisions of the <i>Taxation Administration Act 1996</i> apply in relation to the collection instalments payable by a council under this Act as if this Act were a taxation law and the collection instalments were a tax payable by a council under a taxation law. | 7 8 9 10 |
| | (2) | A tax default occurs if the whole or part of a collection instalment is not paid by the due date. | 11 12 |
| | (3) | The <i>applied provisions</i> of the <i>Taxation Administration Act 1996</i> are the following provisions of that Act: | 13 14 |
| | | (a) Part 3 (Assessment of tax liability), | 15 |
| | | (b) Division 1 (Interest) of Part 5, | 16 |
| | | (c) Division 1 of Part 6 (Returns), | 17 |
| | | (d) sections 50 and 55 (which prohibit the giving of false and misleading information), | 18 19 |
| | | (e) Division 1 (Tax officers), Division 2 (Investigation) and Division 3 (Secrecy) of Part 9, | 20 21 |
| | | (f) sections 114, 115, 116, 118, 119 and 120, | 22 |
| | | (g) any provisions that are relevant to the interpretation of the above provisions. | 23 |
| | (4) | A council is liable to pay collection instalments to the Chief Commissioner whether or not the Chief Commissioner has issued an assessment for the amount payable. | 24 25 |
| | (5) | To avoid doubt, Part 10 of the <i>Taxation Administration Act 1996</i> does not apply to a collection instalment or to the exercise of any of the Chief Commissioner's functions under this Division (including under the applied provisions of the <i>Taxation Administration Act 1996</i>). | 26 27 28 29 |
| 107 | Cour | cils to provide annual return to Chief Commissioner | 30 |
| | (1) Each council is to provide to the Chief Commissioner, on or before 31 July in each year, a return for the financial year ending on 30 June that year relating to the council's liability for collection instalments under this Part. | | 31 32 33 |
| | (2) | The return is to contain the following information about the financial year for which it is made: | 34 35 |
| | | (a) information about the council's liability for collection instalments under this Part, including any information specified in the regulations, | 36 37 |
| | | (b) any other information that the Chief Commissioner reasonably requires for the purpose of exercising his or her functions under this Act. | 38 39 |
| | (3) | The return is to be in a form approved by the Chief Commissioner. | 40 |
| | (4) | The regulations may make further provision for returns under this section. | 41 |
| | (5) | Without limiting subsection (4), the regulations may require the general manager or the public officer of a council to certify the accuracy and completeness of information contained in the return. | 42 43 44 |

108 Provision of other information to Chief Commissioner

(1) The regulations may require councils to provide further information or returns to the Chief Commissioner for the purposes of this Act.

1

2 3

4

5

6

(2) This section does not limit the functions of the Chief Commissioner under Division 2 of Part 9 of the *Taxation Administration Act 1996* as it applies to the collection instalments payable by a council.

Part 8 Harmonisation with council rates and charges

109 Strata lots and company titles treated as separate parcels

(1) Section 495A of the *Local Government Act 1993* applies in respect of the charging of a levy under this Act in the same way as it applies in respect of the levying of an annual charge under that Act.

- (2) Anything that is taken, under that section, to be a separate parcel of rateable land is taken to be a separate parcel of land under this Act.
- (3) Section 518A of the *Local Government Act 1993* applies to the classification and sub-classification of land under this Act in the same way as it applies to the categorisation of land under Part 3 of Chapter 15 of that Act.
- (4) Anything that is taken, under that section, to be a separate parcel of land for the purposes of categorisation is taken to be a separate parcel of land for the purposes of classification or sub-classification under this Act.

110 Application of Act to company title

- (1) Section 547 of the *Local Government Act 1993* applies for the purposes of this Act in the same way as it applies for the purposes of that Act.
- (2) For that purpose, a reference in that section to a rate or charge includes a reference to the levy.

111 Aggregation of parcels

- (1) If a council aggregates the land values of parcels under section 548A of the *Local Government Act 1993* for the purpose of charging a rate, the council must also aggregate the land values of the parcels for the purposes of charging the levy, subject to this section.
- (2) The parcels as aggregated are taken to be a separate parcel for the purposes of this Act and the levy is to be charged on the aggregated land values of those parcels.
- (3) The land values of parcels may be aggregated under this section for the purpose of charging the levy only if the parcels (disregarding the aggregation) have the same classification or sub-classification (if any) under this Act.

112 Levy is a charge on land

- (1) Section 550 of the *Local Government Act 1993* applies to the levy in the same way as it applies to a rate or charge levied under that Act.
- (2) A reference in that section to interest on a rate or charge includes a reference to interest on the levy accrued under this Act.

113 Writing off levy and accrued interest

- (1) Section 567 of the *Local Government Act 1993*, and any regulations under section 607 of that Act, apply to the levy in the same way as they apply to rates and charges under that Act.
- (2) However, a council is not to write off any levy or any accrued interest on a levy payable by a person unless it writes off rates or charges or accrued interest on rates or charges payable by the person in the same proportion.

114 Collection of levy from occupier

Section 569 of the *Local Government Act 1993* applies in respect of the levy in the same way as it applies in respect of a rate or charge under that Act.

| 115 | Tran | sfer of land to pay levy | 1 |
|-----|------|---|----------------|
| | (1) | Section 570 of the <i>Local Government Act 1993</i> applies in relation to the levy in the same way as it applies in relation to rates or charges under that Act. | 2 3 |
| | (2) | If a council accepts land in payment of any rates or charges, or levies, or accrued interest, the council is taken to have received the full amount of any assessed rates or charges, or assessed levies, payable in respect of that land. | 4 5 6 |
| 116 | Tran | sfer of land | 7 |
| | | Section 571 of the <i>Local Government Act 1993</i> applies in relation to the levy in the same way as it applies in relation to rates or charges under that Act. | 8 9 |
| 117 | Sub | division of land | 10 |
| | | Section 573 of the <i>Local Government Act 1993</i> applies in relation to the levy in the same way as it applies in relation to rates or charges under that Act. | 11 12 |
| 118 | Defi | nition of "eligible pensioner" | 13 |
| | | Sections 577–579 of the <i>Local Government Act 1993</i> apply for the purposes of this Act in the same way as they apply for the purposes of Division 1 of Part 8 of Chapter 15 of that Act. | 14 15 16 |
| 119 | Waiv | ring payment of levy | 17 |
| | (1) | Section 582 of the <i>Local Government Act 1993</i> applies to the levy in the same way as it applies to rates or charges under that Act. | 18 19 |
| | (2) | However, a council is not to waive or reduce the levy payable by a person under that section, as it applies to the levy, unless it waives or reduces the rates or charges payable by the person for the same period in the same proportion. | 20 21 22 |
| | (3) | Section 583 of the <i>Local Government Act 1993</i> applies to amounts of the levy that are waived or reduced under this section in the same way as it applies to rates or charges under that Act. | 23 24 25 |
| | (4) | In this section: | 26 |
| | | <i>levy</i> includes any interest payable on a levy. <i>rates or charges</i> includes any interest payable on any rates or charges. | 27 28 |
| 120 | Cert | ificate as to levy | 29 |
| | | A reference in section 603 of the <i>Local Government Act 1993</i> to rates, charges or other amounts includes a reference to the levy. | 30 31 |
| 121 | Noti | ce of transfer of land | 32 |
| | (1) | Section 604 of the <i>Local Government Act 1993</i> applies to leviable land in the same way as it applies to rateable land. | 33 34 |
| | (2) | A person who gives notice as required by section 604 of that Act in relation to rateable land that is also leviable land is taken to have complied with section 604 as it applies to leviable land. | 35 36 37 |
| 122 | Noti | ces | 38 |
| | (1) | Sections 699 and 708–711 of the <i>Local Government Act 1993</i> apply to notices served by a council under this Act in the same way as they apply to the notices served by a council under that Act. | 39 40 41 |
| | (2) | In this section, serve includes give or send. | 42 |
| | | | |

123 Proof of ownership

Section 700 of the *Local Government Act 1993* applies to the levy, and to proceedings under this Act, in the same way as it applies to rates and charges, and to proceedings under that Act.

124 Recovery of levy

- (1) Section 712 of the *Local Government Act 1993* applies to the levy in the same way as it applies to rates and charges under that Act.
- (2) For that purpose:
 - (a) a reference in that section to a rate or charge includes a reference to the levy, and
 - (b) a reference in that section to a rates and charges notice includes a reference to a levy notice.

125 Sale of land to recover levy

- (1) Division 5 of Part 2 of Chapter 17 of the *Local Government Act 1993* applies to the levy in the same way as it applies to a rate or charge.
- (2) To avoid doubt, a reference to vacant land in those provisions is a reference to vacant land within the meaning of that Act.

126 Regulations relating to rates and charges

- (1) Any provision of a regulation under the *Local Government Act 1993* applies to the levy in the same way as it applies to a rate or charge if the regulation is made under, or ancillary to, a provision of that Act that applies to the levy.
- (2) This section is subject to the regulations under this Act.

127 Application of strata provisions

- (1) Section 192 of the *Strata Schemes Development Act 2015* applies to the charging of the levy under this Act in the same way as it applies to the levying of a rate under the *Local Government Act 1993*.
- (2) For the purposes of the levy, a reference in section 192 of the *Strata Schemes Development Act 2015*:
 - (a) to a lot that is rateable is a reference to a lot for which the levy is payable, and
 - (b) to a rateable parcel or a rateable part of a parcel is a reference to a parcel or part of a parcel in respect of which the levy is payable or would be payable but for any exemption or discount applicable to any portion of that parcel or part.

128 Power to modify and apply further legislative provisions

- (1) The regulations may modify the application of, or disapply, any provision of the *Local Government Act 1993*, a regulation under that Act or the *Strata Schemes Development Act 2015* that applies in relation to the levy under this Act.
- (2) The regulations may apply, with or without modification, to or in respect of the levy any other provisions of the *Local Government Act 1993*, the regulations under that Act or the *Strata Schemes Development Act 2015* that relate to rates or charges under the *Local Government Act 1993*.

129 References to council extend to Lord Howe Island Board

(1) To avoid doubt, a reference to a council in a provision of the *Local Government Act 1993*, or the regulations under that Act, that applies to the levy under this Act is taken to include a reference to the Lord Howe Island Board.

1

2 3 4

5

(2) This section is subject to the regulations under this Act.

| Part 9 Division 1 | | • | | 1 |
|----------------------|--------|-------------------------|--|----------------------|
| | | | | 2 |
| 130 | Defin | itions | | 3 |
| | (1) | In this | s Part: | 4 |
| | | breac | h of this Act means: | 5 |
| | | (a) | a contravention of or failure to comply with this Act, or | 6 |
| | | (b) | a threatened or an apprehended contravention of or a threatened or apprehended failure to comply with this Act. | 7 8 |
| | | <i>classi</i> Part 4 | <i>fication obligation</i> means any function conferred or imposed on a council by or 5. | 9 10 |
| | | finan | cial obligation means any of the following functions of the council: | 11 |
| | | (a) | a function conferred or imposed by Part 3, 6 or 7, | 12 |
| | | (b) | a function under Part 8 that relates to the charging and collection of the levy. | 13 |
| | (2) | In this | s Part, to comply with an obligation includes to properly exercise a function. | 14 |
| | (3) | | s Part, a reference to this Act or to a part of this Act includes a reference to any ations, rules or other instruments made under this Act or part, as the case res. | 15 16 17 |
| Divis | sion 2 | 2 | Monitoring role of Valuer-General | 18 |
| 131 | Moni | toring | of council compliance with classification obligations | 19 |
| | | | Valuer-General is to monitor whether councils are complying with their fication obligations. | 20 21 |
| 132 | Revie | ews by | v Valuer-General | 22 |
| | (1) | | /aluer-General may, at the Valuer-General's expense, conduct a review of any following: | 23 24 |
| | | (a) | the compliance by a council with its classification obligations, | 25 |
| | | (b) | the accuracy and completeness of information provided by a council to the Valuer-General under this Act. | 26 27 |
| | (2) | the V | ancil must co-operate with a review under this section, including by providing faluer-General, or a person authorised by the Valuer-General, with access to cil facilities and records during regular business hours. | 28 29 30 |
| 133 | Value | er-Gen | eral may issue compliance certificate | 31 |
| | (1) | initiat assess | Valuer-General may, at the request of the Treasurer or on his or her own tive, give to the Treasurer a certificate, in writing, about the Valuer-General's sment of a council's compliance with any of its classification obligations (a <i>liance certificate</i>). | 32 33 34 35 |
| | (2) | | /aluer-General may, in a compliance certificate, give an opinion as to whether t the council is complying with any of its classification obligations. | 36 37 |
| | (3) | evide obliga | Valuer-General considers that he or she does not have sufficient information or nce to be satisfied that the council is complying with any of its classification ations, the Valuer-General may state in the certificate that he or she is unable to an opinion on the matter, for that reason. | 38 39 40 41 |

- (4)The compliance certificate is to specify any compliance issues identified by the Valuer-General.
- (5) A compliance issue is any fact or circumstance that:
 - forms the basis of the Valuer-General's opinion that the council is not (a) complying with any of its classification obligations, or

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

- because of which, the Valuer-General considers that he or she does not have (b) sufficient information or evidence to be satisfied that the council is complying with any of its classification obligations.
- Before issuing a compliance certificate that specifies compliance issues, the (6) Valuer-General must:
 - give the council an opportunity to make submissions in relation to the (a) compliance issues within a period (of not less than 30 days) specified by the Valuer-General, and
 - (b) take into consideration any submissions made by the council during that period.
- The Valuer-General is to provide a copy of a compliance certificate issued by the (7)Valuer-General to the council to which it relates.

| | (8) | A compliance certificate is admissible in any legal proceedings and is evidence of any compliance issues specified in the certificate. | 18 19 |
|------|------|---|----------------------|
| Divi | sion | 3 Monitoring role of Chief Commissioner | 20 |
| 134 | Moni | toring of council compliance with financial obligations | 21 |
| | | The Chief Commissioner is to monitor whether councils are complying with their financial obligations. | 22 23 |
| 135 | Audi | ts by Chief Commissioner | 24 |
| | (1) | The Chief Commissioner may, at the Chief Commissioner's expense, carry out an audit of any of the following: | 25 26 |
| | | (a) the compliance by a council with its financial obligations, | 27 |
| | | (b) the accuracy and completeness of information provided by a council to the Chief Commissioner under this Act. | 28 29 |
| | (2) | A council must co-operate with an audit under this section, including by providing the Chief Commissioner, or a person authorised by the Chief Commissioner, with access to council facilities and records during regular business hours. | 30 31 32 |
| 136 | Chie | f Commissioner may issue compliance certificate | 33 |
| | (1) | The Chief Commissioner may, at the request of the Treasurer or on his or her own initiative, give to the Treasurer a certificate, in writing, about the Chief Commissioner's assessment of a council's compliance with any of its financial obligations (a <i>compliance certificate</i>). | 34 35 36 37 |
| | (2) | The Chief Commissioner may, in a compliance certificate, give an opinion as to whether or not the council is complying with any of its financial obligations. | 38 39 |
| | (3) | If the Chief Commissioner considers that he or she does not have sufficient information or evidence to be satisfied that the council is complying with any of its financial obligations, the Chief Commissioner may state in the certificate that he or she is unable to give an opinion on the matter, for that reason. | 40 41 42 43 |

- (4) The compliance certificate is to specify any compliance issues identified by the Chief Commissioner.
- (5) A compliance issue is any fact or circumstance that:
 - (a) forms the basis of the Chief Commissioner's opinion that the council is not complying with any of its financial obligations, or

- (b) because of which, the Chief Commissioner considers that he or she does not have sufficient information or evidence to be satisfied that the council is complying with any of its financial obligations.
- (6) Before issuing a compliance certificate that specifies compliance issues, the Chief Commissioner must:
 - (a) give the council an opportunity to make submissions in relation to the compliance issues within a period (of not less than 30 days) specified by the Chief Commissioner, and
 - (b) take into consideration any submissions made by the council during that period.
- (7) The Chief Commissioner is to provide a copy of a compliance certificate issued by the Chief Commissioner to the council to which it relates.
- (8) A compliance certificate is admissible in any legal proceedings and is evidence of any compliance issues specified in the certificate.

Division 4 Treasurer's enforcement powers

137 Order by Treasurer

- (1) The Treasurer may, by order in writing served on a council, require the council to take any action that, in the opinion of the Treasurer, is necessary to remedy or restrain a breach of this Act by the council.
- (2) An action includes refraining from doing something.
- (3) Without limiting subsection (1), the Treasurer may make an order under this section if the Treasurer receives a compliance certificate about the council and the certificate specifies compliance issues.
- (4) A compliance certificate is sufficient evidence to justify making an order under this section in relation to any compliance issues specified in the certificate.
- (5) The Treasurer may, but is not required to, give the council prior notice of the Treasurer's intention to make an order under this section.
- (6) A council must not fail to comply with an order under this section.
- (7) The Treasurer is to advise the Minister administering the Local Government Act 1993 of any order that the Treasurer makes under this section.
 Note. The Minister administering the Local Government Act 1993 has power under Part 6 of Chapter 13 of that Act to take various actions to redress a failure by a council to meet its legislative responsibilities.

138 Withholding collection payments

- (1) The Treasurer may withhold any collection payment that would otherwise be paid or payable to a council if the Treasurer is satisfied that a breach of this Act has been or will be committed by the council.
- (2) Payments may be withheld until the breach is remedied or restrained to the satisfaction of the Treasurer.

| (3) | Without limiting subsection (1), the Treasurer may withhold payments under this |
|-----|--|
| | section if the Treasurer receives a compliance certificate about the council and the |
| | certificate specifies compliance issues. |

- (4) A compliance certificate that specifies compliance issues is sufficient evidence to justify withholding payments under this section.
- (5) The Treasurer may, but is not required to, give the council prior notice of the Treasurer's intention to withhold payments under this section.

| Part 10 | | Miscellaneous | | | |
|---------|-------|--|----------------|--|--|
| 139 | Notic | ce relating to fire and emergency services levy | | | |
| | (1) | The Treasurer may, by order published in the Gazette, require a council to give an approved notice to liable persons for land in the council's area. | 3 4 | | |
| | (2) | An <i>approved notice</i> is a notice in a form approved by the Treasurer. | 5 | | |
| | (3) | An approved notice may contain such information as the Treasurer considers appropriate to inform the public about the levy and the operation of this Act. | 6 7 | | |
| | (4) | The order may specify the way in which councils are to give the approved notice to liable persons. | | | |
| | (5) | Without limiting subsection (4), the order may specify that the approved notice is to be included in: | 10 11 | | |
| | | (a) a notice of classification or sub-classification of land given by the council, or | 12 | | |
| | | (b) a levy notice, or | 13 | | |
| | | (c) a notice that levies a rate or charge under Chapter 15 of the <i>Local Government Act 1993</i> . | 14 15 | | |
| | (6) | A council must not fail to comply with a requirement made under this section. | 16 | | |
| 140 | Shar | ing of information between councils and agencies | 17 | | |
| | (1) | A council may disclose to the Valuer-General or the Chief Commissioner information held by the council: | 18 19 | | |
| | | (a) in accordance with a requirement imposed on the council by or under this Act, or | 20 21 | | |
| | | (b) otherwise for the purposes of the administration or execution of this Act. | 22 | | |
| | (2) | The Valuer-General may disclose to the Chief Commissioner or a council information held by the Valuer-General for the purposes of the administration or execution of this Act. | | | |
| | (3) | The Chief Commissioner may disclose to the Valuer-General information held by the Chief Commissioner for the purposes of the administration or execution of this Act. | | | |
| | (4) | The Chief Commissioner may disclose to a council information obtained by the Chief Commissioner in the exercise of functions under this Act for the purposes of the administration or execution of this Act. | | | |
| | (5) |) The Valuer-General may collect information held by a council or the Chief Commissioner, and use that information, for the purposes of the administration or execution of this Act. | | | |
| | (6) | The Chief Commissioner may collect information held by a council or the Valuer-General, and use that information, for the purposes of the administration or execution of this Act. | 34 35 36 | | |
| | (7) | This section applies despite section 11 of the Valuation of Land Act 1916. | 37 | | |
| | (8) | This section does not limit section 82 of the Taxation Administration Act 1996. | 38 | | |
| | (9) | In this section: | 39 | | |
| | | Chief Commissioner includes a person authorised by the Chief Commissioner. | 40 | | |
| | | <i>council</i> includes a person authorised by a council. | 41 | | |
| | | <i>information</i> includes personal information within the meaning of the <i>Privacy and Personal Information Protection Act 1998</i> . | 42 43 | | |

| | | Valu | er-General includes a person authorised by the Valuer-General. | 1 | | | |
|-----|-------------------------|--|---|----------------|--|--|--|
| 141 | Disc | losure | of information to Treasurer | 2 | | | |
| | (1) | | Valuer-General may disclose to the Treasurer any information held by the er-General for the purposes of the administration or execution of this Act. | 3 4 | | | |
| | (2) | This | section applies despite section 11 of the Valuation of Land Act 1916. | 5 | | | |
| | (3) | | section does not authorise the disclosure of personal information (within the ing of the <i>Privacy and Personal Information Protection Act 1998</i>). | 6 7 | | | |
| | (4) | In thi | s section: | 8 | | | |
| | | | surer includes a person authorised by the Treasurer. er-General includes a person authorised by the Valuer-General. | 9 10 | | | |
| 142 | Noti | ces to | be in writing | 11 | | | |
| | | | tice under this Act must be by instrument in writing, except where this Act or egulations expressly authorises another means of giving the notice. | 12 13 | | | |
| 143 | Act I | oinds t | he Crown | 14 | | | |
| | | | Act binds the Crown in right of New South Wales and, in so far as the legislative or of the Legislature of New South Wales exists, the Crown in all its other cities. | 15 16 17 | | | |
| 144 | Delegation of functions | | | | | | |
| | (1) | The Treasurer may delegate to the Secretary of the Treasury or any person employed in the Public Service any function of the Treasurer under this Act, other than a function under Part 4 or this power of delegation. | | 19 20 21 | | | |
| | (2) | | legate may sub-delegate any function delegated by the Treasurer to any person oyed in the Public Service, if authorised in writing to do so by the Treasurer. | 22 23 | | | |
| | (3) | any f | Chief Commissioner may delegate to any person employed in the Public Service function of the Chief Commissioner under this Act, other than this power of pation. | 24 25 26 | | | |
| 145 | Serv | ice of | documents | 27 | | | |
| | (1) | | cument that is authorised or required by this Act or the regulations to be served by person may be served by any of the following methods: | 28 29 | | | |
| | | (a) | in the case of an individual—by personal delivery to the person, | 30 | | | |
| | | (b) | by post to the address specified by the person for the service of documents of that kind, | 31 32 | | | |
| | | (c) | in the case of an individual who has not specified such an address—by post to the residential or business address of the person last known to the person serving the document, | 33 34 35 | | | |
| | | (d) | in the case of a corporation—by post to the registered office or any other office of the corporation or by leaving it at any such office with a person apparently over the age of 16 years, | 36 37 38 | | | |
| | | (e) | by email to an email address specified by the person for the service of documents of that kind, | 39 40 | | | |
| | | (f) | by any other method authorised by the regulations for the service of documents of that kind. | 41 42 | | | |

| | (2) | Nothing in this section affects the operation of any provision of any other Act or law or of the rules of a court authorising a document to be served on a person by any other method. | 1 2 3 |
|-----|------|--|----------------------|
| | (3) | In this section, serve includes give or send. | 4 |
| 146 | App | lication of Recovery of Imposts Act 1963 | 5 |
| | (1) | The <i>Recovery of Imposts Act 1963</i> applies to the levy as if it were a tax payable to the Crown. | 6 7 |
| | (2) | Section 2 of that Act extends to proceedings for the recovery of the levy from a council. | 8 9 |
| | (3) | For the purposes of that Act, the validity or invalidity of taxation legislation includes the validity or invalidity of a decision of the Treasurer under Part 4. | 10 11 |
| 147 | Gen | eral tax exemption for council does not apply | 12 |
| | | This Act has effect despite section 741 of the <i>Local Government Act 1993</i> . Note. Land owned by a council is exempt from the levy only if it is government land. | 13 14 |
| 148 | Natu | re of proceedings for offences | 15 |
| | | Proceedings for an offence under this Act or the regulations may be dealt with summarily before the Local Court. | 16 17 |
| 149 | Regi | ulations | 18 |
| | (1) | The Governor may make regulations, not inconsistent with this Act, for or with respect to any matter that by this Act is required or permitted to be prescribed or that is necessary or convenient to be prescribed for carrying out or giving effect to this Act. | 19 20 21 22 |
| | (2) | A regulation may create an offence punishable by a penalty not exceeding 100 penalty units. | 23 24 |
| | (3) | Section 5 of the <i>Subordinate Legislation Act 1989</i> does not apply to the first principal statutory rule that is made under this Act. | 25 26 |
| 150 | Revi | ew of Act | 27 |
| | (1) | The Minister is to review this Act to determine whether the policy objectives of the Act remain valid and whether the terms of the Act remain appropriate for securing those objectives. | 28 29 30 |
| | (2) | The review is to be undertaken as soon as possible after the period of 5 years from the date of assent to this Act. | 31 32 |
| | (3) | A report on the outcome of the review is to be tabled in each House of Parliament within 12 months after the end of the period of 5 years. | 33 34 |
| | | | |

Schedule 1 Public benefit land 1 (Section 43) 2 1 Animal shelter, being premises used by a community group for the treatment or 3 adoption of animals. 4 2 Approved education and care service (within the meaning of the Children (Education 5 and Care Services) National Law (NSW)) or a State regulated education and care 6 service (within the meaning of the Children (Education and Care Services) 7 Supplementary Provisions Act 2011). 8 3 Boat shed, ramp or jetty. 9 4 Cemetery or crematorium, but only if the land is included on the Cemeteries and 10 Crematoria Register under the Cemeteries and Crematoria Act 2013. 11 5 Community centre, being a facility used for the physical, social, cultural or 12 intellectual development or welfare of the community. 13 Cultural site, being an aquarium, art gallery, botanic garden, concert hall, library or 6 14 archive, monument, memorial, museum, observatory, theatre or zoo. 15 7 Emergency services, being any of the following: 16 (a) a fire station, an ambulance station or a police station, 17 (b)land used by an emergency service for training purposes, 18 (c) land used by the State Emergency Service (within the meaning of the State 19 *Emergency Service Act 1989*). 20 8 Higher education provider (within the meaning of Division 16 of Part 2-1 of the 21 Higher Education Support Act 2003 of the Commonwealth). 22 9 Hospital (within the meaning of the *Public Health Act 2010*) or rehabilitation clinic. 23 10 Nature, being any of the following: 24 (a) a bicycle track, park, or walking trail, 25 a flora reserve or State forest (within the meaning of the *Forestry Act 2012*), (b)26 a marine park (within the meaning of the Marine Estate Management Act (c) 27 2014). 28 (d) a conservation area or state conservation area (within the meaning of the 29 National Parks and Wildlife Act 1974), 30 land that is the subject of a conservation agreement under the *Biodiversity* (e) 31 Conservation Act 2016, 32 (f) a wilderness area (within the meaning of the *Wilderness Act 1987*), 33 a World Heritage site. (g) 34 11 A non-government school (within the meaning of the Education Act 1990). 35 12 Place of worship. 36 13 Public toilet or amenities block. 37 14 Sporting complex, including sports club facilities. 38

15 TAFE establishment (within the meaning of the *Technical and Further Education Commission Act 1990*).

Schedule 2 Industrial land

| | (Section 46) | 2 | | | |
|----|---|----------------|--|--|--|
| 1 | Manufacturing, including: | 3 | | | |
| | (a) a general purpose factory, being a building or place used for manufacturing, assembly or repairs (for example, a factory, motor vehicle repair facility, or workshop), and | 4 5 6 | | | |
| | (b) a food processing factory, being a purpose-built food processing facility (for example, a cannery, milk production plant, dairy or processing plant), and | 7 8 | | | |
| | (c) a major industrial complex with special purpose improvements for large-scale industrial use (for example, a car plant or paper mill). | 9 10 | | | |
| 2 | Warehousing, distribution or storage, including: | 11 | | | |
| | (a) a general purpose warehouse used for the storage of goods, and | 12 | | | |
| | (b) open-area storage, with extensive hardstand area used for the storage of goods and equipment (for example, a wrecking yard, concrete batching yard, or container storage). | 13 14 15 | | | |
| 3 | Bulk grain storage, a silo or other structure used for the storage of grain, or bunkers used for the storage of grain. | 16 17 | | | |
| 4 | Bulk liquid storage fuel depot, being a building or place with tanks for the storage and distribution of bulk liquids (for example, a fuel depot or oil terminal) but excluding retail fuel outlets and service stations. | | | | |
| 5 | Coolstore or coldstore, being a purpose-built structure used for the cold storage of perishable products. | | | | |
| 6 | Works depot, being a building or place used as a works depot in conjunction with infrastructure maintenance (for example, a municipal depot). | | | | |
| 7 | Tannery or skins drying depot, being a building or place used for the tanning of skins and hides. | | | | |
| 8 | Abattoir, being a purpose-built structure used for the holding and slaughter of stock and the preparation of meat for the wholesale market. | | | | |
| 9 | Stock sales yard, being a purpose-built structure used for the yarding and selling of stock. | | | | |
| 10 | Rendering plant, being a purpose-built structure used for the extraction of lard, tallow and oil from animal parts. | | | | |
| 11 | Oil refinery, being a purpose-built structure used in the refinement and storage of petroleum products. | 33 34 | | | |
| 12 | Petro-chemical manufacturing, being a purpose-built structure used in the production of chemical-based products from petroleum. | 35 36 | | | |
| 13 | Sawmill, being a purpose-built structure used for the milling and curing of timber. | 37 | | | |
| 14 | Mining, being land that is classified as mining for the purposes of the Local Government Act 1993. | 38 39 | | | |
| 15 | Other extractive industries, including the extraction of sand, gravel, stone, clay, limestone, dolomite, gypsum, soil, precious metals, uranium, bauxite, gold, other | 40 41 | | | |
| | | | | | |

| | metals, precious stones, and other minerals and ore, and the extraction of non-metals by a licensed operator, from deep underground, by way of an inclined or vertical passageway or shaft equipped with lifting machinery. | 1 2 3 |
|----|---|-------------|
| 16 | Tailings dump, being the storage or treatment of minerals or non-minerals in tailing dumps or dams. | 4 5 |
| 17 | Oil well or bore. | 6 |
| 18 | Water well, being a narrow hole drilled or dug into the earth for the production of water, including mineral water, water for stock or domestic purposes, and water for use in irrigation. | 7 8 9 |
| 19 | Evaporative salt pan. | 10 |
| 20 | Salt extraction basin, being the extraction of salt from a lake or man-made evaporative basin. | 11 12 |
| 21 | Dredging operations, being activities involving dredging for the extraction, treatment and restoration of submerged materials. | 13 14 |
| 22 | Gas or fuel well, being specialist infrastructure used as a gas or fuel well. | 15 |
| 23 | Gas or fuel refinery. | 16 |
| 24 | Gas or fuel storage excluding retail fuel outlets and service stations. | 17 |
| 25 | Electricity power generators, including thermal power plants, hydroelectric generators, wind farms, and solar electric generation. | 18 19 |
| 26 | Refuse incinerator. | 20 |
| 27 | Refuse transfer station. | 21 |
| 28 | Sanitary landfill, being the disposal of household, commercial, industrial and public waste. | 22 23 |
| 29 | Refuse recycling. | 24 |
| 30 | Hazardous materials or toxic storage centre. | 25 |
| 31 | Toxic by-product storage or decontamination site for the storage of mining waste. | 26 |
| 32 | Sewerage or stormwater treatment plant. | 27 |
| 33 | Sewerage or stormwater pump station. | 28 |
| 34 | Water treatment plant, including a desalination plant. | 29 |
| 35 | Water storage tank, pressure control tower or pumping station. | 30 |
| 36 | Weighbridge. | 31 |
| 37 | Railway switching and marshalling yards. | 32 |
| 38 | Railway maintenance facility. | 33 |
| 39 | Railway passenger terminal, including a station. | 34 |
| 40 | Railway freight terminal facility. | 35 |
| | | |

| 41 | Tramway maintenance and terminal facility. | 1 |
|----|---|----------|
| 42 | Tram stop, including a shelter or a platform. | 2 |
| 43 | Airport hangar. | 3 |
| 44 | Heliport. | 4 |
| 45 | Port dock or berth, including the seabed adjoining a wharf with infrastructure used for the berthing of ships. | 5 6 |
| 46 | Cargo port wharf or pier and apron, being specialist infrastructure to facilitate the movement of containers and cargo to and from ships. | 7 8 |
| 47 | Wharf storage sheds, being enclosed storage facilities within a wharf. | 9 |
| 48 | Piers, storages and slipways used for maintenance and launching of boats. | 10 |
| 49 | Lighthouse and navigation aids, being specialist infrastructure used to assist in sea navigation. | 11 12 |
| 50 | Postal exchange and mail and package sorting centre. | 13 |
| 51 | Telecommunications buildings and maintenance depots. | 14 |
| 52 | Telecommunications towers and aerials. | 15 |
| 53 | Printing works and printing press, being specialist infrastructure used for printing works (for example, newspaper print and magazines). | 16 17 |
| 54 | Purpose-built telephone exchange. | 18 |
| | | |

Savings, transitional and other provisions Schedule 3

Part 1 General

| 1 | 1 Regulations | | | | | |
|------|---|--|----------------------------|--|--|--|
| | (1) | The regulations may contain provisions of a savings or transitional nature consequent on the enactment of this Act or any Act that amends this Act. | 4 5 | | | |
| | (2) Any such provision may, if the regulations so provide, take effect from the date of assent to the Act concerned or a later date. | | | | | |
| | (3) To the extent to which any such provision takes effect from a date that is earlier than the date of its publication on the NSW legislation website, the provision does not operate so as: | | | | | |
| | | (a) to affect, in a manner prejudicial to any person (other than the State or an authority of the State), the rights of that person existing before the date of its publication, or | 11 12 13 | | | |
| | | (b) to impose liabilities on any person (other than the State or an authority of the State) in respect of anything done or omitted to be done before the date of its publication. | 14 15 16 | | | |
| | (4) | A regulation made for the purposes of this clause may make separate savings and transitional provisions or amend this Schedule to consolidate savings and transitional provisions. | 17 18 19 | | | |
| Par | t 2 | Provisions consequent on enactment of this Act | 20 | | | |
| Divi | sion | 1 Preliminary | 21 | | | |
| 2 | Defir | nition | 22 | | | |
| | | In this Part: | 23 | | | |
| | | initial land classification means a first classification of land under this Act. | 24 | | | |
| Divi | sion | 2 Introduction of levy | 25 | | | |
| 3 | First | levy | 26 | | | |
| | | The first levy is payable for the financial year commencing on 1 July 2017. | 27 | | | |
| 4 | Valio | lation of things done in anticipation of commencement | 28 | | | |
| | | Anything done by a council, or by a person who has functions under this Act, in anticipation of the enactment and commencement of this Act, that would have been validly done under this Act if this Act had been in force at the time that it was done, is taken to have been validly done under this Act. | 29 30 31 32 | | | |
| 5 | Refe | rences to 30 April | 33 | | | |
| | | If this Act, or an amendment to another Act that is made by this Act, requires something to be done by 30 April and this Act commences on or after 23 April 2017, the thing is taken to be validly done by 30 April 2017 if it is done by the date that is 7 days after the commencement of this Act or by a later date prescribed by the regulations. | 34 35 36 37 38 | | | |

6 Matters that may be dealt with by regulation

(1) This clause has effect pending the making of regulations under this Act, and ceases to have effect when there is a regulation in force under this Act.

- (2) Land that is exempt from all rates under section 555 (1) (g) or (h) of the *Local Government Act 1993* is taken to be unvalued land under section 19.
- (3) The land referred to in subclause (2) is taken to have a land value of zero (as if a zero land value were specified by the regulations under this Act).
- (4) The land value of land on Lord Howe Island is to be calculated at a rate of \$70 per square metre or part of a square metre (as if that rate were specified by the regulations under this Act).
- (5) For the purposes of section 42 (1) (b) (i) of this Act, each of the following State owned corporations, and any subsidiary of those corporations, is taken to have been declared by the regulations to be a liable State owned corporation (accordingly, land owned by the State owned corporation or subsidiary is not government land):
 - (a) Essential Energy,
 - (b) Ausgrid,
 - (c) Endeavour Energy.
- (6) For the purposes of section 42 (1) (g) of this Act, the following land is taken to have been declared by the regulations to be government land:
 - (a) land owned by WCX M4 Pty Limited (ACN 602 963 806),
 - (b) land owned by the Crown or by a council that is the subject of a lease to the Crown (within the meaning of this Act) or a council, if the dominant use of the land is for the purposes of the lease,
 - (c) land that is the subject of a lease and is used for the purpose of a government school (within the meaning of the *Education Act 1990*), if that use is the dominant use of the land,
 - (d) Schedule 14 lands, or ALR Act lands, within the meaning of Part 4A of the *National Parks and Wildlife Act 1974*, that are leased to the Minister under that Part.

7 Classification deadline

- (1) For the purpose of enabling the levy to be charged on and from 1 July 2017, a council must classify all parcels of land within its area by the date that is 7 days after the commencement of this Act or by a later date prescribed by the regulations.
- (2) A failure by a council to classify any land in its area by that date does not affect the validity of any classification (whether done before or after that date).
- (3) A council may sub-classify land as vacant land as part of an initial land classification, but is not required to do so.

8 Notice of initial land classification

- (1) A council is not required to give notice to a liable person of an initial land classification within 30 days of the classification if the classification is done before 1 April 2017, or the date that is 7 days after the commencement of this Act, whichever is later.
- (2) However, a council must, by 30 April 2017, give notice to a liable person of an initial land classification and the effective date for the classification, in accordance with Part 5 of this Act.

| | (3) | If notice is given by post, a requirement to give notice by a particular date is satisfied if the notice is posted by that date. | 1 2 |
|----|-------|---|----------------------------|
| 9 | Exte | nded time for initial appeals and reviews | 3 |
| | (1) | Despite anything to the contrary in Part 5 of this Act, an application for a review of a classification of land under this Act may be made to a council on or before 31 August 2017, even if that date occurs later than 30 days after notice of the classification is given to the liable person. | 4 5 6 7 |
| | (2) | An appeal may be made on or before 31 August 2017 to the Land and Environment Court against the following, even if that date occurs later than the date by which, but for this clause, it is required to be made under Part 5 of this Act: | 8 9 10 |
| | | (a) an initial land classification, | 11 |
| | | (b) the effective date for an initial land classification, | 12 |
| | | (c) a refusal by a council to sub-classify land as vacant land, | 13 |
| | | (d) the effective date for a sub-classification of land as vacant land. | 14 |
| | (3) | This clause does not limit the period during which an application for review of appeal may be made under Part 5 of this Act. | 15 16 |
| 10 | Estir | nates under Part 4—adjustments for 2017 | 17 |
| | (1) | The first year in which the Valuer-General is required to provide estimates under sections 31–33 is 2017. | 18 19 |
| | (2) | In 2017, the estimates must be given to the Treasurer in accordance with sections 31–33 by a date approved by the Treasurer (the reference to 15 March is to be disregarded). | 20 21 22 |
| | (3) | The first year in which a council is required to provide the levy estimate information under section 34 is 2017. | 23 24 |
| | (4) | In 2017, the levy estimate information must be provided to the Valuer-General in accordance with section 34 by the date prescribed by the regulations or, if no date is prescribed, by the date that is 7 days after the commencement of this Act (the reference to 15 February is to be disregarded). | 25 26 27 28 |
| | (5) | For 2017 only, section 34 applies as if a reference to 31 December in the year before the return is required to be provided were a reference to a date prescribed by the regulations or, if no date is prescribed, the date the information is provided to the Valuer-General. | 29 30 31 32 |
| | (6) | The Valuer-General may treat any information provided by a council in anticipation of the enactment and commencement of this Act as levy estimate information and, accordingly, may use that information to make the estimates required to be made in 2017 under Part 4 of this Act (even if a return has not been provided by a council under section 34 at the time that the estimate is made). | 33 34 35 36 37 |
| 11 | Allow | wance for Monitor budget in first 2 years | 38 |
| | (1) | In the 2017/2018 financial year and the 2018/2019 financial year, the Monitor budget is to be added into the calculation of the FESL revenue target under Division 2 of Part 4 of this Act (so that the FESL revenue target includes the Monitor budget for the financial year). | 39 40 41 42 |
| | (2) | The Treasurer must, before the date of 30 April that occurs before the commencement of the 2017/2018 financial year and the 2018/2019 financial year, prepare and adopt an estimate of the total of the expenditure that will be incurred in | 43 44 45 |

the exercise of the Monitor's functions for that financial year. That estimate is the *Monitor budget* for the financial year for which it is made.

12 Allowance for start-up costs in initial period

- (1) The Treasurer may add into the calculation of the FESL revenue target under Division 2 of Part 4 of this Act for a financial year any amount that the Treasurer certifies is a start-up cost for the levy (so that the FESL revenue target includes the certified start-up cost or costs).
- (2) Each of the following is a *start-up cost* for the levy:
 - (a) any amount paid by the Treasurer to a council to reimburse the costs of the initial implementation of the levy,
 - (b) any cost incurred by or on behalf of the Crown in the initial implementation of the levy.
- (3) The Treasurer is not to certify an amount as a start-up cost for the levy if the amount has already been included in the calculation of the collection cost for the levy under Division 2 of Part 4 of this Act.
- (4) To avoid doubt, this clause extends to start-up costs incurred before the commencement of this Act.
- (5) The fact that the Treasurer does not include a start-up cost, or part of a start-up cost, in the calculation of the FESL revenue target for the financial year after which it is incurred does not prevent the Treasurer from including that start-up cost, or part, in the calculation of the FESL revenue target for a subsequent financial year. Note. This subclause permits initial start-up costs to be spread over the first 5 years of the scheme.
- (6) However, the Treasurer may add a start-up cost, or part of a start-up cost, to the FESL revenue target in a financial year only if the amount has not been included in the calculation of the FESL revenue target for a previous financial year.
- (7) This clause permits a start-up cost to be added to the FESL revenue target for a financial year only in the financial years from the 2017/2018 financial year to 2022/2023 financial year inclusive.

13 Returns under section 107

The first return that is required under section 107 is a return for the 2017/2018 financial year.

| Sch | edule 4 | A | mendment of other legislation | 1 | | |
|-----|-------------------------------|---------|---|----------------------|--|--|
| 4.1 | Emergen | cy Se | ervices Levy Insurance Monitor Act 2016 No 23 | 2 | | |
| [1] | Section 3 Definitions | | | | | |
| | Insert in alphabetical order: | | | | | |
| | _ | | <i>f Commissioner</i> means the Chief Commissioner of State Revenue under <i>Faxation Administration Act 1996</i> . | 5 6 | | |
| | | comp | <i>gency services contribution</i> means a contribution by an insurance pany under the emergency services funding scheme. | 7 8 | | |
| | | relati | rgency services levy winding up arrangements means the arrangements ing to emergency services levy reform, including the matters provided for t under: | 9 10 11 | | |
| | | (a) | the Fire and Emergency Services Levy Act 2017, or | 12 | | |
| | | (b) | Schedule 4 to the Fire Brigades Act 1989, or | 13 | | |
| | | (c) | Schedule 3 to the Rural Fires Act 1997, or | 14 | | |
| | | (d) | Schedule 1 to the State Emergency Service Act 1989. | 15 | | |
| [2] | Section 3, o | definit | tion of "emergency services funding scheme" | 16 | | |
| | | | e scheme for funding certain fire and emergency services from ired to be paid by insurance companies under". | 17 18 | | |
| | from contri | butior | scheme means the scheme for funding certain fire and emergency services as required to be paid by insurance companies under the following force before the enactment of the <i>Fire and Emergency Services Levy Act</i> | 19 20 21 22 | | |
| [3] | Section 3, o | definit | tion of "emergency services levy reform" | 23 | | |
| | Omit the de | finitio | n. Insert instead: | 24 | | |
| | | emer | gency services levy reform means: | 25 | | |
| | | (a) | the abolition, by the <i>Fire and Emergency Services Levy Act 2017</i> , of the emergency services funding scheme, and | 26 27 | | |
| | | (b) | the establishment of a fire and emergency services levy by that Act. | 28 | | |
| [4] | Section 3 (2 | 2) | | 29 | | |
| | Insert at the | end o | f the section: | 30 | | |
| | (2) | Note | s included in this Act do not form part of this Act. | 31 | | |
| [5] | Section 14 | Price | exploitation | 32 | | |
| | Omit section | n 14 (| 1) (b) (ii). Insert instead: | 33 | | |
| | | | (ii) the contributions required to be paid by the insurance company under the emergency services funding scheme, and | 34 35 | | |
| [6] | Section 30 | Inforr | nation relating to emergency services levy reform | 36 | | |
| | Omit section | n 30 (| 1). Insert instead: | 37 | | |
| | (1) | | Monitor may, by order published in the Gazette: | 38 | | |
| | | (a) | require any insurance company or class of insurance companies to provide information about emergency services levy reform or the functions of the Monitor under this Act (or both), including about the | 39 40 41 | | |

| | | | | impact of emergency services levy reform on the price payable for the issue of a regulated contract of insurance, and | 1 |
|-----|-------|-------------------|---------|---|----------------------|
| | | | (b) | specify the way in which the impact of emergency services levy reform is to be calculated, and | |
| | | | (c) | specify the way in which the information is to be provided (including by requiring it to be contained in an invoice or other statement as to the price payable for the issue of a regulated contract of insurance). | (|
| [7] | Sect | ion 30 | (2) | | 8 |
| | | t "that ection | | tement includes the information contained in the notice published under | 9 10 |
| | | | | hat any information required to be provided by an order under provided in accordance with the requirements of the order". | 1 [.] 1: |
| [8] | Sect | ion 30 | (3) | | 1: |
| | Omit | t "subs | ection | (2)". Insert instead "this section". | 14 |
| [9] | Part | 3A | | | 1 |
| | Inser | t after | Part 3: | | 16 |
| | Par | t 3A | Inv | estigation of overcharging under scheme | 1 |
| | 31A | Defir | nitions | \$ | 18 |
| | | | | is Part: | 19 |
| | | | | recovery order—see section 31J. | 20 |
| | | | | <i>2 years of the scheme</i> means the financial years commencing on 1 July and 1 July 2016. | 2 2 |
| | | | | -collection amount—see section 31C. | 23 |
| | | | | nd undertaking—see section 31H. | 24 |
| | | | relev | pant policy holder—see section 31B. | 2 |
| | 31B | Moni | tor to | investigate over-collection amounts | 26 |
| | | (1) | The | Monitor must: | 27 |
| | | | (a) | investigate and assess whether insurance companies are liable for over-collection amounts under the scheme, and | 28 29 |
| | | | (b) | endeavour to ensure that any insurance company that is liable for an over-collection amount: | 30 31 |
| | | | | (i) refunds the over-collection amount to relevant policy holders of the insurance company, if that is practicable, or | 3: 3: |
| | | | | (ii) if that is not practicable—pays the over-collection amount to the Chief Commissioner for payment into the Consolidated Fund. | 3. 3: |
| | | (2) | polic | blicy holder is a <i>relevant policy holder</i> of an insurance company if the by holder was insured under a regulated contract of insurance with the rance company in the final 2 years of the scheme. | 3) 3 3) |
| | 31C | Liabi | lity fo | r an over-collection amount | 39 |
| | | (1) | schei | nsurance company is liable for an over-collection amount under the me if the total amount collected by the insurance company exceeds the amount contributed by the insurance company. | 4) 4 4: |
| | | | | | |

| | (2) | The total amount collected by an insurance company is the total amount that, in the opinion of the Monitor, was collected by the insurance company from relevant policy holders for, or purportedly for, the payment of emergency service contributions in the final 2 years of the scheme. | 1 2 3 4 | | |
|-----|--|--|------------------|--|--|
| | (3) | If the Monitor has insufficient information to decide exactly the total amount collected by an insurance company, the Monitor may estimate that amount. | 5 6 | | |
| | (4) | The total amount contributed by an insurance company is the total of the following amounts: | 7 8 | | |
| | | (a) the final 2-year assessment of the insurance company given to the Monitor under clause 37 of Schedule 4 to the <i>Fire Brigades Act 1989</i> , | 9 10 | | |
| | | (b) the final 2-year assessment of the insurance company given to the Monitor under clause 35 of Schedule 3 to the <i>Rural Fires Act 1997</i> , | 11 12 | | |
| | | (c) the final 2-year assessment of the insurance company given to the Monitor under clause 19 of Schedule 1 to the <i>State Emergency Service Act 1989</i> . | 13 14 15 | | |
| | (5) | The <i>over-collection amount</i> is the amount by which the total amount collected by the insurance company exceeds the total amount contributed by the insurance company. | 16 17 18 | | |
| 31D | Monitor to assess over-collection amount | | | | |
| | (1) | If the Monitor is satisfied that an insurance company is liable for an over-collection amount, the Monitor must issue an assessment for that over-collection amount. | 20 21 22 | | |
| | (2) | The Monitor may, at any time, reassess the over-collection amount of an insurance company by issuing a further assessment. | 23 24 | | |
| | (3) | The Monitor is not required to issue an assessment for an over-collection amount if the Monitor: | 25 26 | | |
| | | (a) comes to an agreement with an insurance company about its liability for an over-collection amount, and | 27 28 | | |
| | | (b) accepts a refund undertaking from the insurance company for the over-collection amount. | 29 30 | | |
| 31E | Notio | ce of assessment to be given to insurance company | 31 | | |
| | (1) | If the Monitor issues an assessment for an over-collection amount, the Monitor | 32 | | |

- (1) If the Monitor issues an assessment for an over-collection amount, the Monitor must serve notice of the assessment on the insurance company that is liable for the over-collection amount.
- (2) The notice must specify the over-collection amount.
- (3) The notice must:
 - (a) advise the insurance company that, if the insurance company fails to give a refund undertaking to the Monitor in relation to the over-collection amount, in terms acceptable to the Monitor, the amount can be referred to the Chief Commissioner for debt recovery action, and

34

35

36

37

38

39

40

41

42

43

44

45

- (b) specify the deadline for giving the refund undertaking to the Monitor.
- (4) The notice must advise the insurance company about how to object to the assessment, unless the assessment is of a kind that is not open to objection by the insurance company.
- (5) The deadline for giving the refund undertaking must not be less than 30 days after the notice is given to the insurance company.

| | (6) | The Monitor may extend the deadline for giving the refund undertaking. | 1 | | |
|-----|--|--|----------------------------|--|--|
| 31F | Objections to assessments | | | | |
| | (1) | An insurance company may object to an assessment of an over-collection amount within 21 days after notice of the assessment is given to the insurance company or within such further period as the Monitor allows. | 3 4 5 | | |
| | (2) | An objection must be made in writing to the Monitor and must specify the grounds of the objection. | 6 7 | | |
| | (3) | Only one objection can be made by an insurance company to an assessment. | 8 | | |
| | (4) | The insurance company has the onus of establishing the grounds of the objection. | 9 10 | | |
| | (5) | An objection is not duly made unless it is made in accordance with this section. | 11 | | |
| 31G | Monitor to decide objection | | | | |
| | (1) | The Monitor must consider an objection to an assessment, if the objection is duly made by the insurance company, and either allow the objection in whole or in part or disallow the objection. | 13 14 15 | | |
| | (2) | If the Monitor delegates the functions conferred by this section, the delegate who considers the objection must be a different person from, and not subordinate to, the person who made the decision against which the objection is lodged. | 16 17 18 19 | | |
| | (3) | The Monitor must give notice in writing to the insurance company of its decision on the objection. | 20 21 | | |
| | (4) | The Monitor may issue a reassessment for the purpose of giving effect to a decision to allow an objection in whole or in part. | 22 23 | | |
| | (5) | A reassessment is not open to objection by an insurance company if it is made for the purpose of giving effect to a decision to allow an objection in whole or in part. | 24 25 26 | | |
| | (6) | If an objection to an assessment is duly made to the Monitor, the Monitor is not to refer the over-collection amount to the Chief Commissioner for debt recovery action: | 27 28 29 | | |
| | | (a) while the objection is pending before the Monitor, or | 30 | | |
| | | (b) before the period of 7 days has elapsed from the time that notice of the Monitor's decision on the objection is given to the insurance company. | 31 32 | | |
| 31H | Refund undertakings by insurance companies | | | | |
| | (1) | If an insurance company is liable for an over-collection amount, the Monitor may accept a refund undertaking from the insurance company in relation to the over-collection amount. | 34 35 36 | | |
| | (2) | A <i>refund undertaking</i> is an undertaking under Division 2 of Part 4 under which an insurance company that is liable for an over-collection amount agrees to refund the whole or part of the over-collection amount to relevant policy holders or to pay the over-collection amount or part of it to the Chief Commissioner. | 37 38 39 40 41 | | |
| | (3) | The Monitor is not to accept a refund undertaking that provides for the payment of an over-collection amount or part of an over-collection amount to the Chief Commissioner unless the Monitor is satisfied it is impracticable for | 42 43 44 | | |

the insurance company to refund the over-collection amount or part to relevant policy holders.

(4) The Monitor is to advise the Chief Commissioner of any refund undertaking that provides for the payment of an amount to the Chief Commissioner.
 Note. A refund undertaking is enforceable by proceedings in the Supreme Court.

311 Referral of amount to Chief Commissioner for debt recovery actions

- (1) The Monitor may refer an over-collection amount to the Chief Commissioner for debt recovery action if the insurance company liable for the over-collection amount fails to give the Monitor a refund undertaking in relation to the amount, in terms that are acceptable to the Monitor, by the deadline for giving the refund undertaking.
- (2) The referral is to be made by notice in writing to the Chief Commissioner in a form approved by the Chief Commissioner.
- (3) The notice must specify the over-collection amount.
- (4) The Monitor is to give the Chief Commissioner:
 - (a) a copy of the notice of assessment in relation to the insurance company, and
 - (b) any other information in relation to the referral that the Chief Commissioner reasonably requires to exercise his or her functions under this Part.

31J Issue of debt recovery order by Chief Commissioner

- (1) The Chief Commissioner may make an order (a *debt recovery order*) in relation to any over-collection amount referred to the Chief Commissioner by the Monitor under this Part.
- (2) A debt recovery order is an order for the recovery of the over-collection amount from the insurance company that is liable for the over-collection amount.
- (3) The debt recovery order must specify:
 - (a) the over-collection amount payable by the insurance company, and
 - (b) the date for payment.
- (4) A copy of the debt recovery order must be given to the insurance company.
- (5) The date for payment must be not less than 30 days after a copy of the order is given to the insurance company.

31K Insurance company must pay over-collection amount

(1) An insurance company that is required by a debt recovery order to pay an over-collection amount must pay that amount to the Chief Commissioner by the date for payment specified in the debt recovery order or within any further time allowed by the Chief Commissioner.

| | (2) | If an insurance company fails to pay the over-collection amount within 30 days of it being payable, the insurance company is guilty of an offence. Maximum penalty: 50 penalty units. | 1 2 3 |
|------|----------------------------|--|----------------------------|
| | (3) | The over-collection amount specified in the debt recovery order is recoverable by the Chief Commissioner from the insurance company as a debt in a court of competent jurisdiction and is payable into the Consolidated Fund. Note. Recovered over-collection amounts are deducted from the FESL revenue target under the <i>Fire and Emergency Services Levy Act 2017</i> and, accordingly, reduce the levy payable under that Act. | 4 5 6 7 8 9 |
| [10] | Section 53 | Purposes for which powers under Part may be exercised | 10 |
| | Omit "this A | Act or the regulations" from section 53 (a). | 11 |
| | Insert inste arrangemen | ad "this Act, the regulations or the emergency services levy winding up ts". | 12 13 |
| [11] | Section 53 | (b) | 14 |
| | Insert "or th | e emergency services levy winding up arrangements" after "this Act". | 15 |
| [12] | Section 53 | (d) | 16 |
| | | e emergency services levy winding up arrangements" after "this Act". | 17 |
| [13] | Section 75 | Exchange of information | 18 |
| | Omit parag | raph (b) of the definition of <i>relevant agency</i> in section 75 (6). Insert instead: | 19 |
| | | (b) the Chief Commissioner, or | 20 |
| | | (b1) the Commissioner of Fire and Rescue NSW, or | 21 |
| | | (b2) the Commissioner of the NSW Rural Fire Service, or | 22 |
| | | (b3) the Commissioner of the State Emergency Service, or | 23 |
| 4.2 | Fires Brig | gades Act 1989 No 192 | 24 |
| [1] | Section 44 | Definitions | 25 |
| | expenditure | definitions of assessment notice, capital account, contributors, estimated e, fire brigade contribution instalments, insurance company, recurrent e account and relevant insurance. | 26 27 28 |
| | Insert in alp | habetical order: | 29 |
| | | <i>assessment notice</i> means an assessment notice given to a local council under section 54. | 30 31 |
| | | <i>fire brigade contribution instalment</i> —see section 51A. | 32 |
| | | <i>fire brigades funding amount</i> —see section 49. | 33 |
| | | fire brigades funding target—see section 45. | 34 |
| | | <i>State Revenue Commissioner</i> —means the Chief Commissioner of State Revenue under the <i>Taxation Administration Act 1996</i> . | 35 36 |
| [2] | Section 44 | definition of "fire brigade contribution" | 37 |
| | Omit "conti | ributor". Insert instead "local council". | 38 |
| [3] | Section 44 | , definition of "fire brigades expenditure" | 39 |
| | Omit "recu | rent" wherever occurring. | 40 |

| [4] | Sect | ion 44, | , definition of "fire brigades expenditure" | 1 |
|-----|------|----------|---|----------------|
| | Omit | t paragi | raph (b). | 2 |
| [5] | Part | 5, Divi | isions 2, 3 and 4 | 3 |
| | Omit | the Di | ivisions. Insert instead: | 4 |
| | Divi | sion | 2 Fire brigades funding target | 5 |
| | 45 | Fire I | brigades funding target | 6 |
| | | (1) | Before 31 March in each year or a later date agreed to by the Treasurer, the Minister must: | 7 8 |
| | | | (a) prepare and, subject to the Treasurer's agreement, adopt a funding target for Fire and Rescue NSW for the next financial year (the <i>fire brigades funding target</i>), and | 9 10 11 |
| | | | (b) prepare and adopt an estimate of the amount of the fire brigades funding target for each fire district for the next financial year. | 12 13 |
| | | (2) | The Minister is to calculate the fire brigades funding target for the financial year using the following formula: | 14 15 |
| | | | $FT_{t} = EE_{t} + (AE_{t-2} - EE_{t-2})$ | |
| | | | where: | 16 |
| | | | FT_t is the fire brigades funding target for the financial year (represented by "t"). | 17 18 |
| | | | EE_t is the estimated fire brigades expenditure for the financial year. | 19 |
| | | | AE_{t-2} is the actual fire brigades expenditure for the financial year that commenced 2 years before the period "t" (represented as "t-2"). | 20 21 |
| | | | EE_{t-2} is the estimated fire brigades expenditure for the financial year that commenced 2 years before the period "t" (represented as "t-2"). | 22 23 |
| | | (3) | To avoid doubt, if $AE_{t-2} - EE_{t-2}$ is a negative amount, that amount is subtracted from EE_t . | 24 25 |
| | | (4) | The estimate of fire brigades expenditure for a financial year is to be reduced, as necessary, by the amount it is estimated will be paid to the Fund for the financial year (other than from contributions made by the Treasurer). | 26 27 28 |
| | | (5) | The actual fire brigades expenditure for a financial year is to be reduced, as necessary, by the amount paid to the Fund for the financial year (other than from contributions made by the Treasurer). | 29 30 31 |
| | | (6) | Without limiting subsections (4) and (5), the Minister may, with the Treasurer's agreement, decide that adjustments are not to be made to the fire brigades funding target in respect of: | 32 33 34 |
| | | | (a) certain money paid into the Fund, or | 35 |
| | | | (b) certain amounts it is estimated will be paid to Fire and Rescue NSW for the financial year. | 36 37 |
| | | (7) | In determining the amount of the fire brigades funding target for each fire district, the Minister may apportion the total estimated expenditure between fire districts in the way the Minister thinks fit. | 38 39 40 |

46 Commissioner to assist

| (1) | To assist the Minister in preparing and adopting the fire brigades funding |
|-----|---|
| | target for a financial year, the Commissioner must prepare and give to the |
| | Minister a written report and recommendations about fire brigades |
| | expenditure for the year and the estimated expenditure for each fire district and |
| | each local council. |

(2) The Minister must consider the Commissioner's report and recommendations in preparing the estimates.

Division 3 Treasurer to contribute to Fund

47 Treasurer to pay contributions

- (1) The Treasurer is to pay an annual contribution to the Fund for each financial year.
- (2) The annual contribution payable is the fire brigades funding amount.
- (3) The Treasurer may direct the payment of additional contributions to the Fund, subject to any terms and conditions approved by the Treasurer.
- (4) The annual contribution, and any additional contributions, are to be paid out of money provided by Parliament.

48 Contribution payable in instalments

The annual contribution is to be paid by the Treasurer in instalments on or before the following dates:

- (a) 1 August,
- (b) 1 October,
- (c) 1 January,
- (d) 1 April.

49 Fire brigades funding amount

- (1) As soon as practicable after the commencement of each financial year, the Minister must prepare and, subject to the Treasurer's agreement, adopt an update of fire brigades funding for the financial year (the *fire brigades funding amount*).
- (2) The fire brigades funding amount is the estimated fire brigades expenditure for the financial year (that is, EE_t in the fire brigades funding target for the financial year), adjusted and updated in accordance with this section.
- (3) Estimated fire brigades expenditure is to be adjusted for the purposes of the fire brigades funding amount so that the estimate:
 - (a) is reduced or increased, as appropriate, for any difference between the estimate of fire brigades expenditure for the financial year, as adopted in the fire brigades funding target, and an updated estimate of that amount, and
 - (b) is reduced or increased, as appropriate, for any difference between the fire brigades funding amount for the previous financial year and the actual amount of fire brigades expenditure for that financial year.
- (4) Without limiting subsection (3), the Minister may, with the Treasurer's agreement, decide that adjustments are not to be made to the fire brigades funding amount in respect of:

| | | (a) | any surplus in the Fund at the end of the previous financial year, or | 1 |
|------|------|-------------------------------|--|----------------------------|
| | | (b) | certain money paid into the Fund, or | 2 |
| | | (c) | certain amounts it is estimated will be paid to Fire and Rescue NSW for the financial year. | 3 4 |
| | (5) | this s | djustment to the estimated fire brigades expenditure that is made under section has no effect on, and is not to be used in the calculation of, the fire ades funding target for a financial year. | 5 6 7 |
| Divi | sion | 4 | Councils to pay fire brigade contributions | 8 |
| 50 | Coun | icil to | pay fire brigade contribution | 9 |
| | | | cal council is to pay to the State Revenue Commissioner a fire brigade ribution for each financial year. | 10 11 |
| 51 | Amo | unt of | contribution | 12 |
| | (1) | | amount of the fire brigade contribution is the amount determined by the ster for each local council. | 13 14 |
| | (2) | | Minister is to determine the contribution payable by a local council on the s of the fire brigades funding target for each fire district. | 15 16 |
| | (3) | | contribution payable by local councils for each fire district is 11.7% of the origades funding target applicable to the fire district. | 17 18 |
| | (4) | or co | contribution to be paid for a fire district is to be paid by the local council buncils of a local government area the whole or part of which is included e fire district. | 19 20 21 |
| | (5) | | ever, if a fire district comprises a reserved area (within the meaning of on 5) and no other area: | 22 23 |
| | | (a) | a local council of an area within the fire district is not required to pay a fire brigade contribution for the fire district, and | 24 25 |
| | | (b) | the fire brigade contribution for the fire district is to be paid from the National Parks and Wildlife Fund established under the <i>National Parks and Wildlife Act 1974</i> . | 26 27 28 |
| | (6) | the f gove curre | The district comprises more than one local government area, the amount of fire brigade contribution payable by the local council of each local ernment area in the fire district is to be apportioned according to the ent 5-year average of the total land values of all rateable land in the local ernment area or in the part of it in the fire district. | 29 30 31 32 33 |
| | (7) | give | cal council or an officer of a local council must, if asked by the Minister, the Minister any document or information required by the Minister to mine the council's fire brigade contribution. | 34 35 36 |
| | (8) | <i>curre</i> arith with | is section: <i>ent 5-year average</i> , in relation to the total land values of land, means the metic mean of the total land values of the land (estimated in accordance section 67 of the <i>Valuation of Land Act 1916</i>) as at 1 July in each of the ancial years before the date on which the average is being calculated. | 37 38 39 40 41 |
| 51A | Cont | ributi | on payable in instalments | 42 |
| | | A fir paya | The brigade contribution payable by a local council for a financial year is ble in 4 instalments (each of which is a <i>fire brigade contribution alment</i>). | 43 44 45 |

| | 51B | Minister to ad | vise State Revenue Commissioner | 1 |
|-----|------|--------------------------------------|--|----------------------|
| | | | nister is to advise the State Revenue Commissioner, by the date of that occurs immediately before the commencement of a financial | 2 3 4 |
| | | | e amount of the fire brigade contribution payable by each local puncil for the financial year, and | 5 6 |
| | | | e amount of the fire brigade contribution instalments payable by each cal council. | 7 8 |
| [6] | Part | 5, Division 5, h | eading | 9 |
| | Omi | the heading. Ins | ert instead: | 10 |
| | Divi | sion 5 P | ayment of contributions by local councils | 11 |
| [7] | Sect | on 52 | | 12 |
| | Omi | sections 52 and | 53. Insert instead: | 13 |
| | 52 | When instalm | ents are to be paid | 14 |
| | | council Commis | council must, in accordance with an instalment notice given to the by the State Revenue Commissioner, pay to the State Revenue sioner a fire brigade contribution instalment on or before each of the g days in a financial year: | 15 16 17 18 |
| | | |) September, | 19 |
| | | | December, | 20 |
| | | | l March,) June. | 21 22 |
| | • | | , suite. | |
| [8] | | ons 54–55A | 55 Incontingtond | 23 |
| | | | 55. Insert instead: | 24 |
| | 54 | Annual asses | | 25 |
| | | required notice th | te Revenue Commissioner must give to each local council that is to pay a fire brigade contribution for a financial year an assessment hat sets out the amount of the fire brigade contribution payable by the funcil for that year. | 26 27 28 29 |
| | | (2) The asse year bet payable. | essment notice must be given to the local council by 30 April in the fore the financial year for which the fire brigade contribution is | 30 31 32 |
| | 55 | Instalment no | tices | 33 |
| | | required | te Revenue Commissioner must give to each local council that is to pay a fire brigade contribution instalment a written notice (an <i>ent notice</i>) that specifies: | 34 35 36 |
| | | | e amount of the fire brigade contribution instalment payable under the stalment notice, and | 37 38 |
| | | (b) th | e date by which the fire brigade contribution instalment is payable. | 39 |
| | 55A | Fire brigade c | ontribution is debt payable to State Revenue Commissioner | 40 |
| | | | nole or part of a fire brigade contribution payable by a local council is to the State Revenue Commissioner as required, the unpaid amount is | 41 42 |

recoverable by the State Revenue Commissioner as a debt in a court of competent jurisdiction.

- (2) The applied provisions of the *Taxation Administration Act 1996* apply in relation to the fire brigade contribution payable by a local council under this Act as if Part 5 of this Act were a taxation law and the contribution were a tax payable by a council under a taxation law.
- (3) The *applied provisions* of the *Taxation Administration Act 1996* are the following provisions of that Act:
 - (a) Division 1 (Interest) of Part 5,
 - (b) sections 50 and 55 (which prohibit the giving of false and misleading information),
 - (c) Division 1 (Tax officers), Division 2 (Investigation) and Division 3 (Secrecy) of Part 9,
 - (d) sections 114, 115, 116, 118, 119 and 120,
 - (e) any provisions that are relevant to the interpretation of the above provisions.
- (4) The applied provisions apply to an assessment made by the State Revenue Commissioner under this Act in the same way as they apply to an assessment under that Act.
- (5) To avoid doubt, Part 10 of the *Taxation Administration Act 1996* does not apply to a fire brigade contribution or to the exercise of any of the State Revenue Commissioner's functions in respect of fire brigade contributions (including under the applied provisions of the *Taxation Administration Act 1996*).

[9] Section 56 How contribution is to be funded

Insert at the end of the section:

(2) Funds of a local council derived from donations and other voluntary contributions made for the purposes of this Act may not be used towards the payment of fire brigade contributions by the local council unless approved by the Minister.

[10] Part 5, Division 6 Contributions by insurance companies

Omit the Division.

[11] Part 5, Division 7, heading

Omit the heading. Insert instead:

Division 7 Miscellaneous

| [12] | Section 64 | A New South Wales Fire Brigades Fund | 36 |
|------|--------------|--|----|
| | Omit section | on 64A (2), (3) and (4). Insert instead: | 37 |
| | (2) | There is to be paid into the Fund: | 38 |
| | | (a) all contributions payable by the Treasurer under this Part and | 39 |

- (a) all contributions payable by the Treasurer under this Part, and
 (b) any other money appropriated by Parliament for payment into the Fund, and
- (c) the proceeds of investment of money in the Fund, and
- (d) any other money required by law to be paid into the Fund.

| [13] | Section 64 | • • | expenditure account". Insert instead "Fund". | 1 2 |
|-------|--------------|---------|---|----------------|
| F4 41 | | | Apenditure decount : Insert instead i fund : | |
| [14] | Section 64 | | ount". Insert instead "Fund". | 3 |
| | * | | ount : insert instead Tuna . | |
| [15] | Section 64 | | Insert instead: | 5 |
| | | | | 6 |
| | 64B Certi | | evidence | 7 |
| | | admi | rtificate signed by the Minister that states any of the following matters is issible in proceedings and, in the absence of evidence to the contrary, is f of the matters stated in the certificate: | 8 9 10 |
| | | (a) | the fire brigades funding target, as adopted by the Minister under this Part, | 11 12 |
| | | (b) | the fire brigades funding target applicable to a fire district, as adopted by the Minister under this Part, | 13 14 |
| | | (c) | the amount of the fire brigade contribution determined by the Minister for a specified local council for the financial year. | 15 16 |
| [16] | Part 5, Divi | ision 8 | 8 Miscellaneous | 17 |
| | Omit the D | ivisior | 1. | 18 |
| [17] | Section 80 | Inform | mation to policy holders | 19 |
| | Omit the se | ction. | | 20 |
| [18] | Schedule 1 | l Prop | portion of premiums subject to contribution | 21 |
| | Omit the So | chedul | e. | 22 |
| [19] | Schedule 4 | l Savi | ngs and transitional provisions | 23 |
| | Insert after | Part 9 | : | 24 |
| | Part 10 | | ovisions relating to the Fire and Emergency rvices Levy Act 2017 | 25 26 |
| | 31 Defir | nitions | 5 | 27 |
| | | In th | is Part: | 28 |
| | | ame | nding Act means the Fire and Emergency Services Levy Act 2017. | 29 |
| | | | contribution—see clause 34. | 30 |
| | | • | <i>return</i> —see clause 33. | 31 |
| | | - | <i>year of the scheme</i> means the financial year commencing on 1 July 2016. | 32 |
| | | unde | <i>rance company</i> means any body corporate, partnership, association, erwriter or person that or who: | 33 34 |
| | | (a) | issues or undertakes liability under policies of insurance against loss of or damage to any property situated in New South Wales, or | 35 36 |
| | | (b) | receives premiums in respect of such policies of insurance on behalf of or for transmission to any body corporate, partnership, association, underwriter or person outside New South Wales. | 37 38 39 |
| | | insu | rer loading—see clause 34. | 40 |

Monitor means the Emergency Services Levy Insurance Monitor appointed under the *Emergency Services Levy Insurance Monitor Act 2016*.

relevant insurance means insurance against loss of or damage to property in the State under the classes of policies specified in Schedule 1, as in force immediately before its repeal by the *Fire and Emergency Services Levy Act 2017*.

total amount, in relation to premiums, includes any brokerage or commission paid or due to be paid or allowed to be paid on:

- (a) the premiums, or
- (b) bonuses or return premiums allowed in respect of policies of insurance the subject of the premiums, or
- (c) such part of the premiums received by or due to the company as is paid or due to be paid by way of reinsurance by the company to another insurance company in the State,

but does not include duty payable under the *Duties Act 1997* in respect of policies of insurance the subject of the premiums.

32 Amendments do not affect pre-1 July 2017 liability

- (1) The amendments to Part 5 of this Act made by the amending Act apply in respect of the financial year commencing on 1 July 2017 and subsequent financial years.
- (2) Subject to this Part, the amendments do not affect any liability for a fire brigade contribution for a financial year commencing before 1 July 2017 and Part 5, section 80 and Schedule 1, as in force immediately before the amendments made to this Act by the amending Act, continue to apply in respect of any such financial year as if the amendments had not been made.

33 Final returns to be provided by insurance companies

- (1) An insurance company must, by 30 September 2017 or another date specified by the Commissioner by notice published in the Gazette, give to the Commissioner:
 - (a) a return in the form approved by the Commissioner showing the total amount of premiums received by or due to the company for the previous financial year for relevant insurance (a *final return*), and
 - (b) a certificate from an auditor that relates to the return and complies with subclause (2).
- (2) The certificate from the auditor must:
 - (a) be in the form approved by the Commissioner, and
 - (b) be from an auditor who is:
 - (i) a registered company auditor within the meaning of the *Corporations Act 2001* of the Commonwealth, or
 - (ii) a person not resident in the State who has qualifications that, in the Commissioner's opinion, are appropriate for the giving of the certificate.
- (3) An insurance company is guilty of an offence if it:
 - (a) fails to give the Commissioner a return and certificate as required by this clause, or

| | | (b) | gives partic | the Commissioner a return that is false or misleading in a material pular. | 1 2 |
|----|------|---------|------------------|--|----------------------|
| | | Maxi | mum p | penalty: 20 penalty units. | 3 |
| 34 | Calc | ulatior | n of fin | al contribution | 4 |
| | (1) | | | ssioner must, by the end of 30 November 2017, calculate a final for each liable insurance company. | 5 6 |
| | (2) | payal | ble by | <i>tribution</i> is the total fire brigade contribution that would have been the insurance company for the final year of the scheme if the had been calculated on the basis of: | 7 8 9 |
| | | (a) | the pr and | remium total provided by the insurance company in its final return, | 10 11 |
| | | (b) | the pr return | remium totals provided by all insurance companies in their final as. | 12 13 |
| | (3) | withi | n the ti | nce company fails to provide a final return to the Commissioner ime required under clause 33, or any further time (not exceeding owed by the Commissioner: | 14 15 16 |
| | | (a) | comp | ommissioner may estimate the premium total for that insurance any using any information available to the Commissioner iding information from previous returns, if any), and | 17 18 19 |
| | | (b) | | stimate is taken to be the premium total provided by the insurance any in its final return. | 20 21 |
| | (4) | unde | r subcla | nissioner estimates the premium total for an insurance company ause (3), the final contribution payable by that insurance company ate plus an insurer loading of 50% of that estimate. | 22 23 24 |
| | (5) | In thi | s claus | e: | 25 |
| | | | | <i>cance company</i> means an insurance company liable for a fire tribution in the final year of the scheme. | 26 27 |
| | | | | <i>tal</i> means the total amount of premiums of an insurance company lect to a fire brigade contribution. | 28 29 |
| 35 | Fina | asses | ssment | t | 30 |
| | (1) | requi | red to 1 | issioner must give to each insurance company that made or is make a fire brigade contribution for the final year of the scheme a nent notice for that year. | 31 32 33 |
| | (2) | The f | inal as | sessment notice must state the following: | 34 |
| | | (a) | | nal contribution payable by the insurance company for the final of the scheme, | 35 36 |
| | | (b) | insura instal | mount of any fire brigade contribution already paid by the ance company for that financial year including, if paid in ments, the total of all instalments already paid (the <i>total ibution already paid</i>), | 37 38 39 40 |
| | | (c) | | final contribution payable by the insurance company exceeds the contribution already paid: | 41 42 |
| | | | (i) | the difference between those amounts (a <i>contribution deficit</i>), and | 43 44 |
| | | | (ii) | the date by which the contribution deficit must be paid to the Commissioner (being a date that is no earlier than 30 days after the assessment notice is given to the insurance company), | 45 46 47 |

| | | (d) | if the total contribution already paid exceeds the final contribution payable by the insurance company—the difference between those amounts (a <i>contribution surplus</i>). | 1 2 3 |
|----|------|---------------------|--|----------------------|
| | (3) | | e assessment notice specifies a contribution deficit, the insurance bany must pay the contribution deficit to the Commissioner. | 4 5 |
| | (4) | paym | contribution deficit is payable to the Commissioner by the date for nent specified in the assessment notice or any later date permitted by the missioner. | 6 7 8 |
| | (5) | on w <i>Taxa</i> | insurance company fails to pay a contribution deficit in full by the date which it is payable to the Commissioner, the applied provisions of the <i>tion Administration Act 1996</i> apply to the unpaid amount as if the unpaid and were a tax and this Act were a taxation law. | 9 10 11 12 |
| | (6) | | applied provisions of the <i>Taxation Administration Act 1996</i> are the wing provisions of that Act: | 13 14 |
| | | (a) | Division 1 (Interest) of Part 5, | 15 |
| | | (b) | Division 2 (Penalty tax) of Part 5 (except sections 28-30), | 16 |
| | | (c) | any provisions that are relevant to the interpretation of the above provisions. | 17 18 |
| | (7) | For t 1996 | he purpose of the applied provisions of the <i>Taxation Administration Act</i> | 19 20 |
| | | (a) | a tax default is taken to occur if the contribution deficit is not paid in full by the date on which it is payable, and | 21 22 |
| | | (b) | a reference to the Chief Commissioner is to be read as a reference to the Commissioner under this Act. | 23 24 |
| | (8) | by ai payal | contribution deficit, or any part of a contribution deficit, is not paid in full n insurance company to the Commissioner by the date on which it is ble, the unpaid amount and any interest and penalty tax payable on the id amount: | 25 26 27 28 |
| | | (a) | constitutes a debt due and payable to the Commissioner, and | 29 |
| | | (b) | may be recovered in a court of competent jurisdiction by the Commissioner. | 30 31 |
| | (9) | A co | ntribution deficit is to be paid into the Fund. | 32 |
| | (10) | to ref | e assessment notice specifies a contribution surplus, the Commissioner is fund the amount of the contribution surplus to the insurance company no than 30 June 2018. | 33 34 35 |
| | (11) | A ref | fund is to be paid from the Fund. | 36 |
| 36 | Liab | ility of | owner where foreign insurer involved | 37 |
| | (1) | for a | Commissioner may notify a relevant owner that the owner is responsible n appropriate proportion of any contribution deficit that is required to be by a foreign insurance company that holds a risk in respect of the owner's erty. | 38 39 40 41 |
| | (2) | relev provi | appropriate proportion is the proportion that the premiums paid by the ant owner for the final year of the scheme bears to the premium total ided (or taken under this Part to have been provided) by the insurance bany in its final return. | 42 43 44 45 |

(3) Any insurer loading is not to be included in the calculation of the contribution deficit under this clause.

- (4) If the relevant owner fails to pay the appropriate proportion of the final contribution within 30 days after it is due, the owner is guilty of an offence. Maximum penalty: 10 penalty units.
- (5) An amount paid by a person under this clause may be deducted from any premium recoverable in the State by or on behalf of the foreign insurance company on the issue or renewal of an insurance policy on the property or may be recovered from the foreign insurance company as a debt by the person making the payment.
- (6) This clause applies whether the premium concerned was received in or outside the State.
- (7) In this clause:

relevant owner means a person who is the owner of property in respect of which a foreign insurance company has received a premium subject to a fire brigade contribution.

37 Commissioner to make final 2-year assessment

- (1) The Commissioner must make an assessment, in relation to each insurance company that was required to make fire brigade contributions in the final year of the scheme or the previous financial year, of the total amount of fire brigade contributions payable by the insurance company for the final 2 years of the scheme (a *final 2-year assessment*).
- (2) The final 2-year assessment is to consist of the total of the following amounts, as assessed by the Commissioner:
 - (a) the final contribution payable by the insurance company for the final year of the scheme (excluding any insurer loading payable by the insurance company),
 - (b) the total of fire brigade contributions payable by the insurance company for the financial year commencing on 1 July 2015.
- (3) The Commissioner is to give the final 2-year assessment to the Monitor.
- (4) The Commissioner must give to the Monitor a final 2-year assessment in relation to an insurance company within 30 days after making an assessment of the final contribution payable by the insurance company for the final year of the scheme.

38 Certificate evidence

A certificate signed by the Commissioner that states any of the following matters is admissible in proceedings and, in the absence of evidence to the contrary, is proof of the matters stated in the certificate:

- (a) the final contribution payable by an insurance company,
- (b) the final 2-year assessment for an insurance company,
- (c) the amount of any contribution deficit,
- (d) the amount of any contribution surplus.

39 Transitional provision for adjusted estimates by Minister

A reference in Part 5 of this Act, as amended by the amending Act, to contributions made to the New South Wales Fire Brigades Fund by the 45

| | Treasurer includes a reference to fire brigade contributions made for the financial year commencing on 1 July 2016 or a previous financial year. | 1 2 |
|-----|---|----------------------------|
| 4.3 | Land and Environment Court Act 1979 No 204 | 3 |
| [1] | Section 19 Class 3—land tenure, valuation, rating and compensation matters | 4 |
| | Insert after section 19 (d): | 5 |
| | (d1) appeals under section 64 of the <i>Fire and Emergency Services Levy Act</i> 2017, | 6 7 |
| [2] | Section 20 Class 4—environmental planning and protection, development contract and strata renewal plan civil enforcement | 8 9 |
| | Insert in alphabetical order in section 20 (3) (a): | 10 |
| | Fire and Emergency Services Levy Act 2017, | 11 |
| 4.4 | Local Government Act 1993 No 30 | 12 |
| [1] | Section 377 General power of the council to delegate | 13 |
| | Insert "under this or any other Act" after "functions of the council" in section 377 (1). | 14 |
| [2] | Section 409 The consolidated fund | 15 |
| | Insert after section 409 (8): | 16 |
| | (9) This section does not affect the requirements of the <i>Fire and Emergency</i> <i>Services Levy Act 2017</i> with respect to the payment of collection instalments to the Treasurer. | 17 18 19 |
| [3] | Section 511A Catching up of income lost due to reductions in valuation | 20 |
| | Omit "on objection under Part 3, or appeal under Part 4, of the <i>Valuation of Land Act 1916</i> " from section 511A (1). | 21 22 |
| | Insert instead "(whether because of an objection under the <i>Valuation of Land Act 1916</i> , an appeal under that Act or for any other reason)". | 23 24 |
| [4] | Section 687 Appearance in Local Court | 25 |
| | Insert "or any other" after "under this" in section 687 (b). | 26 |
| 4.5 | Local Government (General) Regulation 2005 | 27 |
| | Clause 127 Rates and charges notices | 28 |
| | Insert at the end of the clause: | 29 |
| | (2) If the notice includes particulars of a fire and emergency services levy that is payable, the notice may show the total amount payable for instalments for rates or charges and the fire and emergency services levy (instead of showing separately the instalments payable for rates or charges and the instalments payable for the levy). | 30 31 32 33 34 |
| | (3) In this clause: | 35 |
| | <i>fire and emergency services levy</i> means the levy under the <i>Fire and Emergency Services Levy Act 2017.</i> | 36 37 |

| 4.6 | Rur | al Fir | es Ao | ct 1997 No 65 | 1 |
|-----|-------|-------------------|----------|---|----------------|
| [1] | Part | 5, not | е | | 2 |
| | Omit | the no | ote. | | 3 |
| [2] | Sect | ion 10 | 1 Defi | nitions | 4 |
| | | | | ons of assessment notice, contributors, estimated expenditure, insurance at insurance and rural fire brigade contribution instalments. | 5 6 |
| | Inser | t in alp | | ical order: | 7 |
| | | | | ssment notice means an assessment notice given to a relevant council ar section 112. | 8 9 |
| | | | | <i>l fire brigade contribution instalment</i> —see section 110A. | 10 |
| | | | | <i>l fire brigade funding amount</i> —see section 108. | 11 |
| | | | | <i>l fire brigade funding target</i> —see section 103. | 12 |
| | | | | <i>Revenue Commissioner</i> —means the Chief Commissioner of State enue under the <i>Taxation Administration Act 1996</i> . | 13 14 |
| [3] | Sect | ion 10 | 1, def | inition of "rural fire brigade contribution" | 15 |
| | Omit | t "cont | ributoi | r". Insert instead "relevant council". | 16 |
| [4] | Sect | ion 10 | 1, defi | inition of "rural fire brigade expenditure" | 17 |
| | Omit | t "recu | rrent" | wherever occurring. | 18 |
| [5] | Sect | ion 10 | 1, defi | inition of "rural fire brigade expenditure" | 19 |
| | Omit | parag | raph (| 2). | 20 |
| [6] | Sect | ion 10 | 2 New | v South Wales Rural Fire Fighting Fund | 21 |
| | | t "into sectio | | are to be paid all contributions and other money received under this Part" (1). | 22 23 |
| [7] | Sect | ion 10 | 2 (1A) | | 24 |
| | Inser | t after | section | n 102 (1): | 25 |
| | | (1A) | Ther | e is to be paid into the Fund: | 26 |
| | | | (a) | all contributions payable by the Treasurer to the Fund under this Part, and | 27 28 |
| | | | (b) | any other money appropriated by Parliament for payment into the Fund, and | 29 30 |
| | | | (c) | the proceeds of investment of money in the Fund, and | 31 |
| | | | (d) | any other money required by law to be paid into the Fund. | 32 |
| [8] | | ion 10 | | | 33 |
| | Omit | t the se | ction. | Insert instead: | 34 |
| | 103 | Rura | l fire l | brigade funding target | 35 |
| | | (1) | | re 31 March in each year or a later date agreed to by the Treasurer, the ster must: | 36 37 |
| | | | (a) | prepare and, subject to the Treasurer's agreement, adopt a funding target for the NSW Rural Fire Service for the next financial year (the <i>rural fire brigade funding target</i>), and | 38 39 40 |

(b) prepare and adopt an estimate of the amount of the rural fire brigade funding target for each rural fire district for the next financial year. The Minister is to calculate the rural fire brigade funding target for the (2)financial year using the following formula: $FT_{t} = EE_{t} + (AE_{t-2} - EE_{t-2})$ where: FT_t is the rural fire brigade funding target for the financial year (represented by "t"). EE_t is the estimated rural fire brigade expenditure for the financial year. AE_{t-2} is the actual rural fire brigade expenditure for the financial year that commenced 2 years before the period "t" (represented as "t-2"). 10 EE_{t-2} is the estimated rural fire brigade expenditure for the financial year that 11 commenced 2 years before the period "t" (represented as "t-2"). 12 (3) To avoid doubt, if $AE_{t-2} - EE_{t-2}$ is a negative amount, that amount is subtracted 13 from EE_t. 14 (4)The estimate of rural fire brigade expenditure for a financial year is to be 15 reduced, as necessary, by the amount it is estimated will be paid to the Fund 16 for the financial year (other than from contributions made by the Treasurer). 17 The actual rural fire brigade expenditure for a financial year is to be reduced, (5)18 as necessary, by the amount paid to the Fund for the financial year (other than 19 from contributions made by the Treasurer). 20 Without limiting subsections (4) and (5), the Minister may, with the (6)21 Treasurer's agreement, decide that adjustments are not to be made to the rural 22 fire brigade funding target in respect of: 23 certain money paid into the Fund, or (a) 24 certain amounts it is estimated will be paid to the NSW Rural Fire (b) 25 Service for the financial year. 26 (7)In determining the amount of the rural fire brigade funding target for each rural 27 fire district, the Minister may apportion the total estimated expenditure 28 between rural fire districts in the way the Minister thinks fit. 29 Section 104 Relevant councils to give information to Minister 30 Omit "estimated expenditure". Insert instead "rural fire brigade funding target". 31 Section 105 32 Omit the section. Insert instead: 33 105 Commissioner to assist 34 To assist the Minister in preparing and adopting the rural fire brigade funding (1)35 target for a financial year, the Commissioner must prepare and give to the 36 Minister a written report and recommendations about rural fire brigade 37 expenditure for the year and the estimated expenditure for each rural fire 38 district and each relevant council. 39 (2)The Minister must consider the Commissioner's report and recommendations 40 in preparing the estimates. 41

1

2

3

4

5

6

7

8

9

[9]

[10]

| | | sions 4 and 5 | |
|-----|------------|--|--|
| Om | nit the Di | visions. Insert instead: | |
| Div | vision 4 | 4 Treasurer to contribute to Fund | |
| 106 | Treas | surer to pay contributions | |
| | (1) | The Treasurer is to pay an annual contribution to the Fund for each financial year. | |
| | (2) | The annual contribution payable is the rural fire brigade funding amount. | |
| | (3) | The Treasurer may direct the payment of additional contributions to the Fund, subject to any terms and conditions approved by the Treasurer. | |
| | (4) | The annual contribution, and any additional contributions, are to be paid out of money provided by Parliament. | |
| 107 | Cont | ribution payable in instalments | |
| | | The annual contribution is to be paid by the Treasurer in instalments on or before the following dates: | |
| | | (a) 1 August, | |
| | | (b) 1 October, | |
| | | (c) 1 January, (d) 1 A $anil$ | |
| | | (d) 1 April. | |
| 108 | Rura | I fire brigade funding amount | |
| | (1) | As soon as practicable after the commencement of each financial year, the Minister must prepare and, subject to the Treasurer's agreement, adopt an update of rural fire brigade funding for the financial year (the <i>rural fire brigade funding amount</i>). | |
| | (2) | The rural fire brigade funding amount is the estimated rural fire brigade expenditure for the financial year (that is, EE_t in the rural fire brigade funding target for the financial year), adjusted and updated in accordance with this section. | |
| | (3) | Estimated rural fire brigade expenditure is to be adjusted for the purposes of the rural fire brigade funding amount so that the estimate: | |
| | | (a) is reduced or increased, as appropriate, for any difference between the estimate of rural fire brigade expenditure for the financial year, as adopted in the rural fire brigade funding target, and an updated estimate of that amount, and | |
| | | (b) is reduced or increased, as appropriate, for any difference between the rural fire brigade funding amount for the previous financial year and the actual amount of rural fire brigade expenditure for that financial year. | |
| | (4) | Without limiting subsection (3), the Minister may, with the Treasurer's agreement, decide that adjustments are not to be made to the rural fire brigade funding amount in respect of: | |
| | | (a) any surplus in the Fund at the end of the previous financial year, or | |
| | | (b) certain money paid into the Fund, or | |
| | | (c) certain amounts it is estimated will be paid to the NSW Rural Fire Service for the financial year. | |

| | (5) | An adjustment to the estimated rural fire brigade expenditure that is made under this section has no effect on, and is not to be used in the calculation of, the rural fire brigade funding target for a financial year. | 1 2 3 | | | | |
|------|-------------------------------------|--|----------------------|--|--|--|--|
| Divi | ision | 5 Councils to pay rural fire brigade contribution | 4 | | | | |
| 109 | Cou | ncil to pay rural fire brigade contribution | 5 | | | | |
| | | A relevant council is to pay to the State Revenue Commissioner a rural fire brigade contribution for each financial year. | 6 7 | | | | |
| 110 | Amo | ount of contribution | 8 | | | | |
| | (1) | The amount of the rural fire brigade contribution is the amount determined by the Minister for each relevant council. | 9 10 | | | | |
| | (2) | The Minister is to determine the contribution payable by a relevant council on the basis of the rural fire brigade funding target for each rural fire district. | 11 12 | | | | |
| | (3) | The contribution payable by relevant councils for each rural fire district is 11.7% of the rural fire brigade funding target applicable to the rural fire district. | 13 14 15 | | | | |
| | (4) | The contribution to be paid for a rural fire district is to be paid by the relevant council or councils of an area the whole or part of which is included in the rural fire district. | 16 17 18 | | | | |
| | (5) | In determining the contribution payable by a relevant council, the Minister may apportion the rural fire brigade funding target for rural fire districts between councils of an area, the whole or part of which are included in that district, in the way the Minister thinks fit. | 19 20 21 22 | | | | |
| | (6) | A relevant council or an officer of a relevant council must, if asked by the Minister, give the Minister any document or information required by the Minister to determine the council's rural fire brigade contribution. | 23 24 25 | | | | |
| 110A | Contribution payable in instalments | | | | | | |
| | | A rural fire brigade contribution payable by a relevant council for a financial year is payable in 4 instalments (each of which is a <i>rural fire brigade contribution instalment</i>). | 27 28 29 | | | | |
| 110B | Mini | ster to advise State Revenue Commissioner | 30 | | | | |
| | | The Minister is to advise the State Revenue Commissioner, by the date of 15 April that occurs immediately before the commencement of a financial year, of: | 31 32 33 | | | | |
| | | (a) the amount of the rural fire brigade contribution payable by each relevant council for the financial year, and | 34 35 | | | | |
| | | (b) the amount of the rural fire brigade contribution instalments payable by each relevant council. | 36 37 | | | | |
| Part | 5, Div | ision 6, heading | 38 | | | | |
| Omi | t the he | eading. Insert instead: | 39 | | | | |
| Divi | ision | 6 Payment of contributions by relevant councils | 40 | | | | |

[12]

Section 111 [13] 1 Omit the section. Insert instead: 2 111 When instalments are to be paid 3 A relevant council must, in accordance with an instalment notice given to the 4 council by the State Revenue Commissioner, pay to the State Revenue 5 Commissioner a rural fire brigade contribution instalment on or before each of 6 the following days in a financial year: 7 (a) 30 September, 8 (b)31 December. 9 31 March, (c) 10 (d)30 June. 11 Sections 112–113A [14] 12 Omit sections 112 and 113. Insert instead: 13 Annual assessment notice 112 14 (1)The State Revenue Commissioner must give to each relevant council that is 15 required to pay a rural fire brigade contribution for a financial year an 16 assessment notice that sets out the amount of the rural fire brigade contribution 17 payable by the relevant council for that year. 18 (2)The assessment notice must be given to the relevant council by 30 April in the 19 year before the financial year for which the rural fire brigade contribution is 20 payable. 21 113 Instalment notices 22 The State Revenue Commissioner must give to each relevant council that is 23 required to pay a rural fire brigade contribution instalment a written notice (an 24 *instalment notice*) that specifies: 25 the amount of the rural fire brigade contribution instalment payable 26 (a)under the instalment notice, and 27 (b) the date by which the rural fire brigade contribution instalment is 28 payable. 29 113A Rural fire brigade contribution is debt payable to State Revenue Commissioner 30 If the whole or part of a rural fire brigade contribution payable by a relevant (1)31 council is not paid to the State Revenue Commissioner as required, the unpaid 32 amount is recoverable by the State Revenue Commissioner as a debt in a court 33 of competent jurisdiction. 34 (2)The applied provisions of the Taxation Administration Act 1996 apply in 35 relation to the contribution payable by a relevant council under this Act as if 36 Part 5 of this Act were a taxation law and the contribution were a tax payable 37 by a council under a taxation law. 38 (3)The applied provisions of the Taxation Administration Act 1996 are the 39 following provisions of that Act: 40 Division 1 (Interest) of Part 5, (a) 41 (b) sections 50 and 55 (which prohibit the giving of false and misleading 42 information), 43

(c) Division 1 (Tax officers), Division 2 (Investigation) and Division 3 (Secrecy) of Part 9,

1

2

3

4

5

6

7

8

q

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

29

30

31

32

33

34

35

36

37

38

39

40

41

42

- (d) sections 114, 115, 116, 118, 119 and 120,
- (e) any provisions that are relevant to the interpretation of the above provisions.
- (4) The applied provisions apply to an assessment made by the State Revenue Commissioner under this Act in the same way as they apply to an assessment under that Act.
- (5) To avoid doubt, Part 10 of the *Taxation Administration Act 1996* does not apply to a rural fire brigade contribution or to the exercise of any of the State Revenue Commissioner's functions in respect of rural fire brigade contributions (including under the applied provisions of the *Taxation Administration Act 1996*).

[15] Section 114 How contribution is to be funded

Insert at the end of the section:

(2) Funds of a relevant council derived from donations and other voluntary contributions made for the purposes of this Act may not be used towards the payment of rural fire brigade contributions by the relevant council unless approved by the Minister.

[16] Sections 114A and 114B

Insert after section 114:

114A Exemptions

- (1) The Governor may, on the recommendation of the Minister, exempt a relevant council from making a rural fire brigade contribution if the Governor is satisfied the danger from bush fires in the council's area is negligible.
- (2) If a relevant council has been exempted under this section from making rural fire brigade contributions, no payment is to be made in or towards the purpose specified in section 118 for the benefit of the council's area.

114B Certificate evidence

A certificate signed by the Minister that states any of the following matters is admissible in proceedings and, in the absence of evidence to the contrary, is proof of the matters stated in the certificate:

- (a) the rural fire brigade funding target, as adopted by the Minister under this Part,
- (b) the rural fire brigade funding target applicable to a rural fire district, as adopted by the Minister under this Part,
- (c) the amount of the rural fire brigade contribution determined by the Minister for a specified council for the financial year.

[17] Part 5, Division 6A Contributions by insurance companies Omit the Division.

[18] Section 121 Amendment of Schedule 2 Omit the section

| [19] | Schedul | e 2 Contributions of insurance companies | 1 | | | | | |
|------|--|---|----------------------------|--|--|--|--|--|
| | Omit the Schedule. | | | | | | | |
| [20] | Schedule 3, Savings, transitional and other provisions | | | | | | | |
| | Insert aft | er Part 7: | 4 | | | | | |
| | Part 8 | Provisions consequent on enactment of Fire and Emergency Services Levy Act 2017 | 5 6 | | | | | |
| | 29 De | efinitions | 7 | | | | | |
| | | In this Part: | 8 | | | | | |
| | | amending Act means the Fire and Emergency Services Levy Act 2017. | 9 | | | | | |
| | | final contribution—see clause 32. | 10 | | | | | |
| | | <i>final return</i> —see clause 31. | 11 | | | | | |
| | | <i>final year of the scheme</i> means the financial year commencing on 1 July 2016. | 12 | | | | | |
| | | <i>insurance company</i> means any body corporate, partnership, association, underwriter or person that or who: | 13 14 | | | | | |
| | | (a) issues or undertakes liability under policies of insurance against loss of or damage to any property situated in New South Wales, or | 15 16 | | | | | |
| | | (b) receives premiums in respect of such policies of insurance on behalf of | 17 | | | | | |
| | | or for transmission to any body corporate, partnership, association, underwriter or person outside New South Wales. | 18 19 | | | | | |
| | | <i>insurer loading</i> —see clause 32. | 20 | | | | | |
| | | <i>Monitor</i> means the Emergency Services Levy Insurance Monitor appointed under the <i>Emergency Services Levy Insurance Monitor Act 2016</i> . | 21 22 | | | | | |
| | | <i>relevant insurance</i> means insurance against loss of or damage to property in the State under the classes of policies specified in Schedule 2, as in force immediately before its repeal by the <i>Fire and Emergency Services Levy Act</i> 2017. | 23 24 25 26 | | | | | |
| | | <i>total amount</i> , in relation to premiums, includes any brokerage or commission paid or due to be paid or allowed to be paid on: | 27 28 | | | | | |
| | | (a) the premiums, or | 29 | | | | | |
| | | (b) bonuses or return premiums allowed in respect of policies of insurance the subject of the premiums, or | 30 31 | | | | | |
| | | (c) such part of the premiums received by or due to the company as is paid or due to be paid by way of reinsurance by the company to another insurance company in the State, | 32 33 34 | | | | | |
| | | but does not include duty payable under the <i>Duties Act 1997</i> in respect of policies of insurance the subject of the premiums. | 35 36 | | | | | |
| | 30 Ar | nendments have effect from 1 July 2017 | 37 | | | | | |
| | (1 |) The amendments to Part 5 of this Act made by the amending Act apply in respect of the financial year commencing on 1 July 2017 and subsequent financial years. | 38 39 40 | | | | | |
| | (2 |) Subject to this Part, the amendments do not affect any liability for a rural fire brigade contribution for a financial year commencing before 1 July 2017 and Part 5, and Schedule 2, as in force immediately before the amendments made to this Act by the amending Act, continue to apply in respect of any such financial year as if the amendments had not been made. | 41 42 43 44 45 | | | | | |

31 Final returns to be provided by insurance companies

| (1) | An insurance company must, by 30 September 2017 or another date specified by the Commissioner by notice published in the Gazette, give to the Commissioner: | |
|-----|---|--|
| | (a) a return in the form approved by the Commissioner showing the total | |

(a) a return in the form approved by the Commissioner showing the total amount of premiums received by or due to the company for the previous financial year for relevant insurance (a *final return*), and

- (b) a certificate from an auditor that relates to the return and complies with subclause (2).
- (2) The certificate from the auditor must:

| (a) | be in the form | approved by the | Commissioner, and |
|-----|---|-----------------|-------------------|
| () | • | | |

- (b) be from an auditor who is:
 - (i) a registered company auditor within the meaning of the *Corporations Act 2001* of the Commonwealth, or
 - (ii) a person not resident in the State who has qualifications that, in the Commissioner's opinion, are appropriate for the giving of the certificate.
- (3) An insurance company is guilty of an offence if it:
 - (a) fails to give the Commissioner a return and certificate as required by this clause, or
 - (b) gives the Commissioner a return that is false or misleading in a material particular.

Maximum penalty: 20 penalty units.

32 Calculation of final contribution

- (1) The Commissioner must, by the end of 30 November 2017, calculate a final contribution for each liable insurance company.
- (2) A *final contribution* is the total rural fire brigade contribution that would have been payable by the insurance company for the final year of the scheme if the contribution had been calculated on the basis of:
 - (a) the premium total provided by the insurance company in its final return, and
 - (b) the premium totals provided by all insurance companies in their final returns.
- (3) If an insurance company fails to provide a final return to the Commissioner within the time required under clause 31, or any further time (not exceeding 30 days) allowed by the Commissioner:
 - (a) the Commissioner may estimate the premium total for that insurance company using any information available to the Commissioner (including information from previous returns, if any), and
 - (b) that estimate is taken to be the premium total provided by the insurance company in its final return.
- (4) If the Commissioner estimates the premium total for an insurance company under subclause (3), the final contribution payable by that insurance company is that estimate plus an insurer loading of 50% of that estimate.

| | (5) | In th | is clau | se: | 1 | | | | |
|----|-------|---|--|---|----------------------|--|--|--|--|
| | | | | <i>rance company</i> means an insurance company liable for a rural fire atribution in the final year of the scheme. | 2 3 | | | | |
| | | | | <i>otal</i> means the total amount of premiums of an insurance company ject to a rural fire brigade contribution. | 4 5 | | | | |
| 33 | Final | asse | ssmen | ıt | 6 | | | | |
| | (1) | | | | | | | | |
| | (2) | The | final as | ssessment notice must state the following: | 10 | | | | |
| | | (a) | | inal contribution payable by the insurance company for the final of the scheme, | 11 12 | | | | |
| | | (b) | insur insta | mount of any rural fire brigade contribution already paid by the rance company for that financial year including, if paid in lments, the total of all instalments already paid (the <i>total ribution already paid</i>), | 13 14 15 16 | | | | |
| | | (c) | | e final contribution payable by the insurance company exceeds the contribution already paid: | 17 18 | | | | |
| | | | (i) | the difference between those amounts (a <i>contribution deficit</i>), and | 19 20 | | | | |
| | | | (ii) | the date by which the contribution deficit must be paid to the Commissioner (being a date that is no earlier than 30 days after the assessment notice is given to the insurance company), | 21 22 23 | | | | |
| | | (d) | paya | e total contribution already paid exceeds the final contribution ble by the insurance company—the difference between those unts (a <i>contribution surplus</i>). | 24 25 26 | | | | |
| | (3) | If the assessment notice specifies a contribution deficit, the insurance27company must pay the contribution deficit to the Commissioner.28 | | | | | | | |
| | (4) | The contribution deficit is payable to the Commissioner by the date for payment specified in the assessment notice or any later date permitted by the Commissioner. | | | | | | | |
| | (5) | on w <i>Taxa</i> | which in the second sec | the company fails to pay a contribution deficit in full by the date t is payable to the Commissioner, the applied provisions of the <i>dministration Act 1996</i> apply to the unpaid amount as if the unpaid re a tax and this Act were a taxation law. | 32 33 34 35 | | | | |
| | (6) | | | <i>d provisions</i> of the <i>Taxation Administration Act 1996</i> are the provisions of that Act: | 36 37 | | | | |
| | | (a) | | sion 1 (Interest) of Part 5, | 38 | | | | |
| | | (b) | | sion 2 (Penalty tax) of Part 5 (except sections 28–30), | 39 | | | | |
| | | (c) | | provisions that are relevant to the interpretation of the above isions. | 40 41 | | | | |
| | (7) | For t 1996 | | pose of the applied provisions of the Taxation Administration Act | 42 43 | | | | |
| | | (a) | | default is taken to occur if the contribution deficit is not paid in full the date on which it is payable, and | 44 45 | | | | |
| | | (b) | | erence to the Chief Commissioner is to be read as a reference to the missioner under this Act. | 46 47 | | | | |

| | (8) | If a contribution deficit, or any part of a contribution deficit, is not paid in full by an insurance company to the Commissioner by the date on which it is payable, the unpaid amount and any interest and penalty tax payable on the unpaid amount: | 1 2 3 4 |
|----|-------|--|----------------------------|
| | | (a) constitutes a debt due and payable to the Commissioner, and | 5 |
| | | (b) may be recovered in a court of competent jurisdiction by the Commissioner. | 6 7 |
| | (9) | A contribution deficit is to be paid into the Fund. | 8 |
| | (10) | If the assessment notice specifies a contribution surplus, the Commissioner is to refund the amount of the contribution surplus to the insurance company no later than 30 June 2018. | 9 10 11 |
| | (11) | A refund is to be paid from the Fund. | 12 |
| 34 | Liabi | ility of owner where foreign insurer involved | 13 |
| | (1) | The Commissioner may notify a relevant owner that the owner is responsible for an appropriate proportion of any contribution deficit that is required to be paid by a foreign insurance company that holds a risk in respect of the owner's property. | 14 15 16 17 |
| | (2) | The appropriate proportion is the proportion that the premiums paid by the relevant owner for the final year of the scheme bears to the premium total provided (or taken under this Part to have been provided) by the insurance company in its final return. | 18 19 20 21 |
| | (3) | Any insurer loading is not to be included in the calculation of the contribution deficit under this clause. | 22 23 |
| | (4) | If the relevant owner fails to pay the appropriate proportion of the final contribution within 30 days after it is due, the owner is guilty of an offence. | 24 25 |
| | | Maximum penalty: 10 penalty units. | 26 |
| | (5) | An amount paid by a person under this clause may be deducted from any premium recoverable in the State by or on behalf of the foreign insurance company on the issue or renewal of an insurance policy on the property or may be recovered from the foreign insurance company as a debt by the person making the payment. | 27 28 29 30 31 |
| | (6) | This clause applies whether the premium concerned was received in or outside the State. | 32 33 |
| | (7) | In this clause: <i>relevant owner</i> means a person who is the owner of property in respect of which a foreign insurance company has received a premium subject to a rural fire brigade contribution. | 34 35 36 37 |
| 35 | Com | missioner to make final 2-year assessment | 38 |
| | (1) | The Commissioner must make an assessment, in relation to each insurance company that was required to make rural fire brigade contributions in the final year of the scheme or the previous financial year, of the total amount of rural fire brigade contributions payable by the insurance company for the final 2 years of the scheme (a <i>final 2-year assessment</i>). | 39 40 41 42 43 |
| | (2) | The final 2-year assessment is to consist of the total of the following amounts, as assessed by the Commissioner: | 44 45 |

| | | | | ion payable by the insurance company for the final ne (excluding any insurer loading payable by the y), | 1 2 3 | | |
|------|-------------------------------|---------|--|---|----------------------|--|--|
| | | | (b) the total of rural company for the fi | fire brigade contributions payable by the insurance nancial year commencing on 1 July 2015. | 4 5 | | |
| | | (3) | The Commissioner is to | give the final 2-year assessment to the Monitor. | 6 | | |
| | | (4) | relation to an insurance of | t give to the Monitor a final 2-year assessment in company within 30 days after making an assessment payable by the insurance company for the final year | 7 8 9 10 | | |
| | 36 | Certi | cate evidence | | 11 | | |
| | | | matters is admissible in | the Commissioner that states any of the following proceedings and, in the absence of evidence to the natters stated in the certificate: | 12 13 14 | | |
| | | | (a) the final contribution | on payable by an insurance company, | 15 | | |
| | | | (b) the final 2-year as | sessment for an insurance company, | 16 | | |
| | | | (c) the amount of any | contribution deficit, | 17 | | |
| | | | (d) the amount of any | contribution surplus. | 18 | | |
| | 37 | Tran | tional provision for adj | usted estimates by Minister | 19 | | |
| | | | contributions made to the Treasurer includes a refe | of this Act, as amended by the amending Act, to e New South Wales Rural Fire Fighting Fund by the rence to rural fire brigade contributions made for the ng on 1 July 2016 or a previous financial year. | 20 21 22 23 | | |
| 4.7 | Stat | te Em | rgency Service Act | 1989 No 164 | 24 | | |
| [1] | Sect | ion 24 | Definitions | | 25 | | |
| | expe | nditur | | <i>notice, capital account, contributors, estimated urrent expenditure account, relevant insurance</i> and | 26 27 28 | | |
| | Insert in alphabetical order: | | | | | | |
| | | | section 24K. | s an assessment notice given to a local council under | 30 31 | | |
| | | | SES contribution instalı SES funding amount—s | | 32 | | |
| | | | SES funding target—se | | 33 34 | | |
| | | | State Revenue Commis | <i>issioner</i> —means the Chief Commissioner of State <i>tion Administration Act 1996.</i> | 35 36 | | |
| [2] | Sect | ion 24 | , definition of "SES cor | tribution" | 37 | | |
| | Omi | t "cont | outor". Insert instead "lo | cal council". | 38 | | |
| [3] | Sect | ion 24 | , definition of "SES exp | penditure" | 39 | | |
| | | | ent" wherever occurring. | | 40 | | |
| [4] | Sect | ion 24 | , definition of "SES exp | penditure" | 41 | | |
| r .1 | | | ph (b). | | 42 | | |
| | | | | | | | |

| [5] | 5] Part 5A, Division 2, heading | | | | | | | |
|-----|---|-------------|------------------------------|--|----------------|--|--|--|
| | Omit the heading. Insert instead: | | | | | | | |
| | Divi | sion | SE | S funding target | 3 | | | |
| [6] | Sect | Section 24B | | | | | | |
| | Omit | the se | tion. Inser | t instead: | 5 | | | |
| | 24B | SES | unding ta | rget | 6 | | | |
| | | (1) | Before 31 Minister r | March in each year or a later date agreed to by the Treasurer, the must: | 7 8 | | | |
| | | | targ | pare and, subject to the Treasurer's agreement, adopt a funding get for the State Emergency Service for the next financial year (the <i>S funding target</i>), and | 9 10 11 | | | |
| | | | | pare and adopt an estimate of the amount of the SES funding target blicable to the area of each local council for the next financial year. | 12 13 | | | |
| | | (2) | | ster is to calculate the SES funding target for the financial year using ving formula: | 14 15 | | | |
| | | | $FT_t = E$ | $EE_t + (AE_{t-2} - EE_{t-2})$ | | | | |
| | | | where: | | 16 | | | |
| | | | • | SES funding target for the financial year (represented by "t"). | 17 | | | |
| | | | • | e estimated SES expenditure for the financial year. | 18 | | | |
| | | | 2 years be | the actual SES expenditure for the financial year that commenced efore the period "t" (represented as "t-2"). | 19 20 | | | |
| | | | EE_{t-2} is the 2 years be | ne estimated SES expenditure for the financial year that commenced efore the period "t" (represented as "t-2"). | 21 22 | | | |
| | | (3) | To avoid a from EE_t . | doubt, if $AE_{t-2} - EE_{t-2}$ is a negative amount, that amount is subtracted | 23 24 | | | |
| | | (4) | necessary | hate of SES expenditure for a financial year is to be reduced, as by the amount it is estimated will be paid to the Fund for the year (other than from contributions made by the Treasurer). | 25 26 27 | | | |
| | | (5) | by the ar | I SES expenditure for a financial year is to be reduced, as necessary, mount paid to the Fund for the financial year (other than from ons made by the Treasurer). | 28 29 30 | | | |
| | | | Treasurer | limiting subsections (4) and (5), the Minister may, with the 's agreement, decide that adjustments are not to be made to the SES arget in respect of: | 31 32 33 | | | |
| | | | (a) cer | tain money paid into the Fund, or | 34 | | | |
| | | | | tain amounts it is estimated will be paid to the State Emergency vice for the financial year. | 35 36 | | | |
| | | (7) | area, the M | ining the amount of the SES funding target for each local council Minister may apportion the total estimated expenditure between local reas in the way the Minister thinks fit. | 37 38 39 | | | |
| [7] | Sect | ion 24 | Local co | ouncils to give information to Commissioner | 40 | | | |
| | Omit "estimated expenditure". Insert instead "SES funding target".4 | | | | | | | |

| [8] | Sect | ion 24I |) | 1 |
|-----|------|---------|--|----------------|
| | Omit | the sec | ction. Insert instead: | 2 |
| | 24D | Com | nissioner to assist | 3 |
| | | (1) | To assist the Minister in preparing and adopting the SES funding target for a financial year, the Commissioner must prepare and give to the Minister a written report and recommendations about SES expenditure for the year. | 4 5 6 |
| | | (2) | The Minister must consider the Commissioner's report and recommendations in preparing the estimates. | 7 8 |
| [9] | Part | 5A, Div | visions 3 and 4 | 9 |
| | Omit | the Di | visions. Insert instead: | 10 |
| | Divi | sion (| 3 Treasurer to contribute to Fund | 11 |
| | 24E | Treas | surer to pay contributions | 12 |
| | | (1) | The Treasurer is to pay an annual contribution to the Fund for each financial year. | 13 14 |
| | | (2) | The annual contribution payable is the SES funding amount. | 15 |
| | | (3) | The Treasurer may direct the payment of additional contributions to the Fund, subject to any terms and conditions approved by the Treasurer. | 16 17 |
| | | (4) | The annual contribution, and any additional contributions, are to be paid out of money provided by Parliament. | 18 19 |
| | 24F | Cont | ribution payable in instalments | 20 |
| | | | The annual contribution is to be paid by the Treasurer in instalments on or before the following dates: | 21 22 |
| | | | (a) 1 August, | 23 |
| | | | (b) 1 October, | 24 |
| | | | (c) 1 January, | 25 |
| | | | (d) 1 April. | 26 |
| | 24G | SES | funding amount | 27 |
| | | (1) | As soon as practicable after the commencement of each financial year, the Minister must prepare and, subject to the Treasurer's agreement, adopt an update of SES funding for the financial year (the <i>SES funding amount</i>). | 28 29 30 |
| | | (2) | The SES funding amount is the estimated SES expenditure for the financial year (that is, EE_t in the SES funding target for the financial year), adjusted and updated in accordance with this section. | 31 32 33 |
| | | (3) | Estimated SES expenditure is to be adjusted for the purposes of the SES funding amount so that the estimate: | 34 35 |
| | | | (a) is reduced or increased, as appropriate, for any difference between the estimate of SES expenditure for the financial year, as adopted in the SES funding target, and an updated estimate of that amount, and | 36 37 38 |
| | | | (b) is reduced or increased, as appropriate, for any difference between the SES funding amount for the previous financial year and the actual amount of SES expenditure for that financial year. | 39 40 41 |
| | | | | |

| | | (4) | agree | out limiting subsection (3), the Minister may, with the Treasurer's ement, decide that adjustments are not to be made to the SES funding int in respect of: | 1 2 3 | | | |
|------|---|----------|---------|--|----------------|--|--|--|
| | | | (a) | any surplus in the Fund at the end of the previous financial year, or | 4 | | | |
| | | | (b) | certain money paid into the Fund, or | 5 | | | |
| | | | (c) | certain amounts it is estimated will be paid to the State Emergency Service for the financial year. | 6 7 | | | |
| | | (5) | sectio | idjustment to the estimated SES expenditure that is made under this on has no effect on, and is not to be used in the calculation of, the SES ing target for a financial year. | 8 9 10 | | | |
| | Divi | sion | 4 | Local councils to pay SES contribution | 11 | | | |
| | 24H | Cour | ncil to | pay SES contribution | 12 | | | |
| | | | | cal council is to pay to the State Revenue Commissioner a SES ibution for each financial year. | 13 14 | | | |
| | 241 | Amo | unt of | contribution | 15 | | | |
| | | (1) | | amount of the SES contribution is the amount determined by the Minister ach local council. | 16 17 | | | |
| | | (2) | | Minister is to determine the contribution payable by a local council on the of the SES funding target for each local council area. | 18 19 | | | |
| | | (3) | | contribution payable by local councils for each local council area is 11.7% e SES funding target applicable to the area. | 20 21 | | | |
| | | (4) | give | cal council or an officer of a local council must, if asked by the Minister, the Minister any document or information required by the Minister to mine the local council's SES contribution. | 22 23 24 | | | |
| | 24IA Contribution payable in instalments | | | on payable in instalments | 25 | | | |
| | A SES contribution payable by a local council for a financial year is jin 4 instalments (each of which is a <i>SES contribution instalment</i>). | | | | 26 27 | | | |
| | 24IB | Minis | ster to | advise State Revenue Commissioner | 28 | | | |
| | | | | Minister is to advise the State Revenue Commissioner, by the date of pril that occurs immediately before the commencement of a financial of: | 29 30 31 | | | |
| | | | (a) | the amount of the SES contribution payable by each local council for the financial year, and | 32 33 | | | |
| | | | (b) | the amount of the SES contribution instalments payable by each local council. | 34 35 | | | |
| [10] | Part 5A, Division 5, heading | | | | | | | |
| | Omi | t the he | ading. | Insert instead: | 37 | | | |
| | Divi | ision | 5 | Payment of contributions by local councils | 37 38 | | | |

| [11] | Section 24J | | | | | | |
|------|-------------|-----------------------------|--------------------|--|----------------------------|--|--|
| | Omi | t the se | ction. I | Insert instead: | 2 | | |
| | 24J | Whe | n insta | Ilments are to be paid | 3 | | |
| | | | cound Comr | cal council must, in accordance with an instalment notice given to the cil by the State Revenue Commissioner, pay to the State Revenue missioner a SES contribution instalment on or before each of the wing days in a financial year: 30 September, 31 December, | 4 5 6 7 8 9 | | |
| | | | (c) | 31 March, | 10 | | |
| | | | (d) | 30 June. | 11 | | |
| [12] | Sect | ions 2 | 4K–24 | LA | 12 | | |
| | Omit | t sectio | ns 24K | and 24L. Insert instead: | 13 | | |
| | 24K | IK Annual assessment notice | | | | | |
| | | (1) | requit that s | State Revenue Commissioner must give to each local council that is red to pay a SES contribution for a financial year an assessment notice tets out the amount of the SES contribution payable by the local council hat year. | 15 16 17 18 | | |
| | | (2) | | assessment notice must be given to the local council by 30 April in the before the financial year for which the SES contribution is payable. | 19 20 | | |
| | 24L | Insta | Iment notices | | | | |
| | | | requi | State Revenue Commissioner must give to each local council that is red to pay a SES contribution instalment a written notice (an <i>instalment</i> e) that specifies: | 22 23 24 | | |
| | | | (a) | the amount of the SES contribution instalment payable under the instalment notice, and | 25 26 | | |
| | | | (b) | the date by which the SES contribution instalment is payable. | 27 | | |
| : | 24LA | SES | contril | bution is debt payable to State Revenue Commissioner | 28 | | |
| | | (1) | to the recov | whole or part of a SES contribution payable by a local council is not paid e State Revenue Commissioner as required, the unpaid amount is verable by the State Revenue Commissioner as a debt in a court of betent jurisdiction. | 29 30 31 32 | | |
| | | (2) | relation Part 5 | applied provisions of the <i>Taxation Administration Act 1996</i> apply in on to the SES contribution payable by a local council under this Act as if 5A of this Act were a taxation law and the contribution were a tax payable council under a taxation law. | 33 34 35 36 | | |
| | | (3) | | <i>applied provisions</i> of the <i>Taxation Administration Act 1996</i> are the wing provisions of that Act: | 37 38 | | |
| | | | (a) | Division 1 (Interest) of Part 5, | 39 | | |
| | | | (b) | sections 50 and 55 (which prohibit the giving of false and misleading information), | 40 41 | | |
| | | | (c) | Division 1 (Tax officers), Division 2 (Investigation) and Division 3 (Secrecy) of Part 9, | 42 43 | | |
| | | | (d) | sections 114, 115, 116, 118, 119 and 120, | 44 | | |

| | | (| (e) | any provisions that are relevant to the interpretation of the above provisions. | 1 2 |
|------|----------|-----------|--------------|--|----------------------|
| | | Ć | Comr | applied provisions apply to an assessment made by the State Revenue missioner under this Act in the same way as they apply to an assessment t that Act. | 3 4 5 |
| | | a C | pply Comr | void doubt, Part 10 of the <i>Taxation Administration Act 1996</i> does not to a SES contribution or to the exercise of any of the State Revenue missioner's functions in respect of SES contributions (including under oplied provisions of the <i>Taxation Administration Act 1996</i>). | 6 7 8 9 |
| [13] | Sectio | n 24M | How | <i>r</i> contribution is to be funded | 10 |
| | Insert a | at the er | nd of | f the section: | 11 |
| | | c p | ontri | s of a local council derived from donations and other voluntary ibutions made for the purposes of this Act may not be used towards the ent of SES contributions by the local council unless approved by the ster. | 12 13 14 15 |
| [14] | Part 5/ | A, Divis | sion | 6 Contributions by insurance companies | 16 |
| | Omit tl | he Divi | sion. | | 17 |
| [15] | Sectio | n 24V I | New | South Wales State Emergency Service Fund | 18 |
| | | | | (2), (3) and (4). Insert instead: | 19 |
| | | (3) T | There | e is to be paid into the Fund: | 20 |
| | | · · · | (a) | all contributions payable by the Treasurer under this Part, and | 21 |
| | | (| (b) | any other money appropriated by Parliament for payment into the Fund, and | 22 23 |
| | | | (c) | the proceeds of investment of money in the Fund, and | 24 |
| | | (| (d) | any other money required by law to be paid into the Fund. | 25 |
| [16] | Sectio | n 24V (| (5) | | 26 |
| | Omit " | recurre | nt ex | xpenditure account". Insert instead "Fund". | 27 |
| [17] | Sectio | n 24V (| (6) | | 28 |
| | Omit " | capital | acco | ount". Insert instead "Fund". | 29 |
| [18] | Sectio | n 24W | Man | nagement of unspent funds | 30 |
| | Omit tl | he secti | on. | | 31 |
| [19] | Sectio | n 24Y | | | 32 |
| | | | ion. I | Insert instead: | 33 |
| | 24Y | Certific | ate | evidence | 34 |
| | | A a | A cer | tificate signed by the Minister that states any of the following matters is ssible in proceedings and, in the absence of evidence to the contrary, is of the matters stated in the certificate: | 35 36 37 |
| | | | (a) | the SES funding target, as adopted by the Minister under this Part, | 38 |
| | | ` | (b) | the SES funding target applicable to a local council area, as adopted by the Minister under this Part, | 39 40 |
| | | | | | |

| | | | (c) | the amount of the SES contribution determined by the Minister for a specified local council for the financial year. | 1 2 |
|------|---------|----------|----------------------|--|----------------------|
| [20] | Sche | dule 1 | Savi | ngs, transitional and other provisions | 3 |
| | Inser | t at the | end o | of clause 1 (1): | 4 |
| | an | ny othe | r Act | that amends this Act | 5 |
| [21] | Sche | dule 1 | , Part | 6 | 6 |
| | Inser | t after | Part 5 | : | 7 |
| | Par | t 6 | | ovisions consequent on enactment of Fire and lergency Services Levy Act 2017 | 8 9 |
| | 13 Defi | | nitions | | |
| | | | In th | is Part: | 11 |
| | | | | nding Act means the Fire and Emergency Services Levy Act 2017. | 12 |
| | | | | contribution—see clause 16. | 13 |
| | | | 5 | <i>return</i> —see clause 15. <i>Year of the scheme</i> means the financial year commencing on 1 July 2016. | 14 |
| | | | | <i>rance company</i> means any body corporate, partnership, association, | 15 16 |
| | | | | erwriter or person that or who: | 10 |
| | | | (a) | issues or undertakes liability under policies of insurance against loss of or damage to any property situated in New South Wales, or | 18 19 |
| | | | (b) | receives premiums in respect of such policies of insurance on behalf of or for transmission to any body corporate, partnership, association, underwriter or person outside New South Wales. | 20 21 22 |
| | | | insu | rer loading—see clause 16. | 23 |
| | | | <i>Mon</i> unde | <i>Litor</i> means the Emergency Services Levy Insurance Monitor appointed or the <i>Emergency Services Levy Insurance Monitor Act 2016</i> . | 24 25 |
| | | | the S | <i>vant insurance</i> means insurance against loss of or damage to property in State under the classes of policies specified in Schedule 2, as in force ediately before its repeal by the <i>Fire and Emergency Services Levy Act</i> 7. | 26 27 28 29 |
| | | | <i>total</i> paid | <i>amount</i> , in relation to premiums, includes any brokerage or commission or due to be paid or allowed to be paid on: | 30 31 |
| | | | (a) | the premiums, or | 32 |
| | | | (b) | bonuses or return premiums allowed in respect of policies of insurance the subject of the premiums, or | 33 34 |
| | | | (c) | such part of the premiums received by or due to the company as is paid or due to be paid by way of reinsurance by the company to another insurance company in the State, | 35 36 37 |
| | | | | does not include duty payable under the <i>Duties Act 1997</i> in respect of the subject of the premiums. | 38 39 |
| | 14 | Ame | ndme | nts have effect from 1 July 2017 | 40 |
| | | (1) | respe | amendments to Part 5A of this Act made by the amending Act apply in ect of the financial year commencing on 1 July 2017 and subsequent acial years. | 41 42 43 |

| | (2) | Subject to this Part, the amendments do not affect any liability for a SES contribution for a financial year commencing before 1 July 2017 and Part 5A, and Schedule 2, as in force immediately before the amendments made to this Act by the amending Act, continue to apply in respect of any such financial year as if the amendments had not been made. | | | | | |
|----|------------------------------------|---|---|--|----------------|--|--|
| 15 | Final | • | | | 5 | | |
| 10 | (1) | Final returns to be provided by insurance companies | | | | | |
| | (1) | by th | n insurance company must, by 30 September 2017 or another date specified the Commissioner by notice published in the Gazette, give to the commissioner: | | | | |
| | | (a) | amou | arn in the form approved by the Commissioner showing the total int of premiums received by or due to the company for the previous cial year for relevant insurance (a <i>final return</i>), and | 10 11 12 | | |
| | | (b) | | tificate from an auditor that relates to the return and complies with ause (2). | 13 14 | | |
| | (2) | The c | certific | ate from the auditor must: | 15 | | |
| | | (a) | be in | the form approved by the Commissioner, and | 16 | | |
| | | (b) | be fro | om an auditor who is: | 17 | | |
| | | | (i) | a registered company auditor within the meaning of the <i>Corporations Act 2001</i> of the Commonwealth, or | 18 19 | | |
| | | | (ii) | a person not resident in the State who has qualifications that, in the Commissioner's opinion, are appropriate for the giving of the certificate. | 20 21 22 | | |
| | (3) | An insurance company is guilty of an offence if it: | | | | | |
| | | (a) | | to give the Commissioner a return and certificate as required by lause, or | 24 25 | | |
| | | (b) | gives partic | the Commissioner a return that is false or misleading in a material cular. | 26 27 | | |
| | Maximum penalty: 20 penalty units. | | penalty: 20 penalty units. | 28 | | | |
| 16 | Calculation of final contribution | | | | | | |
| | (1) |) The Commissioner must, by the end of 30 November 2017, calculate a final contribution for each liable insurance company. | | | | | |
| | (2) | payal | ble by | <i>ntribution</i> is the total SES contribution that would have been the insurance company for the final year of the scheme if the had been calculated on the basis of: | 32 33 34 | | |
| | | (a) | the pr and | remium total provided by the insurance company in its final return, | 35 36 | | |
| | | (b) | the pr return | remium totals provided by all insurance companies in their final ns. | 37 38 | | |
| | (3) | If an insurance company fails to provide a final return to the Commissioner within the time required under clause 15, or any further time (not exceeding 30 days) allowed by the Commissioner: | | | | | |
| | | (a) | comp | Commissioner may estimate the premium total for that insurance bany using any information available to the Commissioner uding information from previous returns, if any), and | 42 43 44 | | |
| | | (b) | | estimate is taken to be the premium total provided by the insurance bany in its final return. | 45 46 | | |

17

(4)If the Commissioner estimates the premium total for an insurance company 1 under subclause (3), the final contribution payable by that insurance company 2 is that estimate plus an insurer loading of 50% of that estimate. 3 (5)In this clause: 4 *liable insurance company* means an insurance company liable for a SES 5 contribution in the final year of the scheme. 6 premium total means the total amount of premiums of an insurance company 7 that are subject to an SES contribution. 8 Final assessment 9 (1)The Commissioner must give to each insurance company that made or is 10 required to make a SES contribution for the final year of the scheme a final 11 assessment notice for that year. 12 (2)The final assessment notice must state the following: 13 (a) the final contribution payable by the insurance company for the final 14 year of the scheme, 15 the amount of any SES contribution already paid by the insurance (b) 16 company for that financial year including, if paid in instalments, the 17 total of all instalments already paid (the *total contribution already* 18 paid), 19 if the final contribution payable by the insurance company exceeds the (c) 20 total contribution already paid: 21 the difference between those amounts (a *contribution deficit*), 22 (i) and 23 (ii) the date by which the contribution deficit must be paid to the 24 Commissioner (being a date that is no earlier than 30 days after 25 the assessment notice is given to the insurance company), 26 (d) if the total contribution already paid exceeds the final contribution 27 payable by the insurance company, the difference between those 28 amounts (a contribution surplus). 29 If the assessment notice specifies a contribution deficit, the insurance (3) 30 company must pay the contribution deficit to the Commissioner. 31 The contribution deficit is payable to the Commissioner by the date for (4)32 payment specified in the assessment notice or any later date permitted by the 33 Commissioner. 34 If an insurance company fails to pay a contribution deficit in full by the date (5)35 on which it is payable to the Commissioner, the applied provisions of the 36 Taxation Administration Act 1996 apply to the unpaid amount as if the unpaid 37 amount were a tax and this Act were a taxation law. 38 The *applied provisions* of the *Taxation Administration Act 1996* are the (6)39 following provisions of that Act: 40 Division 1 (Interest) of Part 5, (a) 41 Division 2 (Penalty tax) of Part 5 (except sections 28–30), (b) 42 any provisions that are relevant to the interpretation of the above (c) 43 provisions. 44

| (7) For the purpose of the applied provisions of the <i>Taxation Administration A 1996</i> : | ct |
|--|----|
|--|----|

1 2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

29

30

31

32

33

34

35

36

37

38

39

40

41

42

43

44

- (a) a tax default is taken to occur if the contribution deficit is not paid in full by the date on which it is payable, and
- (b) a reference to the Chief Commissioner is to be read as a reference to the Commissioner under this Act.
- (8) If a contribution deficit, or any part of a contribution deficit, is not paid in full by an insurance company to the Commissioner by the date on which it is payable, the unpaid amount and any interest and penalty tax payable on the unpaid amount:
 - (a) constitutes a debt due and payable to the Commissioner, and
 - (b) may be recovered in a court of competent jurisdiction by the Commissioner.
- (9) A contribution deficit is to be paid into the Fund.
- (10) If the assessment notice specifies a contribution surplus, the Commissioner is to refund the amount of the contribution surplus to the insurance company no later than 30 June 2018.
- (11) A refund is to be paid from the Fund.

18 Liability of owner where foreign insurer involved

- (1) The Commissioner may notify a relevant owner that the owner is responsible for an appropriate proportion of any contribution deficit that is required to be paid by a foreign insurance company that holds a risk in respect of the owner's property.
- (2) The appropriate proportion is the proportion that the premiums paid by the relevant owner for the final year of the scheme bears to the premium total provided (or taken under this Part to have been provided) by the insurance company in its final return.
- (3) Any insurer loading is not to be included in the calculation of the contribution deficit under this clause.
- (4) If the relevant owner fails to pay the appropriate proportion of the final contribution within 30 days after it is due, the owner is guilty of an offence. Maximum penalty: 10 penalty units.
- (5) An amount paid by a person under this clause may be deducted from any premium recoverable in the State by or on behalf of the foreign insurance company on the issue or renewal of an insurance policy on the property or may be recovered from the foreign insurance company as a debt by the person making the payment.
- (6) This clause applies whether the premium concerned was received in or outside the State.
 - In this clause: *relevant owner* means a person who is the owner of property in respect of which a foreign insurance company has received a premium subject to a SES contribution.

19 Commissioner to make final 2-year assessment

(1) The Commissioner must make an assessment, in relation to each insurance 45 company that was required to make SES contributions in the final year of the 46

(7)

| | | scheme or the previous financial year, of the total amount of SES contributions payable by the insurance company for the final 2 years of the scheme (a <i>final 2-year assessment</i>). | 1 2 3 | | | |
|------|---|--|----------------------------------|--|--|--|
| | (2) | The final 2-year assessment is to consist of the total of the following amounts, as assessed by the Commissioner: | 4 5 | | | |
| | | (a) the final contribution payable by the insurance company for the final year of the scheme (excluding any insurer loading payable by the insurance company), | 6 7 8 | | | |
| | | (b) the total of SES contributions payable by the insurance company for the financial year commencing on 1 July 2015. | 9 10 | | | |
| | (3) | The Commissioner is to give the final 2-year assessment to the Monitor. | 11 | | | |
| | (4) | The Commissioner must give to the Monitor a final 2-year assessment in relation to an insurance company within 30 days after making an assessment of the final contribution payable by the insurance company for the final year of the scheme. | 12 13 14 15 | | | |
| 20 | Certi | ificate evidence | 16 | | | |
| | | A certificate signed by the Commissioner that states any of the following matters is admissible in proceedings and, in the absence of evidence to the contrary, is proof of the matters stated in the certificate: | 17 18 19 | | | |
| | | (a) the final contribution payable by an insurance company, | 20 | | | |
| | | (b) the final 2-year assessment of an insurance company, | 21 | | | |
| | | (c) the amount of any contribution deficit, | 22 | | | |
| | | (d) the amount of any contribution surplus. | 23 | | | |
| 21 | Tran | sitional provision for adjusted estimates by Minister | 24 | | | |
| | | A reference in Part 5A of this Act, as amended by the amending Act, to contributions made to the New South Wales State Emergency Service Fund by the Treasurer includes a reference to SES contributions made for the financial year commencing on 1 July 2016 or a previous financial year. | 25 26 27 28 | | | |
| Sch | edule 2 | 2 Contributions of insurance companies | 29 | | | |
| Omi | t the So | chedule. | 30 | | | |
| Тах | ation | Administration Act 1996 No 97 | 31 | | | |
| Sect | tion 4 I | Meaning of "taxation laws" | 32 | | | |
| Inse | rt at the | e end of the section: | 33 | | | |
| | | Note. The <i>Fire and Emergency Services Levy Act 2017</i> applies parts of this Act to the levy payable under that Act if the responsibility for levy recovery is transferred to the Chief Commissioner or if the Chief Commissioner requires a waived amount of the levy to be repaid. The <i>Fire and Emergency Services Levy Act 2017</i> also applies parts of this Act to the provide the target of the parts of the Act to the fire and Emergency Services Levy Act 2017 also applies parts of the Act to the provide the target of the parts of the Act to the provide the target of the parts of the Act to the provide the target of the parts of the Act to the provide the target of the parts of the Act to the provide the target of the parts of the p | 34 35 36 37 38 39 | | | |
| | collection instalments payable by a council under that Act as if those instalments were a tax. | | | | | |
| | In addition, some of the provisions of this Act apply to the contributions payable by councils under the following: | | | | | |
| | | (a) Part 5 of the Fire Brigades Act 1989, | 43 | | | |
| | | (b) Part 5 of the <i>Rural Fires Act 1997</i>, (c) Part 5A of the <i>State Emergency Service Act 1989</i>. | 44 | | | |
| | | (c) Part 5A of the State Emergency Service Act 1989. | 45 | | | |

[22]

4.8

| 4.9 | Valu | uatio | n of L | _and Act 1916 No 2 | 1 |
|-----|---|-------|----------------------------------|---|----------------------------------|
| [1] | Section 4A Insert after section 4: | | | | |
| | | | | | |
| | | | (1) | In this Act: | |
| | | | (a) | a reference to a rate or tax includes a reference to the fire and emergency services levy under the <i>Fire and Emergency Services Levy Act 2017</i> , and | 6 7 8 |
| | | | (b) | a reference to the levying of a rate or tax by a council includes a reference to the charging of that fire and emergency services levy by a council. | 9 10 11 |
| | | (2) | subje Note (for e provi | vever, the application of this Act to the fire and emergency services levy is ect to provisions of the <i>Fire and Emergency Services Levy Act 2017</i> . The <i>Fire and Emergency Services Levy Act 2017</i> enables certain unvalued land example land on Lord Howe Island) to be valued for the purposes of the levy as ded for by the regulations under that Act. The valuation method is different from method by which a general valuation for land is ascertained under Part 1B. | 12 13 14 15 16 17 |
| [2] | Section 48 Furnishing valuation lists to authorities | | | 18 | |
| | Omit "4 years" from section 48 (2) (b). Insert instead "3 years". | | | | |
| [3] | Section 48 (3) | | | | 20 |
| | Omit the subsection. | | | | 21 |
| [4] | Section 67 Valuation for the purposes of the Fire Brigades Act 1989 | | | | 22 |
| | Omit "section 53" from section 67 (3) (h). Insert instead "section 51". | | | | 23 |